## UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

### **FORM 10-Q**

## QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2018

Commission File Number: 001-33401

### CINEMARK HOLDINGS, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization) 20-5490327 (I.R.S. Employer Identification No.)

3900 Dallas Parkway
Suite 500
Plano, Texas
(Address of principal executive offices)

75093 (Zip Code)

,		· •	
	Registrant's telephone number, including area code: (972	2) 665-1000	
•	whether the registrant (1) has filed all reports required to be filed by Section 1 h shorter period that the registrant was required to file such reports), and (2)	` '	_
•	whether the registrant has submitted electronically and posted on its corporate Rule 405 of Regulation S-T ( $\S 232.405$ of this chapter) during the preceding files). Yes $\boxtimes$ No $\square$		
	whether the registrant is a large accelerated filer, an accelerated filer, a non-ations of "large accelerated filer," "accelerated filer," "smaller reporting comp		
Large accelerated filer	$\boxtimes$	Accelerated filer	
Non-accelerated filer	☐ (Do not check if a smaller reporting company)	Smaller reporting company	
Emerging growth company			
2 2 2	ompany, indicate by check mark if the registrant has elected not to use the extended pursuant to Section 13(a) of the Exchange Act. $\Box$	ended transition period for complying with any new of	r revised
Indicate by check mark w	hether the registrant is a shell company (as defined in Rule 12b-2 of the Exchan	ge Act). Yes□ No ⊠	
As of July 31, 2018, 116,	829,079 shares of common stock were issued and outstanding.		

## CINEMARK HOLDINGS, INC. AND SUBSIDIARIES TABLE OF CONTENTS

PART I.	FINANCIAL INFORMATION	<u>Page</u>
Item 1.	Financial Statements	4
	Condensed Consolidated Balance Sheets as of June 30, 2018 and December 31, 2017 (unaudited)	4
	Condensed Consolidated Statements of Income for the three and six months ended June 30, 2018 and 2017 (unaudited)	5
	Condensed Consolidated Statements of Comprehensive Income for the three and six months ended June 30, 2018 and 2017 (unaudited)	6
	Condensed Consolidated Statements of Cash Flows for the six months ended June 30, 2018 and 2017 (unaudited)	7
	Notes to Condensed Consolidated Financial Statements (unaudited)	8
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	26
Item 3.	Quantitative and Qualitative Disclosures About Market Risk	38
Item 4.	Controls and Procedures	38
PART II.	OTHER INFORMATION	
Item 1.	<u>Legal Proceedings</u>	39
Item 1A.	Risk Factors	39
Item 6.	<u>Exhibits</u>	40
<u>SIGNATU</u>	<u>RES</u>	41
	$\gamma$	

#### **Cautionary Statement Regarding Forward-Looking Statements**

Certain matters within this Quarterly Report on Form 10Q include "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. The "forward-looking statements" may include our current expectations, assumptions, estimates and projections about our business and our industry. They may include statements relating to future revenues, expenses and profitability, the future development and expected growth of our business, projected capital expenditures, attendance at movies generally or in any of the markets in which we operate, the number or diversity of popular movies released and our ability to successfully license and exhibit popular films, national and international growth in our industry, competition from other exhibitors and alternative forms of entertainment and determinations in lawsuits in which we are defendants. Forward-looking statements can be identified by the use of words such as "may," "should," "could," "estimates," "predicts," "potential," "continue," "anticipates," "plans," "expects," "future" and "intends" and similar expressions. Forward-looking statements may involve known and unknown risks, uncertainties and other factors that may cause the actual results or performance to differ from those projected in the forward-looking statements. These statements are not guarantees of future performance and are subject to risks, uncertainties and other factors, some of which are beyond our control and difficult to predict and could cause actual results to differ materially from those expressed or forecasted in the forward-looking statements. For a description of the risk factors, please review the "Risk Factors" section or other sections in the Company's Annual Report on Form 10-K filed February 23, 2018 and quarterly reports on Form 10-Q, filed with the Securities and Exchange Commission. All forward-looking statements are expressly qualified in their entirety by such risk factors. We undertake

#### PART I - FINANCIAL INFORMATION

#### **Item 1. Financial Statements**

## CINEMARK HOLDINGS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (in thousands, except share data, unaudited)

		ine 30, 2018	December 31, 2017		
Assets					
Current assets					
Cash and cash equivalents	\$	504,712 \$			
Inventories Accounts receivable		17,685	17,5		
		80,308 4,815	89,2 11,7		
Current income tax receivable Prepaid expenses and other		20,667	16,5		
Total current assets		628,187	657,5		
Total current assets		028,187	637,3		
Theatre properties and equipment		3,362,779	3,328,5		
Less: accumulated depreciation and amortization		1,545,042	1,500,5		
Theatre properties and equipment, net		1,817,737	1,828,0		
Other assets					
Goodwill		1,281,170	1,284,0		
Intangible assets - net		332,506	336,7		
Investment in NCM		199,019	200,5		
Investments in and advances to affiliates		145,081	120,0		
Long-term deferred tax asset		4,050	4,0		
Deferred charges and other assets - net		47,488	39,7		
Total other assets		2,009,314	1,985,2		
T. (.)	ď.				
Total assets	\$	4,455,238 \$	4,470,8		
Liabilities and equity					
Current liabilities					
Current portion of long-term debt	\$	7,984 \$			
Current portion of capital lease obligations		26,671	25,5		
Current income tax payable		6,850	5,5		
Current liability for uncertain tax positions		3,650	11,8		
Accounts payable and accrued expenses		394,679	418,9		
Total current liabilities		439,834	468,9		
Long-term liabilities					
Long-term debt, less current portion		1,774,782	1,780,3		
Capital lease obligations, less current portion		239,005	251,1		
Long-term deferred tax liability		141,307	121,7		
Long-term liability for uncertain tax positions		13,405	8,3		
Deferred lease expenses		39,750	40,9		
Deferred revenue - NCM		295,264	351,7		
Other long-term liabilities		48,081	41,9		
Total long-term liabilities		2,551,594	2,596,2		
Commitments and contingencies (see Note 17)					
Equity					
Cinemark Holdings, Inc.'s stockholders' equity:					
Common stock, \$0.001 par value: 300,000,000 shares authorized, 121,449,393 shares issued and 116,829,079 shares outstanding at June 30, 2018 and 121,000,903 shares issued and 116,475,033 shares					
outstanding at December 31, 2017		121	1:		
Additional paid-in-capital		1,147,966	1,141,0		
Treasury stock, 4,620,314 and 4,525,870 shares, at cost, at June 30, 2018					
and December 31, 2017, respectively		(79,259)	(76,3		
Retained earnings		691,910	582,2		
Accumulated other comprehensive loss		(309,306)	(253,2		
Total Cinemark Holdings, Inc.'s stockholders' equity		1,451,432	1,393,7		
Noncontrolling interests		12,378	11,8		
Total equity		1,463,810	1,405,6		
Total liabilities and equity	\$	4,455,238 \$	4,470,8		

# CINEMARK HOLDINGS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF INCOME (in thousands, except per share data, unaudited)

	Three Months	Ended .	June 30,	Six Months Ended June 30,				
	 2018		2017		2018		2017	
Revenues	 							
Admissions	\$ 508,870	\$	449,880	\$	961,494	\$	926,349	
Concession	305,306		262,322		567,078		530,546	
Other	74,877		38,993		140,452		73,910	
Total revenues	889,053		751,195		1,669,024		1,530,805	
Cost of operations								
Film rentals and advertising	287,206		246,556		528,121		499,374	
Concession supplies	51,033		41,839		91,857		83,939	
Salaries and wages	100,344		89,812		193,502		174,013	
Facility lease expense	81,190		82,388		163,281		166,650	
Utilities and other	115,602		91,053		225,034		179,410	
General and administrative expenses	43,031		37,834		85,415		76,050	
Depreciation and amortization	64,290		59,137		128,685		116,493	
Impairment of long-lived assets	2,788		4,301		3,379		4,574	
Loss on disposal of assets and other	16,901		54		20,840		888	
Total cost of operations	762,385		652,974		1,440,114		1,301,391	
Operating income	126,668		98,221		228,910		229,414	
Other income (expense)								
Interest expense	(28,466)		(26,522)		(55,581)		(52,891)	
Loss on debt amendments	_		(246)		(1,484)		(246)	
Interest income	2,862		1,380		5,100		2,713	
Foreign currency exchange gain (loss)	(5,199)		(155)		(3,821)		1,434	
Distributions from NCM	3,424		2,772		9,782		9,560	
Interest expense - NCM	(4,913)		_		(9,892)		_	
Equity in income of affiliates	6,414		5,805		15,050		15,865	
Total other expense	(25,878)		(16,966)		(40,846)		(23,565)	
Income before income taxes	100,790		81,255		188,064		205,849	
Income taxes	 18,326		29,445		43,423		73,845	
Net income	\$ 82,464	\$	51,810	\$	144,641	\$	132,004	
Less: Net income attributable to noncontrolling interests	329		571		485		1,037	
Net income attributable to Cinemark Holdings, Inc.	\$ 82,135	\$	51,239	\$	144,156	\$	130,967	
Weighted average shares outstanding								
Basic	116,091		115,785		115,988		115,707	
Diluted	116,268		116,072		116,238		116,020	
Earnings per share attributable to Cinemark Holdings, Inc.'s common stockholders								
Basic	\$ 0.70	\$	0.44	\$	1.23	\$	1.12	
Diluted	\$ 0.70	\$	0.44	\$	1.23	\$	1.12	
Dividends declared per common share	\$ 0.32	\$	0.29	\$	0.64	\$	0.58	

# CINEMARK HOLDINGS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (In thousands, unaudited)

		Three Months I	Ended J	Six Months En	ded Ju	ne 30,	
	2018			2017	2018		2017
Net income	\$	82,464	\$	51,810	\$ 144,641	\$	132,004
Other comprehensive income (loss), net of tax							
Other comprehensive income (loss) in equity method							
investments		(116)		(95)	20		103
Foreign currency translation adjustments		(56,248)		(18,401)	(56,044)		(3,508)
Total other comprehensive loss, net of tax		(56,364)		(18,496)	(56,024)		(3,405)
Total comprehensive income, net of tax		26,100		33,314	88,617		128,599
Comprehensive income attributable to noncontrolling interests		(329)		(571)	(485)		(1,037)
Comprehensive income attributable to Cinemark							
Holdings, Inc.	\$	25,771	\$	32,743	\$ 88,132	\$	127,562

#### CINEMARK HOLDINGS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (in thousands, unaudited)

		50,		
		2018		2017
Operating activities				
Net income	\$	144,641	\$	132,004
Adjustments to reconcile net income to cash provided by operating activities:				
Depreciation		126,981		115,755
Amortization of intangible and other assets and favorable/unfavorable leases		1,704		738
Amortization of long-term prepaid rents		1,236		989
Amortization of debt issue costs		2,913		3,063
Loss on debt amendments		1,484		246
Amortization of deferred revenues, deferred lease incentives and other		(10,762)		(7,923)
Impairment of long-lived assets		3,379		4,574
Share based awards compensation expense		6,878		6,444
Loss on disposal of assets and other		20,840		888
Deferred lease expenses		(932)		(722)
Equity in income of affiliates		(15,050)		(15,865)
Deferred income tax expenses		7,669		14,515
Distributions from equity investees		16,255		14,919
Changes in assets and liabilities and other		(27,333)		(33,935)
Net cash provided by operating activities		279,903		235,690
Investing activities				
Additions to theatre properties and equipment		(162,589)		(182,800)
Acquisitions of theatres in the U.S. and international markets, net of cash acquired		(11,508)		(40,829)
Proceeds from sale of theatre properties and equipment and other		2,135		14,521
Investment in joint ventures and other, net		(19,467)		(466)
Net cash used for investing activities		(191,429)		(209,574)
Financing activities				
Dividends paid to stockholders		(74,723)		(67,528)
Payroll taxes paid as a result of stock withholdings		(2,905)		(2,921)
Repayments of long-term debt		(3,298)		(1,427)
Payment of debt issue costs		(5,103)		(521)
Fees paid related to debt amendments		(704)		(246)
Payments on capital leases		(12,377)		(10,143)
Other		_		(311)
Net cash used for financing activities		(99,110)		(83,097)
Effect of exchange rate changes on cash and cash equivalents		(7,199)		(75)
Decrease in cash and cash equivalents		(17,835)		(57,056)
Cash and cash equivalents:				
Beginning of period		522,547		561,235
End of period	\$	504,712	\$	504,179

Supplemental information (see Note 14)

In thousands, except share and per share data

#### 1. The Company and Basis of Presentation

Cinemark Holdings, Inc. and subsidiaries (the "Company") operates in the motion picture exhibition industry, with theatres in the United States ("U.S."), Brazil, Argentina, Chile, Colombia, Peru, Ecuador, Honduras, El Salvador, Nicaragua, Costa Rica, Panama, Guatemala, Bolivia, Curacao and Paraguay.

The accompanying condensed consolidated balance sheet as of December 31, 2017, which was derived from audited financial statements, and the unaudited condensed consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("U.S. GAAP") for interim financial information and in accordance with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required by U.S. GAAP for complete consolidated financial statements. In the opinion of management, all adjustments, consisting of normal recurring adjustments, considered necessary for a fair presentation have been included. The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the amounts reported in the financial statements and the accompanying notes. Actual results could differ from these estimates. Majority-owned subsidiaries that the Company has control of are consolidated while those affiliates of which the Company owns between 20% and 50% and does not control are accounted for under the equity method. Those affiliates of which the Company would account for its investment under the equity method. The results of these subsidiaries and affiliates are included in the condensed consolidated financial statements effective with their formation or from their dates of acquisition. Intercompany balances and transactions are eliminated in consolidation.

These condensed consolidated financial statements should be read in conjunction with the audited annual consolidated financial statements and the notes thereto for the year ended December 31, 2017, included in the Annual Report on Form 10-K filed February 23, 2018 by the Company under the Securities Exchange Act of 1934, as amended (the "Exchange Act"). Operating results for the three and six months ended June 30, 2018 are not necessarily indicative of the results to be achieved for the full year.

#### 2. New Accounting Pronouncements

In May 2014, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2014-09, Revenue from Contracts with Customers (Topic 606), ("ASC Topic 606"), which requires an entity to recognize the amount of revenue to which it expects to be entitled for the transfer of promised goods or services to customers. ASC Topic 606 replaces most existing revenue recognition guidance in U.S. generally accepted accounting principles. In addition, the standard requires disclosure of the nature, amount, timing, and uncertainty of revenue and cash flows arising from the contracts with customers. The Company adopted ASC Topic 606 effective January 1, 2018. See Note 3 for further discussion.

In February 2016, the FASB issued ASU 2016-02, Leases (Topic 842), ("ASU 2016-02"). The purpose of ASU 2016-02 is to provide financial statement users a better understanding of the amount, timing, and uncertainty of cash flows arising from leases. The adoption of ASU 2016-02 will result in the recognition of a right-of-use asset and a lease liability for most operating leases. New disclosure requirements include qualitative and quantitative information about the amounts recorded in the financial statements. ASU 2016-02 is effective for fiscal years beginning after December 15, 2018. ASU 2016-02 requires a modified retrospective transition by means of a cumulative-effect adjustment to retained earnings as of the beginning of the fiscal year in which the guidance is effective with the option to elect certain practical expedients. Early adoption is permitted. In July 2018, the FASB issued ASU 2018-10, Codification Improvements to Topic 842, Leases, ("ASU 2018-10"). The amendments in ASU 2018-10 provide clarification of narrow aspects of the guidance issued in ASU 2016-02. In July 2018, the FASB issued ASU 2018-11, Leases (Topic 842), Targeted Improvements, ("ASU 2018-11"). The amendments in ASU 2018-11 provide an additional transition method to adopt the amendments in ASU 2016-02. Under this new transition method, an entity initially applies the amendments in ASU 2016-02 at the adoption date and recognizes a cumulative-effect adjustment to the opening balance of retained earnings in the period of adoption. Under this transition method, an entity is not required to use the modified transition method described in ASU 2016-02. The Company plans to adopt the additional transition method provided in ASU 2018-11. The Company is currently evaluating the impact of ASU 2016-02, ASU 2018-10 and ASU 2018-11 on its condensed consolidated financial statements. The most significant impact of these amendments will be the recognition of new right-of-use assets and lease liabilities for assets related to leased theatres and certai

In August 2016, the FASB issued ASU 2016-15, Statement of Cash Flows (Topic 230): Classification of Certain Cash Receipts and Cash Payments – a consensus of the FASB Emerging Issues Task Force, ("ASU 2016-15"). The purpose of ASU 2016-15 is to

In thousands, except share and per share data

reduce the diversity in practice regarding how certain cash receipts and cash payments are presented and classified in the statement of cash flows. ASU 2016-15 is effective for fiscal years beginning after December 15, 2017, including interim periods within that year. A retrospective transition method should be used in the application of the amendments within ASU 2016-15. The Company adopted ASU 2016-15 effective January 1, 2018. As a result of the adoption of ASU 2016-15, cash paid of \$246 related to the June 2017 amendment of the Company's senior secured credit facility was reclassified from operating activities to financing activities for the six months ended June 30, 2017.

In October 2016, the FASB issued ASU 2016-16, *Income Taxes (Topic 740): Intra-Entity Transfers of Assets Other Than Inventory*, ("ASU 2016-16"). The purpose of ASU 2016-16 is to improve the accounting for the income tax consequences of intra-entity transfers of assets other than inventory. ASU 2016-16 is effective for fiscal years beginning after December 15, 2017, including interim periods within that year. A modified retrospective transition method should be used in the application of the amendments within ASU 2016-16 with a cumulative-effect adjustment to retained earnings as of the beginning of the period of adoption. ASU 2016-16 did not have a material impact on the Company's condensed consolidated financial statements.

In thousands, except share and per share data

#### 3. Revenue Recognition

Revenue Recognition Policy

The Company recognizes admissions and concession revenues when sales are made at the box office and concession stand, respectively. Other revenues include screen advertising, transactional fees and other ancillary revenues such as vendor marketing promotions and meeting rentals located in the Company's theatres. The Company records proceeds from the sale of gift cards and other advanced sale-type certificates in current liabilities and recognizes admissions or concession revenue when a holder redeems the card or certificate. Additionally, the Company recognizes unredeemed gift cards and other advanced sale-type certificates, as a proportion of actual redemptions, as other revenues, which is an estimate primarily based on the Company's historical experience with such cards and certificates.

Screen advertising revenues are generally recognized over the period that the related advertising is delivered on-screen or in-theatre. Advances collected on long-term screen advertising, concession and other contracts are recorded as deferred revenues. In accordance with the terms of the agreements, the advances collected on such contracts are recognized during the period in which the Company satisfies the related performance obligations, which may differ from the period in which the advances are collected. These advances are recognized on either a straight-line basis over the term of the contracts or as the Company has met its performance obligations in accordance with the terms of the contracts.

See additional revenue recognition policy considerations, updated for the adoption of ASC Topic 606, below.

Adoption of ASC Topic 606

The Company adopted ASC 606, Revenue from Contracts with Customers, effective January 1, 2018 under the modified retrospective method (cumulative-effect) and therefore, revenue amounts as presented on the condensed consolidated statements of income have not been adjusted for prior periods presented.

Changes to the way in which the Company recognizes revenue resulted in the following impacts to the condensed consolidated statements of income:

- a) Recording of incremental other revenue and interest expense related to the significant financing component of the Company's Exhibitor Services Agreement ("ESA") with NCM, LLC ("NCM"). See further discussion below, including the estimated interest rates assumed in determining the amount of interest expense.
- b) Deferral of a portion of admissions and concession revenues for transactions that include the issuance of loyalty points to customers. To determine the amount of revenues to defer upon issuance of points to customers under its points-based loyalty programs, the Company estimated the values of the rewards expected to be redeemed by its customers for those points. The estimates are based on the rewards that have historically been offered under the loyalty programs, which the Company believes is representative of the rewards to be offered in the future.
- c) Increase in other revenues and an increase in utilities and other expenses due to presentation of transactional fees on a gross versus net basis.
- d) Increase in other revenues due to the change in amortization methodology for deferred revenue NCM that is now amortized on a straight-line basis and effective for the entire term of the ESA. As a result of the change in amortization method, the Company recorded a cumulative effect of accounting change adjustment of \$40,526, net of taxes, in retained earnings on January 1, 2018 (see also Note 6).

The above noted changes increased (decreased) admissions, concession and other revenue for the three and six months ended June 30, 2018 as follows:

	Three	Months		
	Er	ıded	Six M	Ionths Ended
	June 3	30, 2018	Ju	ne 30, 2018
Admissions revenues	\$	(1,898)	\$	(3,200)
Concession revenues	\$	(662)	\$	(1,207)
Other revenues	\$	32,710	\$	60,169

The Company applied the practical expedient to exclude sales and other similar taxes collected from customers from its transaction price for purposes of recording revenues, as such revenues are presented net of such taxes.

In thousands, except share and per share data

Disaggregation of Revenue

The following table presents revenues for the three and six months ended June 30, 2018, disaggregated based on major type of good or service and by reportable operating segment.

	Three Months Ended							Six Months Ended							
			J	une 30, 2018			June 30, 2018								
		U.S.	International					U.S.		International					
	•	Operating	Operating					Operating	Operating						
Major Goods/Services	5	Segment (1)		Segment		Consolidated	Segment (1)		Segment		Consolidated				
Admissions revenues	\$	408,863	\$	100,007	\$	508,870	\$	758,215	\$	203,279	\$	961,494			
Concession revenues		249,618		55,688		305,306		453,368		113,710		567,078			
Screen advertising and promotional revenues		21,051		15,446		36,497		39,230		29,715		68,945			
Other revenues		29,542		8,838		38,380		54,604		16,903		71,507			
Total revenues	\$	709,074	\$	179,979	\$	889,053	\$	1,305,417	\$	363,607	\$	1,669,024			

<sup>(1)</sup> U.S. segment revenues include eliminations of intercompany transactions with the international operating segment. See Note 15 for additional information on intercompany eliminations.

The following table presents revenues for the three and six months ended June 30, 2018, disaggregated based on timing of revenue recognition (see Revenue Recognition Policy above).

	Three Months Ended June 30, 2018					Six Months Ended June 30, 2018							
		U.S. perating egment (1)		ternational Operating Segment	Consolidated		U.S. Operating Segment (1)		International Operating Segment		C	onsolidated	
Goods and services transferred at a point in time		686,463	\$	161,350	\$	847,813	\$	1,262,289	\$	327,499	\$	1,589,788	
Goods and services transferred over time		22,611		18,629		41,240		43,128		36,108		79,236	
Total	\$	709,074	\$	179,979	\$	889,053	\$	1,305,417	\$	363,607	\$	1,669,024	

<sup>(1)</sup> U.S. segment revenues include eliminations of intercompany transactions with the international operating segment. See Note 15 for additional information on intercompany eliminations.

#### Deferred Revenues

The following table presents changes in the Company's deferred revenues for the six months ended June 30, 2018.

		D	eferred	Othe	er Deferred	
- 2	Deferred Revenues	Reve	nue - NCM	Re	venues (1)	Total
	Balance at January 1, 2018	\$	351,706	\$	86,498	\$ 438,204
	Impact of adoption of ASC Topic 606		(53,605)		_	(53,605)
	Amounts recognized as accounts receivable		_		11,349	11,349
	Cash received from customers in advance		_		53,398	53,398
	Common units received from NCM (see Note 7)		5,012		_	5,012
	Revenue recognized during period		(7,849)		(64,996)	(72,845)
	Foreign currency translation adjustments		_		(1,689)	(1,689)
	Balance at June 30, 2018	\$	295,264	\$	84,560	\$ 379,824

In thousands, except share and per share data

(1) Includes liabilities associated with outstanding gift cards and SuperSavers, points or rebates outstanding under the Company's loyalty and membership programs and revenues not yet recognized for screen advertising and other promotional activities. Classified as accounts payable and accrued expenses or other long-term liabilities on the condensed consolidated balance sheet

The table below summarizes the aggregate amount of the transaction price allocated to performance obligations that are unsatisfied as of June 30, 2018 and when the Company expects to recognize this revenue.

		Twelve Months Ended June 30,											
Remaining Performance Obligations	2019		2020		2021		2022		2023		Thereafter		Total
Deferred revenue - NCM	\$	15,831	\$	15,831	\$	15,831	\$	15,831	\$	15,831	\$	216,109	\$ 295,264
Deferred revenue - other		72,053		12,039		282		186		_		_	84,560
Total	\$	87,884	\$	27,870	\$	16,113	\$	16,017	\$	15,831	\$	216,109	\$ 379,824

Accounts receivable as of June 30, 2018 included approximately \$48,631 of receivables related to contracts with customers. The Company did not record any assets related to the costs to obtain or fulfill a contract with customers during the six months ended June 30, 2018.

Significant Financing Component

As discussed further in Note 7, in connection with the completion of the NCM, Inc. ("NCMI") initial public offering, the Company amended and restated its ESA with NCM and received approximately \$174,000 in cash consideration from NCM. The proceeds were recorded as deferred revenue and are being amortized over the term of the modified ESA, or through February 2037. In addition to the consideration received upon the ESA modification during 2007, the Company also receives consideration in the form of common units from NCM, at each annual common unit adjustment settlement, in exchange for exclusive access to the Company's newly opened domestic screens under the ESA. See Note 7 for additional information regarding the common unit adjustment and related accounting. Due to the significant length of time between receiving the consideration from NCM and fulfillment of the related performance obligation, the ESA includes an implied significant financing component, as per the guidance in ASC Topic 606.

As a result of the significant financing component on deferred revenue - NCM, the Company recognized incremental screen advertising revenue and an offsetting interest expense of \$4,913 and \$9,892 during the three and six months ended June 30, 2018, respectively. The interest expense was calculated using the Company's incremental borrowing rates at the time when the cash and each tranche of common units were received from NCM, which ranged from 5.5% to 8.0%.

#### 4. Earnings Per Share

The Company considers its unvested restricted stock awards, which contain non-forfeitable rights to dividends, participating securities, and includes such participating securities in its computation of earnings per share pursuant to the two-class method. Basic earnings per share for the two classes of stock (common stock and unvested restricted stock) is calculated by dividing net income by the weighted average number of shares of common stock and unvested restricted stock outstanding during the reporting period. Diluted earnings per share is calculated using the weighted average number of shares of common stock plus the potentially dilutive effect of common equivalent shares outstanding determined under both the two class method and the treasury stock method.

In thousands, except share and per share data

The following table presents computations of basic and diluted earnings per share under the two-class method:

		Three Months Ended June 30,						led	
		2018		2017		2018		2017	
Numerator:									
Net income attributable to Cinemark Holdings, Inc.	\$	82,135	\$	51,239	\$	144,156	\$	130,967	
Earnings allocated to participating share-based awards (1)		(465)		(263)		(817)		(620)	
Net income attributable to common stockholders	\$	81,670	\$	50,976	\$	143,339	\$	130,347	
Denominator (shares in thousands):									
Basic weighted average common stock outstanding		116,091		115,785		115,988		115,707	
Common equivalent shares for restricted stock units		177		287		250		313	
Diluted	_	116,268		116,072	_	116,238	_	116,020	
Basic earnings per share attributable to common stockholders	\$	0.70	\$	0.44	\$	1.23	\$	1.12	
Diluted earnings per share attributable to common stockholders	\$	0.70	\$	0.44	\$	1.23	\$	1.12	

<sup>(1)</sup> For the three months ended June 30, 2018 and 2017, a weighted average of approximately 676 and 605 shares of unvested restricted stock, respectively, were considered participating securities. For the six months ended June 30, 2018 and 2017, a weighted average of approximately 663 and 552 shares of restricted stock, respectively, were considered participating securities.

#### 5. Long Term Debt Activity

Senior Secured Credit Facility

On March 29, 2018, Cinemark USA, Inc., our wholly-owned subsidiary, amended its senior secured credit facility to extend the maturity of the term loan to March 29, 2025, reduce the rate at which the term loan bears interest by 0.25% and to reduce the amount of real property required to be mortgaged to secure the loans. Under the amended facility, quarterly principal payments of \$1,649 are due on the term loan through December 31, 2024, with a final principal payment of \$613,351 due on March 29, 2025. The Company incurred debt issue costs of approximately \$4,962 in connection with the amendment, which are reflected as a reduction of long term debt on the condensed consolidated balance sheet as of June 30, 2018. As a result of the amendment, the Company wrote-off \$780 of unamortized debt issue costs and incurred approximately \$704 in legal and other fees, both of which are reflected as loss on debt amendments on the condensed consolidated statements of income for the six months ended June 30, 2018.

Fair Value of Long-Term Debt

The Company estimates the fair value of its long-term debt using the market approach, which utilizes quoted market prices that fall under Level 2 of the U.S. GAAP fair value hierarchy as defined by ASC 820, *Fair Value Measurement* ("ASC Topic 820"). The carrying value of the Company's long-term debt was \$1,813,997 and \$1,817,295 as of June 30, 2018 and December 31, 2017, respectively, excluding unamortized debt discounts and debt issue costs. The fair value of the Company's long-term debt was \$1,800,830 and \$1,840,918 as of June 30, 2018 and December 31, 2017, respectively.

In thousands, except share and per share data

#### 6. Equity

Below is a summary of changes in stockholders' equity attributable to Cinemark Holdings, Inc., noncontrolling interests and total equity for the six months ended June 30, 2018 and 2017:

	Cinemark Holdings, Inc.			
	Stockholders' Equity	1	Noncontrolling Interests	Total Equity
Balance at January 1, 2018	\$ 1,393,79	5 \$	11,893	\$ 1,405,688
Cumulative effect of change in accounting principle, net of taxes of \$13,079 (see Note 3)	40,52	26	_	40,526
Share based awards compensation expense	6,87	78	_	6,878
Stock withholdings related to share based awards that vested during the six months ended June 30, 2018	(2,90	05)	_	(2,905)
Dividends paid to stockholders(1)	(74,72	23)	_	(74,723)
Dividends accrued on unvested restricted stock unit awards(1)	(27	1)	_	(271)
Net income	144,15	56	485	144,641
Other comprehensive income in equity method investees	2	20	_	20
Foreign currency translation adjustments	(56,04	<u> </u>	_	(56,044)
Balance at June 30, 2018	\$ 1,451,43	\$2	12,378	\$ 1,463,810

(1) The Company's board of directors declared a cash dividend for the fourth quarter of 2017 in the amount of \$0.32 per share of common stock payable to stockholders of record on March 8, 2018. The dividend was paid on March 22, 2018. The Company's board of directors declared a cash dividend for the first quarter of 2018 in the amount of \$0.32 per share of common stock payable to stockholders of record on June 8, 2018. The dividend was paid on June 22, 2018.

	Н	Cinemark oldings, Inc. ockholders' Equity	Noncontrolling Interests	Total Equity
Balance at January 1, 2017	\$	1,261,818	\$ 11,142	\$ 1,272,960
Share based awards compensation expense		6,444	_	6,444
Stock withholdings related to share based awards that vested during the six months ended June 30, 2017		(2,921)	_	(2,921)
Dividends paid to stockholders (2)		(67,528)	_	(67,528)
Dividends accrued on unvested restricted stock unit awards (2)		(288)	_	(288)
Dividends paid to noncontrolling interests		` <u>—</u>	(294)	(294)
Net income		130,967	1,037	132,004
Other comprehensive income in equity method investees		103	_	103
Foreign currency translation adjustments		(5,059)	<u></u>	(5,059)
Balance at June 30, 2017	\$	1,323,536	\$ 11,885	\$ 1,335,421

<sup>(2)</sup> The Company's board of directors declared a cash dividend for the fourth quarter of 2016 in the amount of \$0.29 per share of common stock payable to stockholders of record on March 8, 2017. The dividend was paid on March 20, 2017. The Company's board of directors declared a cash dividend for the first quarter of 2017 in the amount of \$0.29 per share of common stock payable to stockholders of record on June 8, 2017. The dividend was paid on June 22, 2017.

#### 7. Investment in National CineMedia

The Company has an investment in National Cinemedia, LLC ("NCM"). NCM operates a digital in-theatre network in the U.S. for providing cinema advertising. Upon joining NCM, the Company entered into an Exhibitor Services Agreement with NCM ("ESA"), pursuant to which NCM provides advertising and promotions to our theatres. As described further in Note 5 to the Company's financial statements as included in its 2017 Annual Report on Form 10-K, on February 13, 2007, National Cinemedia, Inc.

In thousands, except share and per share data

("NCMI"), an entity that serves as the sole manager of NCM, completed an initial public offering ("IPO") of its common stock. In connection with the NCMI initial public offering, the Company amended its operating agreement and the ESA. Following the NCM, Inc. IPO, the Company does not recognize undistributed equity in the earnings on its original NCM membership units (referred to herein as the Company's Tranche 1 Investment) until NCM's future net earnings, less distributions received, surpass the amount of the excess distribution. The Company recognizes equity in earnings on its Tranche 1 Investment only to the extent it receives cash distributions from NCM. The Company recognizes cash distributions it receives from NCM on its Tranche 1 Investment as a component of earnings as Distributions from NCM. The Company believes that the accounting model provided by ASC Topic 323-10-35-22 for recognition of equity investee losses in excess of an investor's basis is analogous to the accounting for equity income subsequent to recognizing an excess distribution.

Below is a summary of activity with NCM included in the Company's condensed consolidated financial statements:

	Inv	estment in NCM	Deferred Revenue	_	istributions rom NCM	quity in arnings	]	Other Revenue	E	nterest xpense - CM (3)	R	Cash Received
Balance as of January 1, 2018	\$	200,550	\$ (351,706)									
Impact of adoption of ASC Topic 606 (1)		_	53,605	\$	_	\$ _	\$	_	\$	_	\$	_
Receipt of common units due to annual common												
unit adjustment ("CUA")		5,012	(5,012)		_	_		_		_		_
Revenues earned under ESA (2)(3)		_	_		_	_		(16,074)		9,892		6,182
Receipt of excess cash distributions		(8,760)	_		(7,735)	_		_		_		16,495
Receipt under tax receivable agreement		(2,294)	_		(2,047)	_		_		_		4,341
Equity in earnings		4,511	_		_	(4,511)		_		_		_
Amortization of deferred revenue		_	7,849		_	_		(7,849)		_		_
Balance as of and for the six months ended June 30, 2018	\$	199,019	\$ (295,264)	\$	(9,782)	\$ (4,511)	\$	(23,923)	\$	9,892	\$	27,018

- (1) As a result of adoption of ASC Topic 606, the Company determined that the deferred revenue associated with the ESA and Common Unit Adjustment agreement should be amortized on a straight-line basis versus the units of revenue method followed prior to adoption. The Company recorded a reduction in the deferred revenue balance and a cumulative effect of a change in accounting principle in retained earnings (see also Note 6). See Note 3 for further discussion of the impact of the adoption of ASC Topic 606.
- (2) Amount includes the per patron and per digital screen theatre access fees due to the Company, net of amounts paid to NCM for on-screen advertising time provided to the Company's beverage concessionaire of approximately \$6,227.
- (3) Reflects impact of significant financing component related to amounts received in advance under the ESA and CUA agreements. See Note 3.

During the three months ended June 30, 2018 and 2017, the Company recorded equity in earnings of \$495 and \$176, respectively. During the six months ended June 30, 2018 and 2017, the Company recorded equity in earnings of \$4,511 and \$3,065, respectively.

The Company made payments to NCM of \$60 and \$50 during the six months ended June 30, 2018 and 2017, respectively, related to installation of certain equipment used for digital advertising, which is included in theatre properties and equipment on the condensed consolidated balance sheets.

Pursuant to a Common Unit Adjustment Agreement dated as of February 13, 2007 between NCMI and the Company, AMC Entertainment, Inc. ("AMC") and Regal Entertainment Group ("Regal") (collectively, "Founding Members"), annual adjustments to the common membership units are made primarily based on increases or decreases in the number of theatre screens operated and theatre attendance generated by each Founding Member. As further discussed in Note 5 to the Company's financial statements as included in its 2017 Annual Report on Form 10-K, the common units received (collectively referred to as the Company's "Tranche 2 Investment") are recorded at estimated fair value as an increase in the Company's investment in NCM with an offset to deferred revenue. The Company's Tranche 2 Investment is accounted for following the equity method, with undistributed equity earnings related to its Tranche 2 Investment are recorded as a reduction of investment basis.

During March 2018, NCM performed its annual common unit adjustment calculation under the Common Unit Adjustment Agreement. As a result of the calculation, on March 29, 2018, the Company received an additional 908,042 common units of NCM, each of which is convertible into one share of NCMI common stock. The Company recorded the additional common units received at estimated fair value with a corresponding adjustment to deferred revenue of approximately \$5,012. The fair value of the common units received was estimated based on the market price of NCMI common stock at the time the common units were determined, adjusted for

In thousands, except share and per share data

volatility associated with the estimated time period it would take to convert the common units and register the respective shares. The deferred revenue will be recognized on a straight-line basis over the remaining term of the ESA, which is approximately 19 years.

As of June 30, 2018, the Company owned a total of 28,779,904 common units of NCM, representing an ownership interest of approximately 18%. The estimated fair value of the Company's investment in NCM was approximately \$241,751 based on the price of NCMI common stock as of June 30, 2018 of \$8.40 per share (Level 1 input as defined in FASB ASC Topic 820). See Note 18 for discussion of subsequent event related to the acquisition of additional NCM common units from AMC.

Below is summary financial information for NCM for the periods indicated:

	Three Mor	ths E	Inded		Six Months Ended					
	 June 28, 2018	June 29, 2017			June 28, 2018	June 29, 2017				
Gross revenues	\$ 113,700	\$	97,100	\$	193,900	\$	169,000			
Operating income	\$ 40,100	\$	28,300	\$	51,100	\$	33,400			
Net income	\$ 25,000	\$	15,400	\$	22,000	\$	7,500			

	As	As of				
	June 28	3, 2018	December 28, 2017			
Current assets	\$	142,800	\$	174,400		
Noncurrent assets	\$	758,800	\$	758,300		
Current liabilities	\$	84,800	\$	123,300		
Noncurrent liabilities	\$	943,500	\$	925,400		
Members deficit	\$	(126,700)	\$	(116,000)		

#### 8. Other Investments

Below is a summary of activity for each of the Company's other investments for the six months ended June 30, 2018:

		AC JV,				
	DCIP	LLC	DCDC	FE Concepts	Other	Total
Balance at January 1, 2018	\$ 106,215	\$ 5,916	\$ 3,598	\$ 104	\$ 4,212	\$ 120,045
Cash contributions made	739	_	_	20,000		20,739
Cash distributions received	(5,201)	_	_	_	_	(5,201)
Equity in income (loss)	9,727	408	481	(77)	_	10,539
Equity in other comprehensive income	20	_	_	_	_	20
Other	_		(1,168)	(104)	211	(1,061)
Balance at June 30, 2018	\$ 111,500	\$ 6,324	\$ 2,911	\$ 19,923	\$ 4,423	\$ 145,081

Digital Cinema Implementation Partners LLC ("DCIP")

On February 12, 2007, the Company, AMC and Regal entered into a joint venture known as DCIP to facilitate the implementation of digital cinema in the Company's theatres and to establish agreements with major motion picture studios for the financing of digital cinema. As of June 30, 2018, the Company had a 33% voting interest in DCIP and a 24.3% economic interest in DCIP. The Company accounts for its investment in DCIP and its subsidiaries under the equity method of accounting.

Below is summary financial information for DCIP for the periods indicated:

		Three Months Ended June 30, 2018 June 30, 2017				Six Months Ended						
	June	June 30, 2018			Jur	ie 30, 2018	June 30, 2017					
Gross revenues	\$	42,521	\$	47,095	\$	83,554	\$	92,574				
Operating income	\$	25,049	\$	29,393	\$	48,494	\$	57,872				
Net income	\$	22,169	\$	25,616	\$	43,703	\$	49,757				
			16									

In thousands, except share and per share data

		As of								
	June	June 30, 2018								
Current assets	\$	51,546	\$	56,296						
Noncurrent assets	\$	725,399	\$	772,438						
Current liabilities	\$	63,982	\$	59,153						
Noncurrent liabilities	\$	213,795	\$	296,889						
Members' equity	\$	499,168	\$	472,692						

As of June 30, 2018, the Company had 3,818 digital projection systems being leased under the master equipment lease agreement with Kasima LLC, which is an indirect subsidiary of DCIP and a related party to the Company. The Company had the following transactions with DCIP, reflected in utilities and other costs on the condensed consolidated statements of income, during the three and six months ended June 30, 2018 and 2017:

		Three Months Ended				Six Months Ended				
	Jui	ne 30, 2018	Ju	ine 30, 2017	J	une 30, 2018	June 30, 2017			
Equipment lease payments	\$	1,243	\$	1,511	\$	2,460	\$	2,881		
Warranty reimbursements from DCIP	\$	(2,617)	\$	(2,023)	\$	(5,118)	\$	(3,907)		
Management service fees	\$	184	\$	206	\$	378	\$	412		

AC JV, LLC

During December 2013, the Company, Regal, AMC (the "AC Founding Members") and NCM entered into a series of agreements that resulted in the formation of AC JV, LLC ("AC"), a joint venture that owns "Fathom Events" formerly operated by NCM. The Fathom Events business focuses on the marketing and distribution of live and pre-recorded entertainment programming to various theatre operators, including concerts, opera and symphony, DVD product releases and marketing events, theatrical premieres, Broadway plays, live sporting events and other special events. The Company paid event fees to AC of \$5,588 and \$6,763 for the six months ended June 30, 2018 and 2017, respectively, which are included in film rentals and advertising costs on the condensed consolidated statements of income. Additionally, the remaining outstanding balance of a note payable from the Company to NCM, related to the formation of AC, was \$2,778 as of June 30, 2018.

#### Digital Cinema Distribution Coalition

Digital Cinema Distribution Coalition ("DCDC") is a joint venture among the Company, Universal, Warner Bros., AMC and Regal. DCDC operates a satellite distribution network that distributes all digital content to U.S. theatres via satellite. The Company has an approximate 14.6% ownership in DCDC. The Company paid approximately \$461 and \$446 to DCDC during the six months ended June 30, 2018 and 2017, respectively, related to content delivery services provided by DCDC. These fees are included in film rentals and advertising costs on the condensed consolidated statements of income.

#### FE Concepts, LLC

During April 2018, the Company, through its wholly-owned indirect subsidiary CNMK Texas Properties, LLC ("CNMK"), formed a joint venture, FE Concepts, LLC ("FE Concepts") with AWSR Investments, LLC ("AWSR"), an entity owned by Lee Roy Mitchell and Tandy Mitchell. FE Concepts will develop and operate a family entertainment center that offers bowling, gaming, movies and other amenities. The Company and AWSR each invested approximately \$20,000 and each have a 50% voting interest in FE Concepts. The Company accounts for its investment in FE Concepts under the equity method of accounting.

In thousands, except share and per share data

#### 9. Treasury Stock and Share Based Awards

Treasury Stock — Treasury stock represents shares of common stock repurchased or withheld by the Company and not yet retired. The Company has applied the cost method in recording its treasury shares. Below is a summary of the Company's treasury stock activity for the six months ended June 30, 2018:

	Number of	
	Treasury	
	Shares	Cost
Balance at January 1, 2018	4,525,870	\$ 76,354
Restricted stock withholdings (1)	75,801	2,905
Restricted stock forfeitures	18,643	
Balance at June 30, 2018	4,620,314	\$ 79,259

<sup>(1)</sup> The Company withheld restricted shares as a result of the election by certain employees to satisfy their tax liabilities upon vesting in restricted stock and restricted stock units. The Company determined the number of shares to be withheld based upon market values ranging from \$33.77 to \$39.03 per share.

As of June 30, 2018, the Company had no plans to retire any shares of treasury stock.

Restricted Stock – During the six months ended June 30, 2018, the Company granted 321,406 shares of restricted stock to employees and directors. The fair value of the restricted stock granted was determined based on the market value of the Company's common stock on the dates of grant, which ranged from \$35.81 to \$39.26 per share. The Company assumed forfeiture rates ranging from 0% to 10% for the restricted stock awards. The restricted stock awards vest over periods ranging from one to four years. The recipients of restricted stock are entitled to receive non-forfeitable dividends and to vote their respective shares, however, the sale and transfer of the restricted shares is prohibited during the restriction period.

Below is a summary of restricted stock activity for the six months ended June 30, 2018:

	Shares of Restricted Stock	Weighted Average Grant Date Fair Value					
Outstanding at January 1, 2018	650,581	\$ 35.	.81				
Granted	321,406	\$ 38.	.79				
Vested	(242,945)	\$ 31.	.16				
Forfeited	(18,643)	\$ 38.	.33				
Outstanding at June 30, 2018	710,399	\$ 38.	.68				
Unvested restricted stock at June 30, 2018	710,399	\$ 38.	.68				
	Six Months Ended June 30,						
	2018	2017					

		June 30,						
	2	018	2017					
Compensation expense recognized during the period	\$	4,798	\$	4,294				
Fair value of restricted shares that vested during the period	\$	7,815	\$	8,091				
Income tax benefit recognized upon vesting of restricted stock awards	\$	1,751	\$	2,633				

As of June 30, 2018, the estimated remaining unrecognized compensation expense related to unvested restricted stock awards was \$19,408 and the weighted average period over which this remaining compensation expense will be recognized is approximately two years.

Restricted Stock Units – During the six months ended June 30, 2018, the Company granted restricted stock units representing 228,194 hypothetical shares of common stock to employees. The restricted stock units vest based on a combination of financial performance factors and continued service. The financial performance factors are based on an implied equity value concept that determines an internal rate of return ("IRR") during the two fiscal year periods ending December 31, 2019 based on a formula utilizing a multiple of Adjusted EBITDA subject to certain adjustments as specified by the Compensation Committee prior to the grant date. The financial performance factors for the restricted stock units have a threshold, target and maximum level of payment

In thousands, except share and per share data

opportunity and vest on a prorata basis according to the IRR achieved by the Company during the performance period. If the IRR for the two-year period is at least 7%, which is the threshold, one-third of the maximum restricted stock units vest. If the IRR for the two-year period is at least 9.5%, which is the target, two-thirds of the maximum restricted stock units vest. If the IRR for the two-year period is at least 13%, which is the maximum, 100% of the maximum restricted stock units vest. Grantees are eligible to receive a ratable portion of the common stock issuable if the IRR is within the targets previously noted. Further, as an example, if the Company achieves an IRR equal to 11%, the number of restricted stock units that shall vest will be greater than the target but less than the maximum number that would have vested had the Company achieved the highest IRR. All restricted stock units granted during 2018 will vest subject to an additional two-year service requirement and will be paid in the form of common stock if the participant continues to provide services through the fourth anniversary of the grant date. Restricted stock unit award participants are eligible to receive dividend equivalent payments from the grant date if, and at the time that, the restricted stock unit awards vest.

Below is a table summarizing the potential number of shares that could vest under restricted stock unit awards granted during the six months ended June 30, 2018at each of the three target levels of financial performance (excluding forfeiture assumptions):

	Number of			
	Shares	Value at		
	Vesting	Grant		
at IRR of at least 7%	76,065	\$ 2,967		
at IRR of at least 9.5%	152,129	\$ 5,934		
at IRR of at least 13%	228,194	\$ 8,901		

Due to the fact that the IRR for the two-year performance period could not be determined at the time of the 2018 grant, the Company estimated that the most likely outcome is the achievement of the target IRR level. The fair value of the restricted stock unit awards was determined based on the closing price of the Company's common stock on the dates of grant, which ranged from \$37.55 to \$39.03 per share. The Company assumed forfeiture rates ranging from 0% to 5% for the restricted stock unit awards. If during the service period, additional information becomes available to lead the Company to believe a different IRR level will be achieved for the two-year performance period, the Company will reassess the number of units that are expected to vest for the grant and adjust its compensation expense accordingly on a prospective basis over the remaining service period.

		Six Months Ended June 30,				
	2018			2017		
Number of restricted stock unit awards that vested during the period		127,084		97,115		
Fair value of restricted stock unit awards that vested during the period	\$	4,846	\$	4,155		
Accumulated dividends paid upon vesting of restricted stock unit awards	\$	501	\$	313		
Compensation expense recognized during the period	\$	2,080	\$	2,150		
Income tax benefit recognized upon vesting of restricted stock unit awards	\$	740	\$	1,745		

During the six months ended June 30, 2018, the Company determined that the IRR reached for the restricted stock units granted in February 2016 was 7.2%, compared to a target IRR of 8.0%. The Company recorded a reduction in compensation expense of \$69 during the six months ended June 30, 2018 to reflect the revised number of shares expected to vest in February 2020 for the 2016 restricted stock unit grant.

As of June 30, 2018, the estimated remaining unrecognized compensation expense related to the outstanding restricted stock unit awards was \$9,854. The weighted average period over which this remaining compensation expense will be recognized is approximately two years. As of June 30, 2018, the Company had restricted stock units outstanding that represented a total of 594,266 hypothetical shares of common stock, net of actual cumulative forfeitures of 18,667 units, assuming an IRR of 7.2% was achieved for the 2016 grants and the maximum IRR level is achieved for all other grants outstanding.

In thousands, except share and per share data

#### 10. Goodwill and Other Intangible Assets

The Company's goodwill was as follows:

	U.S. Operating Segment	(	ternational Operating Segment	Total
Balance at January 1, 2018(1)	\$ 1,174,041	\$	110,038	\$ 1,284,079
Acquisition of theatres in Brazil (2)	_		11,508	11,508
Foreign currency translation adjustments			(14,417)	(14,417)
Balance at June 30, 2018(1)	\$ 1,174,041	\$	107,129	\$ 1,281,170

- (1) Balances are presented net of accumulated impairment losses of \$214,031 for the U.S. operating segment and \$27,622 for the international operating segment.
- (2) Amount represents preliminary purchase price allocation for theatres acquired in Brazil.

The Company evaluates goodwill for impairment annually during the fourth quarter or whenever events or changes in circumstances indicate the carrying value of the goodwill may not be fully recoverable. The Company evaluates goodwill for impairment at the reporting unit level and has allocated goodwill to the reporting unit based on an estimate of its relative fair value. Management considers the reporting unit to be each of its nineteen regions in the U.S. and seven countries internationally with Honduras, El Salvador, Nicaragua, Costa Rica, Panama and Guatemala considered one reporting unit (the Company does not have goodwill recorded for all of its international locations). For the year ended December 31, 2017, the Company performed a quantitative goodwill impairment assessment on all reporting units, in accordance with ASC Topic 350-20-35. No events or changes in circumstances occurred during the six months ended June 30, 2018 that indicated the carrying value of goodwill might exceed its estimated fair value.

Intangible assets consisted of the following:

	_	Balance at January 1,				Balance at June 30,
		2018	Additions (1)	Amortization	Other (2)	2018
Intangible assets with finite lives:						
Gross carrying amount	\$	105,895	\$ _	\$ _	\$ (1,645)	\$ 104,250
Accumulated amortization		(68,869)	_	(2,932)	_	(71,801)
Total net intangible assets with finite lives	\$	37,026	\$ 	\$ (2,932)	\$ (1,645)	\$ 32,449
Intangible assets with indefinite lives:						
Tradename and other		299,735	608	_	(286)	300,057
Total intangible assets — net	\$	336,761	\$ 608	\$ (2,932)	\$ (1,931)	\$ 332,506

- (1) Amount represents the acquisition of tradeable liquor licenses.
- (2) Amount represents foreign currency translation adjustments.

For the year ended December 31, 2017, the Company performed a quantitative assessment for all definite and indefinite-lived tradename assets. No events or changes in circumstances occurred during the six months ended June 30, 2018 that indicated the carrying value of its tradename assets might exceed their estimated fair values.

In thousands, except share and per share data

Estimated aggregate future amortization expense for intangible assets is as follows:

For the six months ended December 31, 2018	\$ 2,661
For the twelve months ended December 31, 2019	5,152
For the twelve months ended December 31, 2020	5,310
For the twelve months ended December 31, 2021	3,574
For the twelve months ended December 31, 2022	3,207
Thereafter	12,545
Total	\$ 32,449

#### 11. Impairment of Long-Lived Assets

The Company reviews long-lived assets for impairment indicators on a quarterly basis or whenever events or changes in circumstances indicate the carrying amount of the assets may not be fully recoverable. See discussion of the Company's long-lived asset impairment evaluation process in Note 1 to the Company's financial statements as included in its 2017 Annual Report on Form 10-K. As noted in the discussion, fair value is determined based on a multiple of cash flows, which was six and a half times for the evaluations performed during the six months ended June 30, 2018 and 2017. As of June 30, 2018, the estimated aggregate fair value of the long-lived assets impaired during the six months ended June 30, 2018 was \$871.

The long-lived asset impairment charges recorded during each of the periods presented are specific to theatres that were directly and individually impacted by increased competition, adverse changes in market demographics or adverse changes in the development or the conditions of the areas surrounding the theatre.

Below is a summary of impairment charges for the periods presented:

	Three Months Ended June 30,					Six Months Ended June 30,			
		2018		2017		2018		2017	
U.S. theatre properties	\$	1,187	\$	84	\$	1,778	\$	357	
International theatre properties		1,601		4,217		1,601		4,217	
Impairment of long-lived assets	\$	2,788	\$	4,301	\$	3,379	\$	4,574	

#### 12. Fair Value Measurements

The Company determines fair value measurements in accordance with ASC Topic 820, which establishes a fair value hierarchy under which an asset or liability is categorized based on the lowest level of input significant to its fair value measurement. The levels of input defined by ASC Topic 820 are as follows:

- Level 1 quoted market prices in active markets for identical assets or liabilities that are accessible at the measurement date;
- Level 2 other than quoted market prices included in Level 1 that are observable for the asset or liability, either directly or indirectly; and
- Level 3 unobservable and should be used to measure fair value to the extent that observable inputs are not available.

The Company did not have any assets or liabilities measured at fair value on a recurring basis under ASC Topic 820 as of December 31, 2017 or June 30, 2018.

The Company uses the market approach for fair value measurements on a nonrecurring basis in the impairment evaluations of its long-lived assets (see Note 10 and Note 11). See additional explanation of fair value measurement techniques used for long-lived assets, goodwill and intangible assets in "Critical Accounting Policies" in the Company's Annual Report on Form 10-K for the year ended December 31, 2017, filed February 23, 2018. There were no changes in valuation techniques and there were no transfers in or out of Level 1, Level 2 or Level 3 during the six months ended June 30, 2018.

In thousands, except share and per share data

#### 13. Foreign Currency Translation

The accumulated other comprehensive loss account in stockholders' equity of \$309,306 and \$253,282 as of June 30, 2018 and December 31, 2017, respectively, primarily includes cumulative foreign currency adjustments of \$309,609 and \$253,565, respectively, from translating the financial statements of the Company's international subsidiaries.

As of June 30, 2018, all foreign countries where the Company has operations are non-highly inflationary, and the local currency is the same as the functional currency in all of the locations. Thus, any fluctuation in the currency results in a cumulative foreign currency translation adjustment recorded to accumulated other comprehensive loss.

A highly inflationary economy is defined as an economy with a cumulative inflation rate of approximately 100 percent or more over a three-year period. If a country's economy is classified as highly inflationary, the financial statements of the foreign entity operating in that country must be remeasured to the functional currency of the reporting entity. There has been a steady devaluation of the Argentine peso relative to the U.S. dollar in recent years. As of June 30, 2018, the Company did not designate Argentina as highly inflationary for accounting purposes. However, the Company has determined that Argentina will be deemed highly inflationary beginning July 1, 2018. The financial information of the Company's Argentina subsidiaries will be remeasured in U.S. dollars in accordance with ASC Topic 830, Foreign Currency Matters, effective July 1, 2018.

Below is a summary of the impact of translating the June 30, 2018 financial statements of the Company's international subsidiaries:

			U	ther
			Comp	rehensive
			Income (I	Loss) for the
	Exchange R	ate as of	Six Mon	ths Ended
Country	June 30, 2018	December 31, 2017	June	30, 2018
Brazil	3.88	3.31	\$	(34,511)
Argentina	28.94	18.65		(16,896)
Peru	3.28	3.24		(580)
Chile	654.28	615.97		(4,372)
All other				315
			\$	(56,044)

#### 14. Supplemental Cash Flow Information

The following is provided as supplemental information to the condensed consolidated statements of cash flows:

	Six Months Ended June 30,						
	:	2018	2017				
Cash paid for interest	\$	51,562	\$	49,603			
Cash paid for income taxes, net of refunds received	\$	28,477	\$	55,371			
Noncash investing and financing activities:							
Change in accounts payable and accrued expenses for the							
acquisition of theatre properties and equipment (1)	\$	2,898	\$	(4,959)			
Theatre properties acquired under capital lease	\$	4,035	\$	7,089			
Interest expense - NCM (see Note 3)	\$	(9,892)	\$	_			
Investment in NCM – receipt of common units (see Note 7)	\$	5,012	\$	18,363			
Dividends accrued on unvested restricted stock unit awards	\$	(271)	\$	(288)			

<sup>(1)</sup> Additions to theatre properties and equipment included in accounts payable as of June 30, 2018 and December 31, 2017 were \$34,174 and \$31,276, respectively.

In thousands, except share and per share data

#### 15. Segments

The Company manages its international market and its U.S. market as separate reportable operating segments, with the international segment consisting of operations in Brazil, Argentina, Chile, Colombia, Peru, Ecuador, Honduras, El Salvador, Nicaragua, Costa Rica, Panama, Guatemala, Bolivia, Curacao and Paraguay. Each segment's revenue is derived from admissions and concession sales and other ancillary revenues. The Company uses Adjusted EBITDA, as shown in the reconciliation table below, as the primary measure of segment profit and loss to evaluate performance and allocate its resources. The Company does not report total assets by segment because that information is not used to evaluate the performance of or allocate resources between segments.

Below is a breakdown of selected financial information by reportable operating segment:

	Three Months Ended June 30,				led			
		2018		2017		2018		2017
Revenues								
U.S.	\$	712,483	\$	554,929	\$	1,312,128	\$	1,136,138
International		179,979		199,926		363,607		401,994
Eliminations		(3,409)		(3,660)		(6,711)		(7,327)
Total revenues	\$	889,053	\$	751,195	\$	1,669,024	\$	1,530,805
Adjusted EBITDA								
U.S.	\$	188,411	\$	129,394	\$	344,255	\$	294,048
International		33,192		41,285		70,778		88,511
Total Adjusted EBITDA	\$	221,603	\$	170,679	\$	415,033	\$	382,559
Capital expenditures								
U.S.	\$	59,675	\$	77,175	\$	129,646	\$	155,992
International		22,751		14,438		32,943		26,808
Total capital expenditures	\$	82,426	\$	91,613	\$	162,589	\$	182,800

The following table sets forth a reconciliation of net income to Adjusted EBITDA:

	Three Months Ended June 30,				led			
		2018		2017		2018		2017
Net income	\$	82,464	\$	51,810	\$	144,641	\$	132,004
Add (deduct):								
Income taxes		18,326		29,445		43,423		73,845
Interest expense (1)		28,466		26,522		55,581		52,891
Other (income) expense (2)		836		(7,030)		(6,437)		(20,012)
Loss on debt amendments		_		246		1,484		246
Other cash distributions from equity investees (3)		3,932		2,870		16,255		14,919
Depreciation and amortization		64,290		59,137		128,685		116,493
Impairment of long-lived assets		2,788		4,301		3,379		4,574
Loss on disposal of assets and other		16,901		54		20,840		888
Deferred lease expenses		(449)		(375)		(932)		(722)
Amortization of long-term prepaid rents		597		496		1,236		989
Share based awards compensation expense		3,452		3,203		6,878		6,444
Adjusted EBITDA (4)	\$	221,603	\$	170,679	\$	415,033	\$	382,559

Includes amortization of debt issue costs.

<sup>(2)</sup> Includes interest income, foreign currency exchange gain (loss), equity in income of affiliates and interest expense - NCM and excludes distributions from NCM.

<sup>(3)</sup> Includes cash distributions received from equity investees that were recorded as a reduction of the respective investment balances (see Notes 7 and 8). These distributions are reported entirely within the U.S. operating segment.

<sup>(4)</sup> The adoption of ASC Topic 606 impacted how the Company records certain revenues. See Note 3 for discussion of the impact of ASC Topic 606.

In thousands, except share and per share data

Financial Information About Geographic Areas

Below is a breakdown of selected financial information by geographic area:

	Three Months Ended June 30,					Six Mont Jun	led	
Revenues		2018		2017		2018		2017
U.S.	\$	712,483	\$	554,929	\$	1,312,128	\$	1,136,138
Brazil		72,977		87,841		153,113		182,540
Other international countries		107,002		112,085		210,494		219,454
Eliminations		(3,409)		(3,660)		(6,711)		(7,327)
Total	\$	889,053	\$	751,195	\$	1,669,024	\$	1,530,805

	As of	As of
Theatre Properties and Equipment-net	June 30, 2018	December 31, 2017
U.S.	\$ 1,465,915	\$ 1,439,168
Brazil	146,467	179,669
Other international countries	205,355	209,217
Total	\$ 1,817,737	\$ 1,828,054

#### 16. Related Party Transactions

The Company manages theatres for Laredo Theatre, Ltd. ("Laredo"). The Company is the sole general partner and owns 75% of the limited partnership interests of Laredo. Lone Star Theatres, Inc. owns the remaining 25% of the limited partnership interests in Laredo and is 100% owned by Mr. David Roberts, Lee Roy Mitchell's son-in-law. Lee Roy Mitchell is the Company's Chairman of the Board of Directors and directly and indirectly owns approximately 8% of the Company's common stock. Under the agreement, management fees are paid by Laredo to the Company at a rate of 5% of annual theatre revenues up to \$50,000 and 3% of annual theatre revenues in excess of \$50,000. The Company recorded \$327 and \$305 of management fee revenues during the six months ended June 30, 2018 and 2017, respectively. All such amounts are included in the Company's condensed consolidated financial statements with the intercompany amounts eliminated in consolidation.

The Company has an Aircraft Time Sharing Agreement with Copper Beech Capital, LLC ("Copper Beech") to use, on occasion, a private aircraft owned by Copper Beech. Copper Beech is owned by Mr. Mitchell and his wife, Tandy Mitchell. The private aircraft is used by Mr. Mitchell and other executives who accompany Mr. Mitchell to business meetings for the Company. The Company reimburses Copper Beech for the actual costs of fuel usage and the expenses of the pilots, landing fees, storage fees and similar expenses incurred during the trip. For the six months ended June 30, 2018 and 2017, the aggregate amounts paid to Copper Beech for the use of the aircraft was \$51 and \$63, respectively.

The Company leases 14 theatres and one parking facility from Syufy Enterprises, LP ("Syufy") or affiliates of Syufy. Raymond Syufy is one of the Company's directors and is an officer of the general partner of Syufy. Of these 15 leases, 14 have fixed minimum annual rent. The one lease without minimum annual rent has rent based upon a specified percentage of gross sales as defined in the lease. For the six months ended June 30, 2018 and 2017, the Company paid total rent of approximately \$12,980 and \$13,420, respectively, to Syufy.

The Company has a 50% voting interest in FE Concepts, a joint venture with AWSR, an entity owned by Lee Roy Mitchell and Tandy Mitchell. FE Concepts will develop and operate a family entertainment center that offers bowling, gaming, movies and other amenities. See Note 8 for further discussion.

In thousands, except share and per share data

#### 17. Commitments and Contingencies

Joseph Amey, et al. v. Cinemark USA, Inc., Case No. 3:13cv05669, In the United States District Court for the Northern District of California, San Francisco Division. The case presents putative class action claims for penalties and attorney's fees arising from alleged violations of the California wage statement law. The claim is also asserted as a representative action under the California Private Attorney General Act (PAGA) for penalties. The company denies the claims, denies that class certification is appropriate, denies that the plaintiff has standing to assert the claims alleged and is vigorously defending against the claims. The company denies any violation of law and plans to vigorously defend against all claims. The parties are awaiting the Court's ruling on Plaintiff's motion for class certification. The Company is unable to predict the outcome of this litigation or the range of potential loss.

Flagship Theatres of Palm Desert, LLC d/b/a Cinemas Palme D'Or v. Century Theatres, Inc., and Cinemark USA, Inc.; Superior Court of the State of California, County of Los Angeles. Plaintiff in this case alleges that the Company violated California antitrust and unfair competition laws by engaging in "circuit dealing" with various motion picture distributors and tortuously interfered with Plaintiff's business relationships. Plaintiff seeks compensatory damages, trebling of those damages under California law, punitive damages, injunctive relief, attorneys' fees, costs and interest. Plaintiff also alleges that the Company's conduct ultimately resulted in closure of its theatre in June 2016. The Company denied the allegations. In 2008, the Company moved for summary judgment on Plaintiff's claims, arguing primarily that clearances between the theatres at issue were lawful and that Plaintiff lacked proof sufficient to support certain technical elements of its antitrust claims. The trial court granted that motion and dismissed Plaintiff's claims. Plaintiff appealed and, in 2011, the Court of Appeal reversed, holding, among other things, that Plaintiff's claims were not about the illegality of clearances but were focused, instead, on "circuit dealing." Having re-framed the claims in that manner, the Court of Appeal held that the trial court's decision to limit discovery to the market where the theatres at issue operated was an error, as "circuit dealing" necessarily involves activities in different markets. Upon return to the trial court, the parties engaged in additional, broadened discovery related to Plaintiff's "circuit dealing" claim. Thereafter, the Company moved again for summary judgment on all of Plaintiff's claims. That new motion for summary judgment was pending when, on or about April 11, 2014, the trial court granted the Company's motion for terminating sanctions and entered a judgment dismissing the case with prejudice. Plaintiff then appealed that second dismissal, seeking to have the judgment reversed and the case remanded to the trial court. The Court of Appeal issued a ruling on May 24, 2016, reversing the granting of terminating sanctions and instead imposed a lesser evidentiary and damages preclusion sanction. The case returned to the trial court on October 6, 2016. On May 10, 2018, after a five-week jury trial, the jury found no liability on one circuit dealing claim and awarded Plaintiff damages on the other claim, which are tripled for antitrust damage awards. Plaintiff would also be entitled to certain court costs and to seek at least some portion of its attorney's fees. During the three and six months ended June 30, 2018, the Company recorded a litigation reserve based on an estimate of the jury award, which is reflected in loss on disposal of assets and other on the condensed consolidated income statement. The Company has filed a motion for a judgment notwithstanding the verdict and a motion for a new trial. Depending on the ruling on these motions, the Company intends to appeal any final judgment. Although the Company denies that it engaged in any form of circuit dealing, it cannot predict the outcome of its pending motions or future appeals.

The Company received a Civil Investigative Demand ("CID") from the Antitrust Division of the United States Department of Justice. The CID relates to an investigation under Sections 1 and 2 of the Sherman Act. The Company also received CIDs from the Antitrust Section of the Office of the Attorney General of the State of Ohio and later from other states regarding similar inquiries under state antitrust laws. The CIDs request the Company to answer interrogatories, and produce documents, or both, related to the investigation of matters including film clearances, potential coordination and/or communication with other major theatre circuits and related joint ventures. The Company intends to fully cooperate with all federal and state government agencies. Although the Company does not believe that it has violated any federal or state antitrust or competition laws, it cannot predict the ultimate scope, duration or outcome of these investigations.

From time to time, the Company is involved in various other legal proceedings arising from the ordinary course of its business operations, such as personal injury claims, employment matters, landlord-tenant disputes, patent claims and contractual disputes, some of which are covered by insurance. The Company believes its potential liability with respect to proceedings currently pending is not material, individually or in the aggregate, to the Company's financial position, results of operations and cash flows.

#### 18. Subsequent Event - Acquisition of common units of NCM

On July 5, 2018 the Company acquired 10,738,740 common units of NCM from AMC for \$78,400 in cash, or approximately \$7.30 per common unit. As a result of the acquisition of these shares, the Company's ownership of NCM increased from approximately 18% to 25%. The amount paid for the additional common units will be recorded as an increase in the Company's investment in NCM. See Note 7 for further discussion of the accounting for the Company's investment in NCM.

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion and analysis should be read in conjunction with our condensed consolidated financial statements and related notes and schedules included elsewhere in this report.

We are a leader in the motion picture exhibition industry, with theatres in the U.S., Brazil, Argentina, Chile, Colombia, Ecuador, Peru, Honduras, El Salvador, Nicaragua, Costa Rica, Panama, Guatemala, Bolivia, Curacao and Paraguay. As of June 30, 2018, we managed our business under two reportable operating segments – U.S. markets and international markets. See Note 15 to our condensed consolidated financial statements.

We generate revenues primarily from filmed entertainment box office receipts and concession sales with additional revenues from screen advertising sales, transactional fees and other revenue streams, such as vendor marketing promotions and meeting rentals located in some of our theatres. We also offer alternative entertainment, such as live and pre-recorded sports programs, concert events, the Metropolitan Opera, in-theatre gaming and other special events in our theatres through AC JV, LLC. NCM provides our domestic theatres with various forms of in theatre advertising. Our Flix Media subsidiaries have also allowed us to expand our screen advertising and alternative content within our international circuit and to other international exhibitors.

Films leading the box office during the six months ended June 30, 2018 included the carryover of Jumanji: Welcome to the Jungle, The Greatest Showman and Star Wars: The Last Jedi and new releases such as Black Panther, Avengers: Infinity War, Incredibles 2, Deadpool 2, Jurassic World: Fallen Kingdom, Solo: A Star Wars Story, A Quiet Place, Ready Player One, Peter Rabbit, Ocean's 8 and other films. Films scheduled for release during the remainder of 2018 include well-known franchise films such as Ralph Breaks the Internet, Dr. Seuss' The Grinch, Fantastic Beasts: The Crimes of Grindelwald, Ant-Man and the Wasp, Mission Impossible – Fallout and Hotel Transylvania 3: Summer Vacation and original titles including AquaMan, Christopher Robin and Mary Poppins, among other films.

Film rental costs are variable in nature and fluctuate with our admissions revenues. Film rental costs as a percentage of revenues are generally higher for periods in which more blockbuster films are released. Advertising costs, which are expensed as incurred, are primarily related to campaigns for new and renovated theatres, loyalty and membership programs and brand advertising that vary depending on the timing of such campaigns.

Concession supplies expense is variable in nature and fluctuates with our concession revenues. We negotiate prices for concession supplies directly with concession vendors and manufacturers to obtain volume rates.

Although salaries and wages include a fixed cost component (i.e. the minimum staffing costs to operate a theatre facility during non-peak periods), salaries and wages move in relation to revenues as theatre staffing is adjusted to respond to changes in attendance. In some international locations, staffing levels are also subject to local regulations.

Facility lease expense is primarily a fixed cost at the theatre level as most of our facility leases require a fixed monthly minimum rent payment. Certain leases are subject to percentage rent only, while others are subject to percentage rent in addition to their fixed monthly rent if a target annual performance level is achieved. Facility lease expense as a percentage of revenues is also affected by the number of theatres under operating leases, the number of theatres under capital leases and the number of fee-owned theatres.

Utilities and other costs include both fixed and variable costs and primarily consist of utilities, expenses for projection and sound equipment maintenance and monitoring, property taxes, janitorial costs, repairs, maintenance and security services.

General and administrative expenses are primarily fixed in nature and consist of the costs to support the overall management of the Company, including salaries and wages, incentive compensation and benefit costs for our corporate office personnel, facility expenses for our corporate offices, consulting fees, legal fees, audit fees, supplies and other costs that are not specifically associated with the operations of our theatres.

#### **Recent Developments**

On July 5, 2018 we acquired 10,738,740 common units of NCM from AMC for \$78.4 million in cash, or approximately \$7.30 per common unit. As a result of the acquisition of these shares, our ownership of NCM increased from approximately 18% to 25%. The amount paid for the additional common units will be recorded as an increase in our investment in NCM. See Note 7 to our condensed consolidated financial statements for further discussion of the accounting for our investment in NCM.

#### **Critical Accounting Policies**

We adopted ASC Topic 606 effective January 1, 2018, therefore, our revenue recognition policy has been modified as discussed in Note 3 to our condensed consolidated financial statements.

#### **Results of Operations**

The following table sets forth, for the periods indicated, certain operating data and the percentage of revenues represented by certain items reflected in our condensed consolidated statements of income.

	Three Mon		nded		led		
	2018		2017		2018		2017
Operating data (in millions):							
Revenues							
Admissions	\$ 508.9	\$	449.9	\$	961.5	\$	926.4
Concession	305.3		262.3		567.1		530.5
Other	74.8		39.0		140.4		73.9
Total revenues	\$ 889.0	\$	751.2	\$	1,669.0	\$	1,530.8
Cost of operations							
Film rentals and advertising	287.1		246.6		528.1		499.4
Concession supplies	51.1		41.8		91.9		83.9
Salaries and wages	100.4		89.8		193.5		174.0
Facility lease expense	81.2		82.3		163.3		166.6
Utilities and other	115.6		91.0		225.0		179.4
General and administrative expenses	43.0		37.9		85.4		76.1
Depreciation and amortization	64.3		59.2		128.7		116.5
Impairment of long-lived assets	2.8		4.3		3.4		4.6
Loss on disposal of assets and other	 16.9		0.1		20.8		0.9
Total cost of operations	762.4		653.0		1,440.1		1,301.4
Operating income	\$ 126.6	\$	98.2	\$	228.9	\$	229.4
Operating data as a percentage of total revenues:							
Revenues							
Admissions	57.2 %		59.9 %		57.6 %		60.5 %
Concession	34.3 %		34.9 %		34.0 %		34.7 %
Other	 8.5 %		5.2 %	_	8.4 %	_	4.8 %
Total revenues	 100.0 %		100.0 %		100.0 %		100.0 %
Cost of operations (1)							
Film rentals and advertising	56.4 %	)	54.8 %		54.9 %		53.9 %
Concession supplies	16.7 %	)	15.9 %		16.2 %		15.8 %
Salaries and wages	11.3 %	)	12.0 %		11.6%		11.4 %
Facility lease expense	9.1 %	)	11.0%		9.8 %		10.9 %
Utilities and other	13.0 %	)	12.1 %		13.5 %		11.7 %
General and administrative expenses	4.8 %	)	5.0%		5.1 %		5.0 %
Depreciation and amortization	7.2 %	)	7.9 %		7.7 %		7.6 %
Impairment of long-lived assets	0.3 %	)	0.6%		0.2 %		0.3 %
Loss on disposal of assets and other	1.9 %		%		1.2 %		0.1 %
Total cost of operations	85.8 %		86.9 %		86.3 %		85.0 %
Operating income	14.2 %		13.1 %		13.7 %		15.0 %
Average screen count (month end average)	5,986		5,904		5,973		5,902
Average operating screen count (month end average)	5,911		5,751		5,895		5,770
Total revenues per average screen (dollars)	\$ 148,522	\$	127,235	\$	279,428	\$	259,371

<sup>(1)</sup> All costs are expressed as a percentage of total revenues, except film rentals and advertising, which are expressed as a percentage of admissions revenues and concession supplies, which are expressed as a percentage of concession revenues.

#### Three months ended June 30, 2018 versus June 30, 2017

Revenues. Total revenues increased \$137.8 million to \$889.0 million for the three months ended June 30, 2018 ("second quarter of 2018") from \$751.2 million for the three months ended June 30, 2017 ("second quarter of 2017"), representing an 18.3% increase. The table below, presented by reportable operating segment, summarizes our revenue performance and certain key performance indicators for the three months ended June 30, 2018 and 2017.

	U.S. Operating Segment					Internatio	nal Operating Seg	ment		Consolidated					
								=	Currei						
					%			%		%				%	
		2018		2017	Change	2018	 2017	Change	2018	Change	2018		2017	Change	
Admissions revenues (1)	\$	408.9	\$	335.0	22.1 % \$	100.0	\$ 114.9	(13.0)% \$	111.7	(2.8)%	\$ 508.9	\$	449.9	13.1 %	
Concession revenues (1)	\$	249.6	\$	197.3	26.5 % \$	55.7	\$ 65.0	(14.3)% \$	61.5	(5.4)%	\$ 305.3	\$	262.3	16.4 %	
Other revenues (1)(2)	\$	50.5	\$	18.9	167.2 % \$	24.3	\$ 20.1	20.9 % \$	27.7	37.8%	\$ 74.8	\$	39.0	91.8%	
Total revenues (1)(2)	\$	709.0	\$	551.2	28.6 % \$	180.0	\$ 200.0	(10.0)% \$	200.9	0.4 %	\$ 889.0	\$	751.2	18.3 %	
Attendance (1)		50.6		43.0	17.7%	25.8	26.4	(2.3)%			76.4		69.4	10.1 %	
Average ticket price (1)	\$	8.08	\$	7.79	3.7 % \$	3.88	\$ 4.35	(10.8)% \$	4.33	(0.5)%	\$ 6.66	\$	6.48	2.8 %	
Concession revenues per patron (1)	\$	4.93	\$	4.59	7.4 % \$	2.16	\$ 2.46	(12.2)% \$	2.38	(3.3)%	\$ 4.00	\$	3.78	5.8 %	

- (1) Revenues and attendance amounts in millions. Average ticket price is calculated as admissions revenues divided by attendance. Concession revenues per patron is calculated as concession revenues divided by attendance.
- (2) U.S. operating segment revenues include eliminations of intercompany transactions with the international operating segment. See Note 15 to our condensed consolidated financial statements.
- (3) Constant currency revenue amounts, which are non-GAAP measurements, were calculated using the average exchange rate for the corresponding month for 2017. We translate the results of our international operating segment from local currencies into U.S. dollars using currency rates in effect at different points in time in accordance with U.S. GAAP. Significant changes in foreign exchange rates from one period to the next can result in meaningful variations in reported results. We are providing constant currency amounts for our international operating segment to present a period-to-period comparison of business performance that excludes the impact of foreign currency fluctuations.
  - <u>U.S.</u> Admissions revenues grew \$73.9 million primarily due to a 17.7% increase in attendance and a 3.7% increase in average ticket price. The increase in attendance was due to the number of blockbuster films in the second quarter of 2018 compared to the second quarter of 2017 as well as the favorable impact of luxury lounger conversions. Growth in average ticket price was primarily due to price increases. Concession revenues grew \$52.3 million primarily due to the 17.7% increase in attendance and a 7.4% increase in concession revenues per patron. Concession revenues per patron grew primarily due to price increases and incremental sales. Other revenues increased \$31.6 million primarily due to the impact of changes in revenue recognition as discussed in Note 3 to our condensed consolidated financial statements.
  - <u>International.</u> Admissions revenues decreased \$14.9 million as reported primarily due to a 2.3% decrease in attendance and a 10.8% decrease in average ticket price. Admissions revenues decreased \$3.2 million in constant currency. The decline in attendance was driven by weaker consumer appeal of the film slate during the second quarter of 2018 compared to the second quarter of 2017. Concession revenues decreased \$9.3 million as reported primarily due to the 2.3% decrease in attendance and a 12.2% decrease in concession revenues per patron. Concession revenues decreased \$3.5 million in constant currency. Average ticket price and concession revenues per patron decreased, as reported, primarily due to the impact of changes in foreign currency exchange rates in certain countries in which we operate and the mix of film during the second quarter of 2018. Other revenues increased primarily due to incremental screen advertising revenues as well as the impact of changes in revenue recognition as discussed in Note 3 to our condensed consolidated financial statements, partially offset by the impact of changes in foreign currency exchange rates in certain countries in which we operate.

Cost of Operations. The table below summarizes our theatre operating costs (in millions) by reportable operating segment for the three months ended June 30, 2018 and 2017.

	U.S. Operating Segment					Inter	al Operating Se	Consolidated					
		2018		2017		2018		2017	Constant Currency 2018 (1)		2018		2017
Film rentals and advertising	\$	238.7	\$	190.4	\$	48.4	\$	56.2	\$ 54.3	\$	287.1	\$	246.6
Concession supplies	\$	38.4	\$	27.8	\$	12.7	\$	14.0	\$ 14.0		51.1		41.8
Salaries and wages	\$	79.4	\$	66.7	\$	21.0	\$	23.1	\$ 24.0		100.4		89.8
Facility lease expense	\$	61.0	\$	59.9	\$	20.2	\$	22.4	\$ 22.1		81.2		82.3
Utilities and other	\$	83.7	\$	61.1	\$	31.9	\$	29.9	\$ 36.0		115.6		91.0

<sup>(1)</sup> Constant currency expense amounts, which are non-GAAP measurements were calculated using the average exchange rate for the corresponding month for 2017. We translate the results of our international operating segment from local currencies into U.S. dollars using currency rates in effect at different points in time in accordance with U.S. GAAP. Significant changes in foreign exchange rates from one period to the next can result in meaningful variations in reported results. We are providing constant currency amounts for our international operating segment to present a period-to-period comparison of business performance that excludes the impact of foreign currency fluctuations.

• <u>U.S.</u> Film rentals and advertising costs were \$238.7 million, or 58.4% of admissions revenues, for the second quarter of 2018 compared to \$190.4 million, or 56.8% of admissions revenues, for the second quarter of 2017. The increase in the film rentals and advertising rate was primarily due to the relative contribution of blockbuster films to overall box office during the second quarter of 2018. The second quarter of 2018 included such blockbuster releases as *Avengers: Infinity War, Incredibles 2, Jurassic World: Fallen Kingdom* and *Deadpool 2*, which grossed in excess of \$600 million, \$400 million, \$300 million and \$300 million, respectively, for the quarter. Conc ession supplies expense was \$38.4 million, or 15.4% of concession revenues, for the second quarter of 2018 compared to \$27.8 million, or 14.1% of concession revenues, for the second quarter of 2017. The increase in the concessions supplies rate was primarily due to product mix.

Salaries and wages increased to \$79.4 million for the second quarter of 2018 from \$66.7 million for the second quarter of 2017 primarily due to increased staffing levels to support the growth in attendance and expanded concession offerings, staffing at new and recently remodeled theatres, as well as increases in minimum and other wage rates. Facility lease expense increased to \$61.0 million for the second quarter of 2018 from \$59.9 million for the second quarter of 2017 primarily due to higher percentage rent due to the increase in revenues. Utilities and other costs increased to \$83.7 million for the second quarter of 2018 from \$61.1 million for the second quarter of 2017 primarily due to the presentation of transactional fees on a gross basis versus net basis due to the adoption of ASC Topic 606 (see Note 3 to our condensed consolidated financial statements).

• <u>International.</u> Film rentals and advertising costs were \$48.4 million (\$54.3 million in constant currency), or 48.4% of admissions revenues, for the second quarter of 2018 compared to \$56.2 million, or 48.9% of admissions revenues, for the second quarter of 2017. Concession supplies expense was \$12.7 million (\$14.0 million in constant currency), or 22.8% of concession revenues, for the second quarter of 2018 compared to \$14.0 million, or 21.5% of concession revenues, for the second quarter of 2017. The increase in the concessions supplies rate was primarily due to product mix.

Salaries and wages decreased to \$21.0 million (increased to \$24.0 million in constant currency) for the second quarter of 2018 compared to \$23.1 million for the second quarter of 2017. The as reported decrease was due to the impact of changes in foreign currency exchange rates in certain countries in which we operate, partially offset by increased local currency wages that were primarily driven by inflation, new theatres and limited flexibility in scheduling staff caused by shifting government regulations. Facility lease expense decreased to \$20.2 million (\$22.1 million in constant currency) for the second quarter of 2018 compared to \$22.4 million for the second quarter of 2017. The as reported decrease was due to lower percentage rent due to the decline in revenues. Utilities and other costs increased to \$31.9 million (\$36.0 million in constant currency) for the second quarter of 2018 compared to \$29.9 million for the second quarter of 2017. The as reported increase was primarily due to the presentation of transactional fees on a gross basis versus net basis due to the adoption of ASC Topic 606 (see Note 3 to our condensed consolidated financial statements).

General and Administrative Expenses. General and administrative expenses increased to \$43.0 million for the second quarter of 2018 from \$37.9 million for the second quarter of 2017. The increase was primarily due to higher salaries, insurance costs, benefits costs and professional fees.

Depreciation and Amortization. Depreciation and amortization expense increased to \$64.3 million during the second quarter of 2018 compared to \$59.2 million during the second quarter of 2017. The increase was primarily due to theatre remodels and new theatres.

Impairment of Long-Lived Assets. We recorded asset impairment charges on assets held and used of \$2.8 million during the second quarter of 2018 compared to \$4.3 million during the second quarter of 2017. The long-lived asset impairment charges recorded during each of the periods presented were specific to theatres that were directly and individually impacted by increased competition, adverse changes in market demographics or adverse changes in the development or the conditions of the areas surrounding the theatre. Impairment charges for the second quarter of 2018 impacted five of our twenty-seven reporting units. See Note 11 to our condensed consolidated financial statements.

Loss on Disposal of Assets and Other. We recorded a loss on disposal of assets and other of \$16.9 million during the second quarter of 2018 compared to \$0.1 million during the second quarter of 2017. Activity for the second quarter of 2018 was primarily due to the retirement of assets related to theatre remodels and the accrual of a reserve for outstanding litigation (see Note 17 to the condensed consolidated financial statements). Activity for the second quarter of 2017 was primarily due to the retirement of assets related to theatre remodels, offset by a gain on a landlord buyout of a theatre lease.

Interest Expense. Interest costs incurred, including amortization of debt issue costs, were \$28.5 million during the second quarter of 2018 compared to \$26.5 million during the second quarter of 2017. The increase was primarily due to an increase in the variable rate at which our term loan accrues interest.

Distributions from NCM. We recorded distributions from NCM of \$3.4 million during the second quarter of 2018 compared to \$2.8 million recorded during the second quarter of 2017, which were in excess of the carrying value of our Tranche 1 investment. See Note 7 to our condensed consolidated financial statements.

Interest expense – NCM. We recorded non-cash interest expense of \$4.9 million for the second quarter of 2018 related to the significant financing component associated with certain of our agreements with NCM. See Note 3 to our condensed consolidated financial statements for further discussion.

Equity in Income of Affiliates. We recorded equity in income of affiliates of \$6.4 million during the second quarter of 2018 compared to \$5.8 million during the second quarter of 2017. See Notes 7 and 8 to our condensed consolidated financial statements for information about our equity investments.

Income Taxes. Income tax expense of \$18.3 million was recorded for the second quarter of 2018 compared to \$29.4 million recorded for the second quarter of 2017. The effective tax rate was approximately 18.2% for the second quarter of 2018 compared to 36.2% for the second quarter of 2017. The effective tax rate in 2018 benefited from a lower U.S. federal tax rate. On December 22, 2017, the U.S. government enacted comprehensive tax legislation, the Tax Cuts and Jobs Act (the "Tax Act"). The Tax Act made changes to the U.S. tax code, which included (1) reduced U.S. corporate tax rate from 35 percent to 21 percent, (2) generally eliminated U.S. federal income taxes on dividends from foreign subsidiaries, and (3) a one-time transition tax on certain undistributed earnings of foreign subsidiaries. Additionally, the Company recorded a benefit to tax expense of \$7.1 million for the second quarter of 2018 related to the settlement of audits in the non-U.S. tax jurisdiction of Chile.

As of June 30, 2018, the amounts recorded for the Tax Act remain provisional for the transition tax, the re-measurement of deferred taxes, and our reassessment of valuation allowances. These estimates may be impacted by further analysis due to future clarification and guidance regarding earnings and profits computations, state tax conformity to federal tax changes, and potential tax planning options under consideration.

Income tax provisions for interim (quarterly) periods are based on estimated annual income tax rates and are adjusted for the effects of significant, infrequent or unusual items (i.e. discrete items) occurring during the interim period. As a result, the interim rate may vary significantly from the normalized annual rate.

#### Six months ended June 30, 2018 versus June 30, 2017

Revenues. Total revenues increased \$138.2 million to \$1,669.0 million for the six months ended June 30, 2018 ("2018 period") from \$1,530.8 million for the six months ended June 30, 2017 ("2017 period"), representing a 9.0% increase. The table below, presented by reportable operating segment, summarizes our revenue performance and certain key performance indicators for the six months ended June 30, 2018 and 2017.

	U.S. Operating Segment						Internatio	nal Operating Seg	gmen	Consolidated							
										Constant Currency (3)							
					%				%			%					%
		2018		2017	Change		2018	 2017	Change	20	18	Change		2018		2017	Change
Admissions revenues (1)	\$	758.2	\$	691.2	9.7	7% \$	203.3	\$ 235.2	(13.6)% \$	\$ 2	219.4	(6.7)%	\$	961.5	\$	926.4	3.8 %
Concession revenues (1)	\$	453.4	\$	400.7	13.2	2% \$	113.7	\$ 129.8	(12.4)% \$	\$	121.6	(6.3)%	\$	567.1	\$	530.5	6.9 %
Other revenues (1)(2)	\$	93.8	\$	36.9	154.2	2% \$	46.6	\$ 37.0	25.9 % 5	\$	51.5	39.2 %	\$	140.4	\$	73.9	90.0%
Total revenues (1)(2)	\$	1,305.4	\$	1,128.8	15.6	\$ %	363.6	\$ 402.0	(9.6)% \$	\$ 3	392.5	(2.4)%	\$	1,669.0	\$	1,530.8	9.0%
Attendance (1)		95.2		89.5	6.4	1%	49.7	54.2	(8.3)%					144.9		143.7	0.8 %
Average ticket price (1)	\$	7.96	\$	7.72	3.1	% \$	4.09	\$ 4.34	(5.8)% \$	\$	4.41	1.6 %	\$	6.64	\$	6.45	2.9 %
Concession revenues per patron (1)	\$	4.76	\$	4.48	6.2	2% \$	2.29	\$ 2.39	(4.2)% \$	\$	2.45	2.5 %	\$	3.91	\$	3.69	6.0 %

- (1) Revenues and attendance amounts in millions. Average ticket price is calculated as admissions revenues divided by attendance. Concession revenues per patron is calculated as concession revenues divided by attendance.
- U.S. operating segment revenues include eliminations of intercompany transactions with the international operating segment. See Note 15 to our condensed consolidated financial statements.

  Constant currency revenue amounts, which are non-GAAP measurements, were calculated using the average exchange rate for the corresponding month for 2017. We translate the results
- (3) Constant currency revenue amounts, which are non-GAAP measurements, were calculated using the average exchange rate for the corresponding month for 2017. We translate the results of our international operating segment from local currencies into U.S. dollars using currency rates in effect at different points in time in accordance with U.S. GAAP. Significant changes in foreign exchange rates from one period to the next can result in meaningful variations in reported results. We are providing constant currency amounts for our international operating segment to present a period-to-period comparison of business performance that excludes the impact of foreign currency fluctuations.
  - <u>U.S.</u> Admissions revenues grew \$67.0 million primarily due to a 6.4% increase in attendance and a 3.1% increase in average ticket price. The growth in attendance was due to the number of blockbuster films in the 2018 period compared to the 2017 period as well as the favorable impact of luxury lounger conversions. Average ticket price growth was primarily due to price increases. The increase in concession revenues of \$52.7 million was primarily due to the 6.4% increase in attendance and a

6.2% increase in concession revenues per patron. The increase in concession revenues per patron was primarily due to price increases and incremental sales. Other revenues increased \$56.9 million primarily due to the impact of changes in revenue recognition as discussed in Note 3 to our condensed consolidated financial statements.

• <u>International.</u> Admissions revenues decreased \$31.9 million as reported primarily due to an 8.3% decrease in attendance and a 5.8% decrease in average ticket price. Admissions revenues decreased \$15.8 million in constant currency. The decrease in attendance was driven by weaker consumer appeal of the film slate during the 2018 period compared to the 2017 period. Concession revenues decreased \$16.1 million as reported primarily due to the 8.3% decrease in attendance and a 4.2% decrease in concession revenues per patron. Concession revenues decreased \$8.2 million in constant currency. Average ticket price and concession revenues per patron decreased, as reported, primarily due to the impact of changes in foreign currency exchange rates in certain countries in which we operate and the mix of films during the 2018 period. Other revenues increased \$9.6 million primarily due to incremental screen advertising revenues as well as the impact of changes in revenue recognition as discussed in Note 3 to our condensed consolidated financial statements.

Cost of Operations. The table below summarizes our theatre operating costs (in millions) by reportable operating segment for the six months ended June 30, 2018 and 2017.

	U.S. Operating Segment					Intern	al Operating Se	Consolidated				
		2018 2017				2018	2017	Constant Currency 2018 (1)	2018	2017		
Film rentals and advertising	\$	431.6	\$	386.8	\$	96.5	\$	112.6	\$ 104.5	\$ 528.1	\$	499.4
Concession supplies	\$	66.9	\$	55.9	\$	25.0	\$	28.0	\$ 26.8	\$ 91.9	\$	83.9
Salaries and wages	\$	151.1	\$	129.9	\$	42.4	\$	44.1	\$ 46.7	\$ 193.5	\$	174.0
Facility lease expense	\$	122.0	\$	121.3	\$	41.3	\$	45.3	\$ 43.8	\$ 163.3	\$	166.6
Utilities and other	\$	162.7	\$	121.1	\$	62.3	\$	58.3	\$ 68.1	\$ 225.0	\$	179.4

- (1) Constant currency expense amounts, which are non-GAAP measurements were calculated using the average exchange rate for the corresponding month for 2017. We translate the results of our international operating segment from local currencies into U.S. dollars using currency rates in effect at different points in time in accordance with U.S. GAAP. Significant changes in foreign exchange rates from one period to the next can result in meaningful variations in reported results. We are providing constant currency amounts for our international operating segment to present a period-to-period comparison of business performance that excludes the impact of foreign currency fluctuations.
  - <u>U.S.</u> Film rentals and advertising costs were \$431.6 million, or 56.9% of admissions revenues, for the 2018 period compared to \$386.8 million, or 56.0% of admissions revenues, for the 2017 period. The increase in the film rentals and advertising rate was primarily due to the relative contribution of blockbuster films to overall box office during the 2018 period. The 2018 period included such blockbuster releases as *Black Panther, Avengers: Infinity War, Incredibles 2, Jurassic World: Fallen Kingdom* and *Deadpool 2*, which grossed in excess of \$600 million, \$600 million, \$400 million, \$300 million and \$300 million, respectively, for the period. Concession supplies expense was \$66.9 million, or 14.8% of concession revenues, for the 2018 period compared to \$55.9 million, or 14.0% of concession revenues, for the 2017 period. The increase in the concessions supplies rate was primarily due to product mix.
    - Salaries and wages increased to \$151.1 million for the 2018 period from \$129.9 million for the 2017 period primarily due to increased staffing levels to support the increased attendance and expanded concession offerings, staffing at new and recently remodeled theatres, increases in minimum and other wage rates. Facility lease expense increased to \$122.0 million for the 2018 period from \$121.3 million for the 2017 period primarily due to increased percentage rent due to the increase in revenues. Utilities and other costs increased to \$162.7 million for the 2018 period from \$121.1 million for the 2017 period primarily due to the presentation of transactional fees on a gross basis versus net basis due to the adoption of ASC Topic 606 (see Note 3 to our condensed consolidated financial statements).
  - <u>International.</u> Film rentals and advertising costs were \$96.5 million (\$104.5 million in constant currency), or 47.5% of admissions revenues, for the 2018 period compared to \$112.6 million, or 47.9% of admissions revenues, for the 2017 period. Concession supplies expense was \$25.0 million (\$26.8 million in constant currency), or 22.0% of concession revenues, for the 2018 period compared to \$28.0 million, or 21.6% of concession revenues, for the 2017 period.
    - Salaries and wages decreased to \$42.4 million (increased to \$46.7 million in constant currency) for the 2018 period compared to \$44.1 million for the 2017 period. The as reported decrease was due to the impact of changes in foreign currency exchange rates in certain countries in which we operate, partially offset by increased local currency wages that were primarily driven by inflation, new theatres and limited flexibility in scheduling staff caused by shifting government regulations. Facility lease expense decreased to \$41.3 million (\$43.8 million in constant currency) for the 2018 period compared to \$45.3 million for the 2017 period. The as reported decrease was due to lower percentage rent due to the decline in revenues. Utilities and other costs increased to \$62.3 million (\$68.1 million in constant currency) for the 2018 period compared to \$58.3 million for the 2017 period. The as reported increase was primarily due to the presentation of transactional fees on a gross basis versus net basis due to the adoption of ASC Topic 606 (see Note 3 to our condensed consolidated financial statements).

General and Administrative Expenses. General and administrative expenses increased to \$85.4 million for the 2018 period from \$76.1 million for the 2017 period. The increase was primarily due to higher salaries, insurance costs, benefits costs and professional fees.

Depreciation and Amortization. Depreciation and amortization expense increased to \$128.7 million during the 2018 period compared to \$116.5 million during the 2017 period. The increase was primarily due to theatre remodels and new theatres.

Impairment of Long-Lived Assets. We recorded asset impairment charges on assets held and used of \$3.4 million during the 2018 period compared to \$4.6 million during the 2017 period. The long-lived asset impairment charges recorded during each of the periods presented were specific to theatres that were directly and individually impacted by increased competition, adverse changes in market demographics or adverse changes in the development or the conditions of the areas surrounding the theatre. Impairment charges for the 2018 period impacted six of our twenty-seven reporting units. See Note 11 to our condensed consolidated financial statements.

Loss on Disposal of Assets and Other. We recorded a loss on disposal of assets and other of \$20.8 million during the 2018 period compared to \$0.9 million during the 2017 period. Activity for the 2018 period was primarily due to the retirement of assets related to theatre remodels and the accrual of a reserve for outstanding litigation (see Note 17 to the condensed consolidated financial statements). Activity for the 2017 period was primarily due to the retirement of assets related to theatre remodels, partially offset by gains related to the sale of excess land parcels and a gain on a landlord buyout of a theatre lease.

Interest Expense. Interest costs incurred, including amortization of debt issue costs, were \$55.6 million during the 2018 period compared to \$52.9 million during the 2017 period. The increase was primarily due to an increase in the variable rate at which our term loan accrues interest.

Loss on Debt Amendments. We recorded a loss on debt amendments of \$1.5 million for the 2018 period compared to \$0.2 million for the 2017 period related to amendments to our senior secured credit facility. See Note 5 to our condensed consolidated financial statements for further discussion.

Distributions from NCM. We recorded distributions from NCM of \$9.8 million during the 2018 period compared to \$9.6 million recorded during the 2017 period, which were in excess of the carrying value of our Tranche 1 investment. See Note 7 to our condensed consolidated financial statements.

 $Interest\ expense-NCM. \ We\ recorded\ non-cash\ interest\ expense\ of\ \$9.9\ million\ for\ the\ 2018\ period\ related\ to\ the\ significant\ financing\ component\ associated\ with\ certain\ of\ our\ agreements\ with\ NCM.\ See\ Note\ 3\ to\ our\ condensed\ consolidated\ financial\ statements\ for\ further\ discussion.$ 

Equity in Income of Affiliates. We recorded equity in income of affiliates of \$15.1 million during the 2018 period compared to \$15.9 million during the 2017 period. See Notes 7 and 8 to our condensed consolidated financial statements for information about our equity investments.

Income Taxes. Income tax expense of \$43.4 million was recorded for the 2018 period compared to \$73.8 million recorded for the 2017 period. The effective tax rate was approximately 23.1% for the 2018 period compared to 35.9% for the 2017 period. The effective tax rate in 2018 benefited from a lower U.S. federal tax rate. On December 22, 2017, the U.S. government enacted comprehensive tax legislation, the Tax Cuts and Jobs Act (the "Tax Act"). The Tax Act made changes to the U.S. tax code, which included (1) reduced U.S. corporate tax rate from 35 percent to 21 percent, (2) generally eliminated U.S. federal income taxes on dividends from foreign subsidiaries, and (3) a one-time transition tax on certain undistributed earnings of foreign subsidiaries. Additionally, the Company recorded a benefit to tax expense of \$7.1 million for the 2018 period related to the settlement of audits in the non-U.S. tax jurisdiction of Chile.

As of June 30, 2018, the amounts recorded for the Tax Act remain provisional for the transition tax, the re-measurement of deferred taxes, and our reassessment of valuation allowances. These estimates may be impacted by further analysis due to future clarification and guidance regarding earnings and profits computations, state tax conformity to federal tax changes, and potential tax planning options under consideration.

Income tax provisions for interim (quarterly) periods are based on estimated annual income tax rates and are adjusted for the effects of significant, infrequent or unusual items (i.e. discrete items) occurring during the interim period. As a result, the interim rate may vary significantly from the normalized annual rate.

#### **Liquidity and Capital Resources**

Operating Activities

We primarily collect our revenues in cash, mainly through box office receipts and the sale of concessions. In addition, nearly all of our theatres provide the patron a choice of using a credit card, debit card or advanced-sale type certificates such as a gift card, in place of cash. Because our revenues are received in cash prior to the payment of related expenses, we have an operating "float" and historically have not required traditional working capital financing. Cash provided by operating activities was \$279.9 million for the six months ended June 30, 2018 compared to \$235.7 million for the six months ended June 30, 2017. The increase in cash provided by operating activities was primarily due to the increase in net income and the timing of payments to suppliers.

Investing Activities

Our investing activities have been principally related to the development, remodel and acquisition of theatres. New theatre openings and acquisitions historically have been financed with internally generated cash and by debt financing, including borrowings under our senior secured credit facility. Cash used for investing activities was \$191.4 million for the six months ended June 30, 2018 compared to\$209.6 million for the six months ended June 30, 2017. The decrease was primarily due to a decrease in capital expenditures and a decrease in cash paid related to the acquisition of theatres.

Capital expenditures for the six months ended June 30, 2018 and 2017 were as follows (in millions):

		New		Existing	
<u>Period</u>	T	heatres	T	heatres (1)	Total
Six Months Ended June 30, 2018	\$	31.5	\$	131.1	\$ 162.6
Six Months Ended June 30, 2017	\$	31.1	\$	151.7	\$ 182.8

(1) The amounts for the six months ended June 30, 2018 and 2017 include \$4.3 million and \$5.9 million, respectively, related to the remodel of our corporate headquarters building in Plano, TX.

Capital expenditures for existing properties in the table above includes the costs of remodeling certain of our existing theatres to include Luxury Loungers and expanded concession offerings. During the six months ended June 30, 2018 and 2017, we had an average of 78 and 132 of our domestic screens, respectively, temporarily closed for such remodels

Our U.S. theatre circuit consisted of 339 theatres with 4,566 screens at June 30, 2018. During the six months ended June 30, 2018, we built one new theatre and twelve screens and closed one theatre with seven screens. At June 30, 2018, we had signed commitments to open three new theatres and 30 screens in domestic markets during the remainder of 2018 and open nine new theatres with 95 screens subsequent to 2018. We estimate the remaining capital expenditures for the development of these 125 domestic screens will be approximately \$99 million.

Our international theatre circuit consisted of 200 theatres with 1,432 screens at June 30, 2018. During the six months ended June 30, 2018, we built three theatres and fifteen screens and acquired three theatres and nineteen screens. At June 30, 2018, we had signed commitments to open five new theatres and 29 screens in international markets during the remainder of 2018 and open seven new theatres and 51 screens subsequent to 2018. We estimate the remaining capital expenditures for the development of these 80 international screens will be approximately \$46 million.

Actual expenditures for continued theatre development, remodels and acquisitions are subject to change based upon the availability of attractive opportunities. We plan to fund capital expenditures for our continued development with cash flow from operations, borrowings under our senior secured credit facility, and proceeds from debt issuances, sale leaseback transactions and/or sales of excess real estate.

Financing Activities

Cash used for financing activities was \$99.1 million for the six months ended June 30, 2018 compared to \$83.1 million for the six months ended June 30, 2017. The increase was primarily due to the increase in the quarterly cash dividend from \$0.29 to \$0.32 per share of common stock and the payment of debt issuance costs associated with the amendment of our senior secured credit facility (see Note 5 to our condensed consolidated financial statements).

We, at the discretion of the board of directors and subject to applicable law, anticipate paying regular quarterly dividends on our common stock. The amount, if any, of the dividends to be paid in the future will depend upon our then available cash balance,

anticipated cash needs, overall financial condition, loan agreement restrictions as discussed below, future prospects for earnings and cash flows, as well as other relevant factors.

We may from time to time, subject to compliance with our debt instruments, purchase our debt securities on the open market depending upon the availability and prices of such securities. Long-term debt consisted of the following as of June 30, 2018 (in millions):

Cinemark USA, Inc. term loan	\$ 656.2
Cinemark USA, Inc. 5.125% senior notes due 2022	400.0
Cinemark USA, Inc. 4.875% senior notes due 2023	755.0
Other	2.8
Total long-term debt	\$ 1,814.0
Less current portion	8.0
Subtotal long-term debt, less current portion	\$ 1,806.0
Less: Debt discounts and debt issuance costs, net of accumulated amortization	31.2
Long-term debt, less current portion, net of debt issuance costs	\$ 1,774.8

As of June 30, 2018, Cinemark USA, Inc. had \$100.0 million in available borrowing capacity on its revolving credit line.

#### Contractual Obligations

In March 2018, Cinemark USA, Inc. amended its senior secured credit facility, which extended the maturity of the term loan to March 2025. Included below is an updated summary of long-term debt obligations and related estimated scheduled interest payment obligations as of June 30, 2018, reflecting the amended agreement.

	rayments Due by Period												
			Less Than						After				
Contractual Obligations	Total One Year 1 - 3 Years 3 - 5 Years							5 Years					
Long-term debt (1)	\$ 1,814.0	\$	8.0	\$	14.6	\$	413.2	\$	1,378.2				
Scheduled interest payments on long-term deb(2)	\$ 437.4	\$	82.6	\$	164.2	\$	148.9	\$	41.7				

- Amounts are presented before adjusting for debt issuance costs.
- (2) Amounts include scheduled interest payments on fixed rate and variable rate debt agreements. Estimates for the variable rate interest payments were based on interest rates in effect on June 30, 2018. The average interest rates in effect on our fixed rate and variable rate debt are 5.0% and 3.9%, respectively, as of June 30, 2018.

There have been no other material changes in our contractual obligations previously disclosed in "Liquidity and Capital Resources" in our Annual Report on Form 10-K for the year ended December 31, 2017 filed February 23, 2018.

Off-Balance Sheet Arrangements

Other than operating leases and purchase commitments disclosed in our Annual Report on Form 10-K for the year ended December 31, 2017 filed February 23, 2018, we do not have any off-balance sheet arrangements.

Senior Secured Credit Facility

Cinemark USA, Inc. has a senior secured credit facility that includes a \$700.0 million term loan, with a maturity date of March 2025, and a \$100.0 million revolving credit line, with a maturity date of December 2022 (collectively referred to as the "Credit Agreement").

On March 29, 2018, Cinemark USA, Inc., amended its senior secured credit facility to extend the maturity of the term loan to March 29, 2025, reduce the rate at which the term loan bears interest by 0.25% and to reduce the amount of real property required to be mortgaged to secure the loans. Under the amended facility, quarterly principal payments of \$1.6 million are due on the term loan through December 31, 2024, with a final principal payment of \$613.4 million due on March 29, 2025.

Interest on the term loan accrues at Cinemark USA, Inc.'s option at: (A) the base rate equal to the greater of (1) the US "Prime Rate" as quoted in The Wall Street Journal or if no such rate is quoted therein, in a Federal Reserve Board statistical release, (2) the federal funds effective rate plus 0.50%, and (3) a one-month Eurodollar-based rate plus 1.0%, plus, in each case, a margin of 0.75%

per annum, or (B) a Eurodollar-based rate for a period of 1, 2, 3, 6, 9 or 12 months plus a margin of 1.75% per annum. Interest on the revolving credit line accrues, at our option, at: (A) a base rate equal to the greater of (1) the US "Prime Rate" as quoted in The Wall Street Journal or if no such rate is quoted therein, in a Federal Reserve Board statistical release, (2) the federal funds effective rate plus 0.50%, and (3) a one-month Eurodollar-based rate plus 1.0%, plus, in each case, a margin that ranges from 0.50% to 1.25% per annum, or (B) a Eurodollar-based rate for a period of 1, 2, 3, 6, 9 or 12 months plus a margin that ranges from 1.50% to 2.25% per annum. The margin of the revolving credit line is determined by the consolidated net senior secured leverage ratio as defined in the Credit Agreement.

Cinemark USA, Inc.'s obligations under the Credit Agreement are guaranteed by Cinemark Holdings, Inc. and certain of Cinemark USA, Inc.'s domestic subsidiaries and are secured by mortgages on certain fee and leasehold properties and security interests in substantially all of Cinemark USA, Inc.'s and the guarantors' personal property, including, without limitation, pledges of all of Cinemark USA, Inc.'s capital stock, all of the capital stock of certain of Cinemark USA, Inc.'s domestic subsidiaries and 65% of the voting stock of certain of its foreign subsidiaries.

The Credit Agreement contains usual and customary negative covenants for agreements of this type, including, but not limited to, restrictions on Cinemark USA, Inc.'s ability, and in certain instances, its subsidiaries' and our ability, to consolidate or merge or liquidate, wind up or dissolve; substantially change the nature of its business; sell, transfer or dispose of assets; create or incur indebtedness; create liens; pay dividends or repurchase stock; and make capital expenditures and investments. If Cinemark USA, Inc. has borrowings outstanding on the revolving credit line, it is required to satisfy a consolidated net senior secured leverage ratio covenant as defined in the Credit Agreement.

The dividend restriction contained in the Credit Agreement prevents the Company and any of its subsidiaries from paying a dividend or otherwise distributing cash to its stockholders unless (1) the Company is not in default, and the distribution would not cause Cinemark USA, Inc. to be in default, under the Credit Agreement; and (2) the aggregate amount of certain dividends, distributions, investments, redemptions and capital expenditures made since December 18, 2012, including dividends declared by the board of directors, is less than the sum of (a) the aggregate amount of cash and cash equivalents received by Cinemark Holdings, Inc. or Cinemark USA, Inc. as common equity since December 18, 2012, (b) Cinemark USA, Inc.'s consolidated EBITDA minus 1.75 times its consolidated interest expense, each as defined in the Credit Agreement, and (c) certain other defined amounts. As of June 30, 2018, Cinemark USA, Inc. could have distributed up to approximately \$2,790.4 million to its parent company and sole stockholder, Cinemark Holdings, Inc., under the terms of the Credit Agreement, subject to its available cash and other borrowing restrictions outlined in the agreement.

At June 30, 2018, there was \$656.2 million outstanding under the term loan and no borrowings outstanding under the revolving credit line. Cinemark USA, Inc. had \$100.0 million in available borrowing capacity on the revolving credit line. The average interest rate on outstanding term loan borrowings under the Credit Agreement at June 30, 2018 was approximately 3.9% per annum.

Cinemark USA, Inc. 5.125% Senior Notes

On December 18, 2012, Cinemark USA, Inc. issued \$400.0 million aggregate principal amount of 5.125% Senior Notes due 2022, at par value (the "5.125% Senior Notes"). Interest on the 5.125% Senior Notes is payable on June 15 and December 15 of each year. The 5.125% Senior Notes mature on December 15, 2022.

The indenture to the 5.125% Senior Notes contains covenants including limitations on the amount of dividends that could be paid by Cinemark USA, IncAs of June 30, 2018, Cinemark USA, Inc. could have distributed up to approximately \$2,442.4 million to its parent company and sole stockholder, Cinemark Holdings, Inc., under the terms of the indenture to the 5.125% Senior Notes, subject to its available cash and other borrowing restrictions outlined in the indenture. The indenture allows Cinemark USA, Inc. to incur additional indebtedness if it satisfies the coverage ratio specified in the indenture, after giving effect to the incurrence of the additional indebtedness, and in certain other circumstances. The required minimum coverage ratio is 2 to 1 and our actual ratio as of June 30, 2018 was approximately 6.3 to 1.

Cinemark USA, Inc. 4.875% Senior Notes

On May 24, 2013, Cinemark USA, Inc. issued \$530.0 million aggregate principal amount of 4.875% Senior Notes due 2023, at par value (the "4.875% Senior Notes"). On March 21, 2016, Cinemark USA, Inc. issued an additional \$225.0 million aggregate principal amount of the 4.875% Senior Notes at 99.0% of the principal amount plus accrued and unpaid interest from December 1, 2015. Proceeds, after payment of fees, were used to finance the redemption of Cinemark, USA, Inc.'s \$200.0 million 7.375% Senior Subordinated Notes. These additional notes have identical terms, other than the issue date, the issue price and the first interest payment date, and constitute part of the same series as the Company's existing 4.875% Senior Notes. Interest on the 4.875% Senior Notes is payable on June 1 and December 1 of each year. The 4.875% Senior Notes mature on June 1, 2023.

The indenture to the 4.875% Senior Notes contains covenants that include limitations on the amount of dividends that Cinemark USA, Inc. can payAs of June 30, 2018, Cinemark USA, Inc. could have distributed up to approximately \$2,437.1 million to its parent

company and sole stockholder, Cinemark Holdings, Inc., under the terms of the indenture to the 4.875% Senior Notes, subject to its available cash and oher borrowing restrictions outlined in the indenture. The indenture allows Cinemark USA, Inc. to incur additional indebtedness if it satisfies the coverage ratio specified in the indenture, after giving effect to the incurrence of the additional indebtedness, and in certain other circumstances. The required minimum coverage ratio is 2 to 1 and our actual ratio as of June 30, 2018 was approximately 6.3 to 1.

Covenant Compliance

As of June 30, 2018, we believe we were in full compliance with all agreements, including all related covenants, governing our outstanding debt.

#### Seasonality

Our revenues have historically been seasonal, coinciding with the timing of releases of motion pictures by the major distributors. Generally, the most successful motion pictures have been released during summer months in the U.S., extending from May to July, and during the holiday season, extending from November through year-end. The timing of releases, however, has become less pronounced as distributors have begun releasing content more evenly throughout the year. In our Latin American markets, while Hollywood content has similar release dates as in the U.S., the local holidays and seasons can vary. The unexpected emergence of a hit film during other periods can impact this seasonality trend. The timing and quality of film releases can have a significant impact on our results of operations, and the results of one period are not necessarily indicative of results for the following period or for the same period in the following year.

### Item 3. Quantitative and Qualitative Disclosures About Market Risk

We have exposure to financial market risks, including changes in interest rates, foreign currency exchange rates and other relevant market prices.

#### Interest Rate Risk

We are currently party to a variable rate debt facility. An increase or decrease in interest rates would affect our interest expense relating to our variable rate debt. At June 30, 2018, we had an aggregate of approximately \$656.2 million of variable rate debt outstanding. Based on the interest rates in effect on the variable rate debt outstanding at June 30, 2018, a 100 basis point increase in market interest rates would increase our annual interest expense by approximately \$6.6 million.

The table below provides information about our fixed rate and variable rate long-term debt agreements as of June 30, 2018:

	Expected Maturity for the Twelve-Month Periods Ending June 30, (in millions)										Average Interest				
	2019		2020		2021		2022		2023	T	hereafter	Total	F	air Value	Rate
Fixed rate	\$ 1.4	\$	1.4	\$		\$	_	\$	400.0	\$	755.0	\$ 1,157.8	\$	1,143.8	5.0%
Variable rate	6.6		6.6		6.6		6.6		6.6		623.2	656.2		657.0	3.9%
Total debt	\$ 8.0	\$	8.0	\$	6.6	\$	6.6	\$	406.6	\$	1,378.2	\$ 1,814.0	\$	1,800.8	

#### Foreign Currency Exchange Rate Risk

Other than the devaluation of the Argentine peso, discussed in Note 13 to the condensed consolidated financial statements, there have been no material changes in foreign currency exchange rate risk previously disclosed in "Quantitative and Qualitative Disclosures About Market Risk" in our Annual Report on Form 10-K for the year ended December 31, 2017 filed February 23, 2018.

### **Item 4. Controls and Procedures**

#### Evaluation of the Effectiveness of Disclosure Controls and Procedures

As of June 30, 2018, we carried out an evaluation required by the Exchange Act, under the supervision and with the participation of our principal executive officer and principal financial officer, of the effectiveness of the design and operation of our disclosure controls and procedures, as defined in Rule 13a-15(e) of the Exchange Act. Based on this evaluation, our principal executive officer and principal financial officer concluded that, as of June 30, 2018, our disclosure controls and procedures were effective to provide reasonable assurance that information required to be disclosed by us in the reports that we file or submit under the Exchange Act is recorded, processed, summarized, and reported within the time periods specified in the SEC's rules and forms and were effective to provide reasonable assurance that such information is accumulated and communicated to our management, including our principal executive officer and principal financial officer, as appropriate to allow timely decisions regarding required disclosures.

#### Changes in Internal Control Over Financial Reporting

There have been no changes in our internal control over financial reporting identified in connection with the evaluation required by paragraph (d) of Exchange Act Rules 13a-15 that occurred during the quarter ended June 30, 2018 that materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

### PART II - OTHER INFORMATION

## Item 1. Legal Proceedings

Other than the recent developments noted at Note 17, there have been no material changes from legal proceedings previously reported under "Business – Legal Proceedings" in our Annual Report on Form 10-K for the year ended December 31, 2017 filed February 23, 2018

## Item 1A. Risk Factors

There have been no material changes from risk factors previously disclosed in "Risk Factors" inour Annual Report on Form 10-K for the year ended December 31, 2017 filed February 23, 2018.

### Item 6. Exhibits

k10 1	Limited Liebility Commons	A amagina of EE Composite	LLC datad as of Amril 20, 2019
*10.1	Limited Liability Company	Agreement of FE Concepts.	LLC dated as of April 20, 2018.

- \*10.2 Management Services Agreement by and between FE Concepts, LLC and Cinema Operations, L.L.C. dated as of April 20, 2018.
- \*10.3 Theatre Services Agreement by and between FE Concepts, LLC and CNMK Texas Properties, LLC dated as of April 20, 2018.
- \*31.1 Certification of Mark Zoradi, Chief Executive Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- \*31.2 Certification of Sean Gamble, Chief Financial Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- \*32.1 Certification of Mark Zoradi, Chief Executive Officer, pursuant to 18 U.S.C. Section 1350, as added by Section 906 of the Sarbanes-Oxley Act of 2002.
- \*32.2 Certification of Sean Gamble, Chief Financial Officer, pursuant to 18 U.S.C. Section 1350, as added by Section 906 of the Sarbanes-Oxley Act of 2002.
- \* 101 Financial Statements from the quarterly report on Form 10-Q of Cinemark Holdings, Inc. for the quarter ended June 30, 2018, filed August 8, 2018, formatted in XBRL: (i) the Condensed Consolidated Balance Sheets, (ii) the Condensed Consolidated Statements of Income, (iii) the Condensed Consolidated Statements of Comprehensive Income, (iv) the Condensed Consolidated Statements of Cash Flows and (v) the Notes to Condensed Consolidated Financial Statements tagged as detailed text.
- \* filed herewith.

## **SIGNATURES**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

## CINEMARK HOLDINGS, INC.

Registrant

DATE: August 8, 2018

/s/Mark Zoradi

Mark Zoradi Chief Executive Officer

/s/Sean Gamble

Sean Gamble

Chief Financial Officer

# Limited Liability Company Agreement OF FE CONCEPTS, LLC

a Texas limited liability company

(Dated Effective as of April 20, 2018)

### **NOTICE**

THE INTERESTS DESCRIBED IN THIS LIMITED LIABILITY COMPANY AGREEMENT HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED OR THE SECURITIES LAWS OF ANY OF THE STATES OF THE UNITED STATES OR THE SECURITIES LAWS OF ANY FOREIGN JURISDICTION (THE "ACTS"). SUCH INTERESTS ARE BEING OFFERED AND SOLD IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT FOR SUCH INTERESTS UNDER THE ACTS OR AN EXEMPTION THEREFROM UNDER THE ACTS.

THE DELIVERY OF THIS LIMITED LIABILITY COMPANY AGREEMENT SHALL NOT CONSTITUTE AN OFFER TO SELL OR THE SOLICITATION OF AN OFFER TO BUY NOR SHALL THERE BE ANY OFFER, SOLICITATION OR SALE OF INTERESTS IN FE CONCEPTS, LLC IN ANY JURISDICTION IN WHICH SUCH OFFER, SOLICITATION OR SALE IS NOT AUTHORIZED OR TO ANY PERSON TO WHOM IT IS UNLAWFUL TO MAKE SUCH OFFER, SOLICITATION OR SALE.

THESE SECURITIES ARE SUBJECT TO RESTRICTIONS ON TRANSFERABILITY AND RESALE, MAY NOT BE TRANSFERRED OR RESOLD EXCEPT AS PERMITTED UNDER THE ACTS PURSUANT TO REGISTRATION OR EXEMPTION THEREFROM AND MAY NOT BE SOLD OR OTHERWISE TRANSFERRED EXCEPT IN ACCORDANCE WITH THE REQUIREMENTS AND CONDITIONS SET FORTH IN THIS LIMITED LIABILITY COMPANY AGREEMENT.

## TABLE OF CONTENTS

		Page
ARTICLE I DEFI	INITIONS	1
Section 1.1	Definitions	1
Section 1.2	Additional Definitions	11
Section 1.3	Construction	12
ARTICLE II ORG	GANIZATION, PURPOSES AND POWERS	12
Section 2.1	Formation of Company	12
Section 2.2	Certificates and Documents	13
Section 2.3	Company Name	13
Section 2.4	Term of the Company	13
Section 2.5	Principal Place of Business	13
Section 2.6	Registered Agent and Office	13
Section 2.7	Company Purposes	13
Section 2.8	Powers	13
Section 2.9	Organization Expenses	13
Section 2.10	Confidentiality	13
ARTICLE III ME	EMBERS; MEMBERSHIP UNITS	14
Section 3.1	Authorization of Membership Units	14
Section 3.2	Member Meetings	14
Section 3.3	Voting and Consents of Members	15
Section 3.4	Consent of Class	15
Section 3.5	Securities Laws; Legends	15
ARTICLE IV CA	PITAL CONTRIBUTIONS	16
Section 4.1	Capital Contributions of Members	16
Section 4.2	No Interest on Contributions; No Withdrawal	17
ARTICLE V MAN	NAGEMENT	17
Section 5.1	Management of the Company; Board	17
Section 5.2	Authority of the Board	19
Section 5.3	Restrictions on Authority of the Board	20
Section 5.4	Books and Records	22
Section 5.5	Committees	22
Section 5.6	Officers	23
Section 5.7	Management Services Agreement	23
Section 5.8	Theater Services Agreement	23

		Page
ARTICLE VI TR	ANSFERS	23
Section 6.1	Conditions to Transfers	23
Section 6.2	Effect of Membership Units in Hands of the Transferee	27
Section 6.3	Joinder	28
Section 6.4	Imposition of Restrictions	28
Section 6.5	Limitation on Admission	28
Section 6.6	Other Limitations	28
ARTICLE VII CA	APITAL ACCOUNTS; TAX ALLOCATIONS; TAX MATTERS	29
Section 7.1	Capital Accounts	29
Section 7.2	Allocations of Profits and Losses	30
Section 7.3	Special Allocations	30
Section 7.4	Other Allocation Rules	32
Section 7.5	Tax Allocations	32
Section 7.6	Company Tax Returns	33
Section 7.7	Tax Year and Accounting Matters	33
Section 7.8	Tax Elections	33
Section 7.9	Distributions	33
Section 7.10	Tax Distributions.	34
Section 7.11	Tax Matters Member	34
ARTICLE VIII T	ERM; DISSOLUTION	38
Section 8.1	Term; Termination	38
Section 8.2	Liquidation of Company	39
Section 8.3	Distribution Upon Dissolution of the Company	39
ARTICLE IX RE	PRESENTATION, WARRANTIES AND COVENANTS	39
Section 9.1	Representations, Warranties and Covenants of the Members	39
ARTICLE X ADD	DITIONAL AGREEMENTS	40
Section 10.1	Drag-Along Sale	40
Section 10.2	Class A Member Call Option	42
Section 10.3	Assignment of Option Units	42
Section 10.4	Class B Default	42
Section 10.5	Class B Conversion	43
Section 10.6	Sale of Class C Membership Units	44
Section 10.7	Sale of the Company	45

Section 11.1	Conflicts	45
Section 11.2	Limitation of Liability; Exculpation	46
Section 11.3	Indemnification	46
ARTICLE XII MI	ISCELLANEOUS PROVISIONS	47
Section 12.1	Notices	47
Section 12.2	Amendment	48
Section 12.3	Jurisdiction; Waiver of Jury Trial	48
Section 12.4	Headings	48
Section 12.5	Governing Law; Severability	48
Section 12.6	Successors; Counterparts	49
Section 12.7	Entire Agreement	49
Section 12.8	No Waiver	49

**EXHIBITS** 

Exhibit A Members Exhibit B Appointments

Form of Management Services Agreement Form of Theater Services Agreement Form of Capital Note Form of Assignment of Membership Units Exhibit C Exhibit D

Exhibit E

Exhibit F

### FE CONCEPTS, LLC

## LIMITED LIABILITY COMPANY AGREEMENT

This Limited Liability Company Agreement (this "Agreement") of FE Concepts, LLC (the "Company"), dated as of April 20, 2018 (the "Effective Date"), is entered into by and between CNMK Texas Properties, LLC, a Texas limited liability company, and AWSR Investments, LLC, a Texas limited liability company, and such other Persons (as defined herein) who are or, who from time to time, become signatories to this Agreement.

## **RECITALS**

- A. CNMK Texas Properties, LLC is a wholly owned indirect subsidiary of Cinemark USA, Inc., a corporation engaged in the business of owning and operating movie theaters.
  - B. AWSR Investments, LLC is an affiliate of Copper Beech Capital, LLC.
- C. The parties hereto desire to enter into this Agreement for the purpose of jointly developing, owning and operating family entertainment facilities offering a combination of the entertainment activities conducted by the respective Affiliates (as defined herein) of each party.

## STATEMENTS OF AGREEMENT

**NOW**, **THEREFORE**, for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto hereby agree as follows:

# ARTICLE I DEFINITIONS

- Section 1.1 **Definitions**. The following capitalized terms shall have the following meanings when used in this Agreement and the Exhibits hereto:
- "Active Management Role" means serving as a member of the Board and having primary responsibility for the strategic planning of the Company.
- "Adjusted Capital Account" shall mean, with respect to any Member, a special account maintained for such Member, the balance of which shall equal such Member's Capital Account balance, increased by the amount (if any) of such Member's share of the Company Minimum Gain and Member Minimum Gain.
- "Adjusted Capital Account Deficit" shall mean, with respect to any Member, the deficit balance, if any, in such Member's Capital Account as of the end of the relevant fiscal year, after giving effect to the following adjustments:
- (a) credit to such Capital Account any amounts which such Member is obligated to restore or is deemed to be obligated to restore pursuant to the penultimate sentences of Treasury Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5); and

- (b) debit to such Capital Account the items described in Treasury Regulations Sections 1.704-1(b)(2)(ii)(d)(4), (5) and (6).
- "Affiliate" means, with respect to any Person, any Person controlling, controlled by or under common control with such Person, with the concept of control in such context meaning the possession, directly or indirectly, of the power to direct the management and policies of another, whether through the ownership of voting securities, by contract or otherwise.
  - "Agreement" shall have the meaning assigned to such term in the preamble.
  - "Assignment" means an Assignment of Membership Units in the form of Exhibit F.
- "Approved Budget" means the annual operating and capital budget adopted and approved by the Board as may be revised from time to time with the approval of the Board.
- "Assumed Tax Rate" means the highest marginal federal income tax rate prescribed for the relevant type of income for any Member in the relevant Fiscal Year.
  - "Board" means the board of Managers of the Company.
- "Book Basis" shall mean, with respect to any asset, the asset's adjusted basis for federal income tax purposes; provided, however, (a) if property is contributed to the Company, the initial Book Basis of such property shall equal its fair market value on the date of contribution and (b) if the Capital Accounts of the Company are adjusted pursuant to Treasury Regulations Section 1.704-1(b) to reflect the fair market value of any Company asset, the Book Basis of such asset shall be adjusted to equal its respective fair market value as of the time of such adjustment in accordance with such Treasury Regulations. The Book Basis of all assets shall be adjusted thereafter by depreciation and amortization as provided in Treasury Regulations Section 1.704-1 (b)(2)(iv)(g).

"Business Code" means the Texas Business Organizations Code.

"Call Option Trigger" means the occurrence of any of the following:

- (a) the eighth anniversary of the Effective Date;
- (b) a Company Mitchell Trigger Event;
- (c) a Cinemark Mitchell Trigger Event;
- (d) a Class B Default;
- (e) a Class B Conversion;
- (f) Class A Member Change in Control or a Class A Parent Change in Control to a third party and in each case the Class A Member does not invoke its drag-along rights under <u>Section 10.1</u>.

### "Call Purchase Price" means:

- (a) the amount of Unreturned Capital of the Class B Member (or Class C Member, as the case may be) if the Call Option Trigger is (i) a Class B Default or (ii) a Class B Conversion (other than as a result of a Company Mitchell Trigger Event or a Cinemark Mitchell Trigger Event); or
- (b) the greater of (i) the product of (A) 8 x consolidated EBITDA less Long-Term Debt, Severance Payments, and if not duplicative, Management Payments and Exit Bonus plus cash and (B) Membership Interest of the Class B Member (or Class C Member, as the case may be) expressed as a percentage and (ii) the Designated Price, if the Call Option Trigger is an event described in (a), (b) or (c) of the definition of Call Option Trigger.

"Capital Account" means, with respect to any Member, the Capital Account maintained for such Member in accordance with the following provisions:

- (i) to each Member's Capital Account there shall be credited such Member's Capital Contributions, such Member's distributive share of Profits, any items in the nature of income or gain which are specially allocated pursuant to Section 7.3 or Section 7.4 hereof, and the amount of any Company liabilities assumed by such Member or which are secured by any asset of the Company distributed to such Member;
- (ii) to each Member's Capital Account there shall be debited the amount of cash and the fair market value of any Company asset (as determined by the Board) distributed to such Member pursuant to any provision of this Agreement, such Member's distributive share of Losses, any items in the nature of expenses or losses which are specially allocated pursuant to Section 7.3 or Section 7.4 hereof, the amount of any liabilities of such Member assumed by the Company or which are secured by any property contributed by such Member to the Company;
- (iii) except as otherwise provided in this Agreement, in the event all or a portion of an interest in the Company is transferred in accordance with the terms of this Agreement, the transferree shall succeed to the Capital Account of the transferrer to the extent it relates to the transferred interest; and
- (iv) in determining the amount of any credit or debit for purposes of maintaining each Member's Capital Account there shall be taken into account Code Section 752(c) and any other applicable provisions of the Code and Treasury Regulations.

The foregoing provisions and the other provisions of this Agreement relating to the maintenance of Capital Accounts are intended to comply with the Treasury Regulations Section 1.704 and shall be interpreted and applied in a manner consistent with such Treasury Regulations. In the event the Company shall determine that it is prudent to modify the manner in which the Capital Accounts, or any debits or credits thereto (including, without limitation, debits or credits relating to liabilities which are secured by contributed or distributed property or which are assumed by the Company, or any Member), are computed in order to comply with such Treasury Regulations, the Board, after consultation from the Tax Matters Member, may make such modifications. The Board may also (i) make any adjustments that are necessary or appropriate to maintain equality between the Capital Accounts of the Members and the amount of Company capital reflected on the Company's balance sheet, as computed for book purposes, in accordance with the Treasury Regulations Section 1.704-1(b)(2)(iv)(q) and (ii) make any appropriate modifications in the event unanticipated events might otherwise cause this Agreement not to comply with Treasury Regulations Section 1.704-1(b).

"Capital Contributions" means, with respect to any Member, the aggregate amount of money or the initial Book Basis of any other property (other than money), net of any liabilities assumed by the Company to which such property is subject, contributed to the Company with respect to the interest in the Company held by such Member, which may include an Initial Capital Contribution or a Subsequent Capital Contribution. Loans to the Company shall not be considered Capital Contributions or included in the Capital Account of any Member.

"Capital Leases" means any lease or similar arrangement which is of a nature that payment obligations of the lessee or obligor thereunder at the time are or should be capitalized and shown as liabilities (other than current liabilities) upon a balance sheet of such lessee or obligor prepared in accordance with GAAP, including Statement of Financial Accounting Standards No. 13.

"Capital Members" mean holders of the Capital Units who are Members of the Company.

"Capital Units" means the Class A Membership Units, the Class B Membership Units and the Class C Membership Units.

"CB Proposed Valuation" means the aggregate value of 100% of the Membership Interests in the Company proposed by the CB Member in connection with the submission of a CB Offer Notice.

"CB Purchase Price" means the product of (A) the CB Proposed Valuation and (B) Membership Interest attributable to the CB Units expressed as a percentage.

"Change in Control" means the occurrence of any one or more of the following:

(i) any Person (other than a Member or a Permitted Transferee) becomes the beneficial owner, directly or indirectly, of Membership Interests of the Company representing more than fifty percent (50%) of the Company's then outstanding Membership Interests;

- (ii) any consolidation, merger, recapitalization or similar transaction involving the Company; or
- (iii) the sale, lease, exchange or other transfer (in one transaction or a series of related transactions) of all or any material portion of the assets of the Company.
- "Cinemark Mitchell Trigger Event" means, after the first Facility has been in operation for a full year, Lee Roy Mitchell no longer serves as a board member or executive of Cinemark Holdings, Inc. or any subsequent parent of Cinemark Holdings, Inc. for any reason including without limitation, retirement, death or disability.
  - "Class A Competing Business" means the motion picture exhibition business.
  - "Class A Member" means CNMK Texas Properties, LLC.
- "Class A Membership Unit" means a single Class A Membership Unit in the Company held by a Member who has made a Capital Contribution to the Company. The number of Class A Membership Units authorized shall be as determined from time to time by the Board.
- "Class A Parent" means Cinemark Holdings, Inc. or any of its direct or indirect Subsidiaries that may be a direct or indirect beneficial owner of the Class A Member.
  - "Class A Change in Control" means a Change in Control of the Class A Parent.
  - "Class B Member" means AWSR Investments, LLC.
- "Class B Change in Control" means a Change in Control of the Class B Member, the Class C Member, Management Co. or Management Parent.
  - "Class B Competing Business" means an entertainment facility incorporating restaurants, games and bowling.
- "Class B Conversion" means the conversion of Class B Membership Units into Class C Membership Units pursuant to Section 10.5.
- "Class B Membership Unit" means a single Class B Membership Unit in the Company held by a Member who has made a Capital Contribution to the Company. The number of Class B Membership Units authorized shall be as determined from time to time by the Board.
  - "Class C Member" means a holder of Class C Membership Units.
- "Class C Membership Unit" means a single Class C Membership Unit in the Company held by a Member who has made a Capital Contribution to the Company. The number of Class C Membership Units authorized shall be as determined from time to time by the Board.
  - "Code" means the Internal Revenue Code of 1986, as amended.

- "Company Mitchell Trigger Event" means, after the first Facility has been in operation for a full year, Lee Roy Mitchell no longer serves in an Active Management Role in the Company for any reason including without limitation, retirement, death or disability.
- "Company Sale Transaction" means any transaction constituting a Class A Change in Control, a sale of all outstanding Membership Units to a third party purchaser, a merger, consolidation, joint venture or similar transaction involving the Company, a sale of all or substantially all assets of the Company or other transaction resulting in a Change in Control of the Company or disposition of the assets or business of the Company.
- "Company Minimum Gain" means "partnership minimum gain" determined in accordance with Treasury Regulations Section 1.704-2(d) and allocated pursuant to Section 7.3(a).
  - "Consultant" means Cinemark USA, Inc., a Texas corporation or any of its Subsidiaries.
- "Designated Price" means the sum of (i) the Unreturned Capital contributed by the Class B Member (or Class C Member, as the case may be) in connection with each Facility plus (ii) an eight percent (8%) compounded annual return on the Unreturned Capital contributed by the Class B Member (or Class C Member, as the case may be) in connection with each Facility calculated from the date such capital was contributed to the Company, less the Class B Members (or Class C Members) pro rata share of any Exit Bonus.
- "Distributable Cash" means: (a) all cash, revenues and funds received by the Company from Company operations less (b) the sum of the following: (i) all principal, interest, fees, charges and other amounts owed to any Member in connection with any loan by any Member to the Company; (ii) all expenses payable by the Company to a third party pursuant to the Approved Budget; and (iii) such reserves, if any, as the Board deems reasonably necessary for the proper operation of the Company's business.
- "EBITDA" means (a) Net Income, plus (b) Interest Expense, plus (c) to the extent deducted in the calculation of Net Income, depreciation expense and amortization expense, plus (d) income tax expense.
- "Exit Bonus" means any payments to be paid to the Manager by the Company in connection with a Trigger Event (as such term is defined in the Management Services Agreement).
- "Facility" or "Facilities" means an entertainment facility incorporating restaurants, bars, games, rides, bowling and motion picture theater auditoriums branded as Strike + Reel.
- "Fiscal Year" means the Company's fiscal year which will be a 52-53 week tax year ending on the Sunday closest to December 31 of each year.
- "GAAP" is generally accepted accounting principles set forth in the opinions and pronouncements of the Accounting Principles Board of the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or in such other statements by such other Person as may be approved by a significant segment of the accounting profession, which are applicable to the circumstances as of the date of determination.

- "Independent Third Party" means a Person who is not an Affiliate of, or related to, an Offering Member or a Transferring Member.
- "Initial Capital Contribution" means the Initial Class A Contribution or the Initial Class B Contribution.
- "Interest Expense" means for any fiscal period, interest expense (whether cash or non-cash) determined in accordance with GAAP for the relevant period ending on such date, including, without limitation or duplication, all commissions, discounts, or related amortization and other fees and charges with respect to letters of credit and bankers' acceptance financing and the net costs associated with interest rate swap, cap, and similar arrangements, and the interest portion of any deferred payment obligation (including leases of all types).
- "Involuntary Transfer" means a transfer (directly or indirectly) of all or any part of a Membership Unit or other interest in the Company to another Person resulting (i) from bankruptcy proceedings involving a Member; (ii) from an assignment of any Membership Units for the benefit of creditors of a Member; or (iii) an execution of judgment or a foreclosure by a court of law against any Membership Units or the acquisition of record or beneficial ownership of any Membership Units by a lender or creditor.
- "Lien" means a claim, mortgage, deed of trust, levy, charge, pledge, security interest or other encumbrance of any kind, whether voluntarily incurred or arising by operation of law or otherwise against any property.
- "Long-Term Debt" means the aggregate long-term liabilities of the Company as determined in accordance with GAAP, including Capital Leases.
- "Loss" shall mean, with respect to the Company, for each taxable year, each item of the Company's taxable loss or deduction for such taxable year, as determined under Section 703(a) of the Code, and Section 1.703-1 of the Treasury Regulations (for this purpose, all items of deduction and loss required to be stated separately pursuant to Section 703(a)(1) of the Code shall be included in taxable loss), but with the following adjustments:
- (a) Any expenditures of the Company described in Section 705(a)(2)(B) of the Code, including any items treated under Section 1.704-1 (b)(2)(iv)(i) of the Treasury Regulations as items described in Section 705(a)(2)(B) of the Code, shall be considered an item of taxable deduction or loss;
- (b) In the event the Book Basis of any Company asset is reduced as a result of an adjustment to Book Basis under Treasury Regulations Section 1.704-1(b) the amount of such reduction shall be taken into account as loss from the disposition of such asset for purposes of computing Loss;
- (c) Loss resulting from any disposition of property with respect to which loss is recognized for federal income tax purposes shall be computed by reference to the Book Basis of the property disposed of, notwithstanding that the adjusted tax basis of such property differs from its Book Basis;

- (d) Any items which are specially allocated pursuant to <u>Section 7.3</u> shall not be taken into account in computing Loss;
- and
- (e) In lieu of the depreciation, amortization and other cost recovery deductions taken into account in computing such taxable loss or deduction, there shall be taken into account depreciation and amortization as determined pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv)(g), if applicable for such taxable year or other period.
  - "Manager" means a voting member of the Board.
- "Management Co." means Cinema Operations, L.L.C., a Texas limited liability company and direct or indirect subsidiary of Management Parent.
- "Management Parent" means Entertainment Properties Group, Inc., or any of its direct or indirect Subsidiaries that may be a direct or indirect beneficial owner of Management Co.
- "Management Payments" means any payments made or due to the Manager under the Management Services Agreement or the Theater Services Agreement.
- "Management Services Agreement" means the Management Services Agreement between the Company and Management Co., in substantially the form attached as Exhibit C hereto.
- "Material Contract" means a contract for which the Company (i) is obligated to pay or is entitled to receive amounts of Fifty Thousand Dollars (\$50,000.00) or more that is not included in the Approved Budget or (ii) grants any Person any lien or other encumbrance on any asset of the Company, other than Permitted Liens.
- "*Member Minimum Gain*" means "partner nonrecourse debt minimum gain" within the meaning set forth in Treasury Regulations Section 1.704-2(i) with respect to and allocated pursuant to <u>Section 7.3(a)</u>.
- "Member Nonrecourse Debt Minimum Gain" means "partner nonrecourse debt minimum gain" within the meaning set forth in Treasury Regulations Section 1.704-2(i)(2) and shall be determined in accordance with Treasury Regulations Section 1.704-2(i)(3).
- "Member Nonrecourse Deductions" shall have the meaning set forth in Treasury Regulations Section 1.704-2(i)(1) with respect to Partner Nonrecourse Deductions and shall be determined in accordance with Treasury Regulations Section 1.704-2(i)(2).
  - "Members" mean, collectively, the Class A Member, the Class B Member and any Class C Member of the Company.
- "Membership Interest" means the ownership interest of a Member in the Company expressed as the percentage of the aggregate number of all outstanding Class A Membership Units, Class B Membership Units and Class C Membership Units of the Company held by such Member.

- "Membership Unit" means a single ownership interest in the Company, which such interests are classified as Class A Membership Units, Class B Membership Units and Class C Membership Units. Subject to Section 5.3 and Section 4.1, the number and classes of Membership Units authorized and the amount of Capital Contribution required of a Member acquiring a Membership Unit shall be as determined by the Board.
  - "Mitchell Foundation" means a foundation organized by Lee Roy Mitchell and Tandy Mitchell.
- "Net Income" means, as calculated on a consolidated basis for the Company and its Subsidiaries for any period as at any date of determination, the net profit (or loss), after provision for taxes and before compensation expenses attributable to any equity based compensation, of the Company and its Subsidiaries for such period taken as a single accounting period calculated in accordance with GAAP.
- "Net Profit and Net Loss" shall mean for each taxable year or other period the excess of items of Profit over items of Loss for such period, or the items of Loss over the items of Profit for such period, as appropriate. Net Profit and Net Loss shall not include items of Profit and Loss allocated pursuant to Section 7.3.
  - "Nonrecourse Deductions" shall have the meaning set forth in Treasury Regulations 1.702-2(b)(1).
  - "Operating Documents" means this Agreement, the Management Services Agreement and the Theater Services Agreement.
- "Permitted Liens" means (i) liens for taxes not yet due and payable; (ii) liens for taxes which are being contested in good faith; (iii) liens of any landlord, carrier, warehouseman, mechanic, materialman and any like liens arising in the ordinary course of business for sums that are not delinquent; (iv) easements, rights of way, zoning ordinances and other similar liens affecting real property; (v) liens and other encumbrances in favor of banks which are borrowings of the Company approved by the Board and (vi) liens and other encumbrances arising by operation of law.
- "*Person*" means any individual or entity, including, without limitation, any corporation, association, partnership, limited liability company, joint venture, trust, estate, governmental authority or other entity or organization.
- "*Profit*" shall mean, with respect to the Company, for each taxable year, each item of the Company's taxable income or gain for such taxable year, as determined under Section 703(a) of the Code, and Section 1.703-1 of the Treasury Regulations (for this purpose, all items of income and gain required to be stated separately pursuant to Section 703(a)(l) of the Code shall be included in taxable income or gain), but with the following adjustments:
- (a) Any tax-exempt income, as described in Section 705(a)(l)(B) of the Code, realized by the Company during such taxable year shall be considered an item of taxable income;

- (b) In the event the Book Basis of any Company asset is increased pursuant to Treasury Regulations Section 1.704-1(b), the amount of such adjustment shall be taken into account as gain from the disposition of such asset for purposes of computing Profit;
- (c) Gain resulting from any disposition of property with respect to which gain is recognized for federal income tax purposes shall be computed by reference to the Book Basis of the property disposed of, notwithstanding that the adjusted tax basis of such property differs from its Book Basis; and
  - (d) Any items which are specially allocated pursuant to Section 6.3 shall not be taken into account in computing Profit.
- "Qualified Purchase Offer" means a Purchase Offer that is (i) an all cash purchase price, (ii) not less than the CB Proposed Valuation for 100% of the Membership Interests in the Company, and (iii) from a bona fide third party purchaser that is not engaged in a Class A Competing Business or a Class B Competing Business.
- "Revised Partnership Audit Procedures" means Code Sections 6221 through 6241, as amended by the Bipartisan Budget Act of 2015, as such sections may be subsequently amended, and including any Treasury Regulations or other administrative guidance promulgated thereunder.
  - "Securities Act" means the Securities Act of 1933, as amended.
- "Severance Payments" means the Class C Member's share of any severance payments required to be paid by the Class C Member under the Management Agreement.
- "Subsidiary" means any Person of which the Company owns equity interests having a majority of the voting power in electing the governing body of such Person directly or through one or more Subsidiaries or, in the case of a partnership, limited liability partnership or other similar entity, equity interests conveying, directly or indirectly, a majority of the economic interests in such partnership or entity.
- "Substitute Member" means the transferee of a Member's Membership Units, when admitted and shown as such on the books and records of the Company.
- "Theater Services Agreement" means the Theater Services Agreement between the Company and Consultant, substantially in the form attached as Exhibit D hereto.
- "Transfer" means any transfer, sale, assignment, pledge, encumbrance or other disposition, irrespective of whether any of the foregoing is effected voluntarily, by operation of law or otherwise, or whether inter vivos or upon death.
  - "Transferee" means any Person to whom Membership Units have been Transferred.
- "Treasury Regulations" means the regulations promulgated under the Code by the United States Treasury Department, as such regulations may be amended from time to time (including corresponding provisions of succeeding regulations).

"Unreturned Capital" means the amount of Capital Contributions contributed by a Member less the amount of any cash or property distributed to such Member as a return of capital.

"Voting Members" means the Members holding Voting Units.

"Voting Units" means the Class A Membership Units and the Class B Membership Units.

Section 1.2 **Additional Definitions**. The following terms have the meanings ascribed to such terms in the following Sections of this Agreement.

	9 ( 10 2/1)
Call Option Notice	Section 10.2(b)
Call Right Closing Date	Section 10.2(b)
Capital Contribution Default	Section 4.1(d)
Capital Note	Section 4.1(d)(ii)
CB Acceptance Notice	Section 10.6(b)
CB Member	Section 10.6(a)
CB Notice Period	Section 10.6(a)
CB Offer	Section 10.6(a)
CB Offer Notice	Section 10.6(a)
CB Units	Section 10.6(a)
Certificate	Section 2.2
Class A Managers	Section $5.1(b)(i)(A)$
Class A Member Call Option	Section 10.2(a)
Class B Default	Section 10.4
Class B Managers	Section 5.1(b)(i)(B)
Committee(s)	Section 5.5
Company	Preamble
Company Sale Notice	Section 10.7(c)
Company Tax Liability	Section 7.11(e)
Confidential Information	Section 2.10
Drag-Along Sale	Section 10.1(a)
Dragged Members	Section 9.1(a)
Effective Date	Preamble
Exercising Member	Section 6.1(d)(iv)
Indemnitee	Section 11.3(a)
Initial Class A Contribution	Section 4.1(a)
Initial Class B Contribution	Section 4.1(b)
Offer Election Period	Section 10.6(a)
Offered Units	Section 5.1(c)
Offering Member	Section 6.1(c)
Offering Member Notice	Section 6.1(c)(i)
Officer(s)	Section 5.6
Option Units	Section 10.2(a)
Partnership Representative	Section 7.11(a)
Pass-Thru Member	Section 7.11(d)
Permitted Transfer	Section 7.11(d) Section 6.1(a)
1 clinited Transici	Section 6.1(a)

Permitted Transferee	Section 6.1(b)
Proposed Purchaser	Section 6.1(d)(i)
Purchase Offer	Section 10.7(a)
Purchasing Member	Section 6.1(c)(iv)
ROFO	Section 6.1(c)
ROFO Acceptance Notice	Section 6.1 (c)(iv)
ROFO Notice Period	Section 6.1(c)(ii)
ROFR	Section 6.1(d)
ROFR Acceptance Notice	Section 6.1(d)(iv)
ROFR Notice Period	Section 6.1(d)(ii)
Sale Election Notice	Section 10.7(a)
Sale Election Period	Section 10.7(a)
Sale Process	Section 10.7(a)
Sale Right	Section 10.7(a)
Subject Units	Section 6.1(d)
Subsequent Capital Contribution	Section 4.1(d)
Tax Distributions	Section 7.10
Tax Matters Member	Section 7.11(a)
Tax Matters Partner	Section 7.11(a)
Tax Proceeding	Section 7.11(c)(ii)
Transfer Notice	Section 6.1(d)(i)
Transferring Member	Section 7.1(d)
Waived ROFO Transfer Period	Section $6.1(c)(v)$
Waived ROFR Transfer Period	Section $6.1(d)(v)$

Section 1.3 **Construction**. Any reference to any law will be deemed also to refer to such law as amended and all rules and regulations promulgated thereunder, unless the context requires otherwise. The words "include," "includes," and "including" will be deemed to be followed by "without limitation." Pronouns in masculine, feminine and neuter genders will be construed to include the plural and vice versa, unless the context otherwise requires. The words "this Agreement," "herein," "hereof," "hereby," "hereunder," and words of similar import refer to this Agreement as a whole and not to any particular subdivision unless expressly so limited. Any agreement, schedule, attachment or exhibit referred to herein shall mean such agreement, schedule, attachment or exhibit as amended, restated, supplemented or modified from time to time to the extent permitted by the applicable provisions of this Agreement. Unless otherwise stated, references to recitals, articles, sections, paragraphs, schedules and exhibits of this Agreement.

# ARTICLE II ORGANIZATION, PURPOSES AND POWERS

Section 2.1 Formation of Company. The Company was formed pursuant to the Business Code.

- Section 2.2 **Certificates and Documents.** The organizer has executed and filed the Certificate of Formation of the Company (the "Certificate"). The Board shall execute and file all other certificates, notices, statements or other instruments required by law for the formation or operation of a limited liability company in all jurisdictions where the Company may propose to do business.
- Section 2.3 **Company Name**. The name of the Company is FE Concepts, LLC. The business of the Company will be conducted under the name of the Company or such other name or names as the Board may adopt upon (i) causing an amendment to the Certificate to be filed with the Secretary of State of the State of Texas and (ii) sending notice thereof to the Members.
- Section 2.4 **Term of the Company**. The existence of the Company initially commenced on the date of the filing of the Certificate in the office of the Secretary of State of the State of Texas and will continue indefinitely unless dissolved in accordance with the provisions of this Agreement.
- Section 2.5 **Principal Place of Business**. The address of the principal place of business of the Company shall be 12400 Coit Road, Suite 800, Dallas, Texas 75251, or such other address as shall be designated from time to time by the Board. The Company may maintain other offices at such other locations as the Board shall determine from time to time.
- Section 2.6 **Registered Agent and Office**. The Company's registered office in Texas shall be at 12400 Coit Road, Suite 800, Dallas, Texas 75251 and the name of the registered agent for service of process is Gary Witherspoon. At any time, the Board may designate another registered agent and/or registered office.
- Section 2.7 **Company Purposes.** The business purposes for which the Company is formed are to build, own and operate a Facility and such other related lawful acts or activities for which limited liability companies may be formed under the Business Code.
- Section 2.8 **Powers.** Subject to the provisions of this Agreement, the Company shall have the power to do any and all acts and things necessary, appropriate, advisable or convenient for the furtherance and accomplishment of the purposes of the Company set forth in Section 2.7, so long as said acts and things may be lawfully engaged in or performed under the Business Code.
- Section 2.9 **Organization Expenses**. Each Member shall pay its own expenses incurred in connection with the formation and organization of the Company and the drafting and negotiation of the Operating Documents through the date hereof.
- Section 2.10 **Confidentiality**. Each Member shall, and shall cause its Affiliates to, keep all information concerning the Company, including the terms of the Operating Documents (collectively, "*Confidential Information*"), strictly confidential. Notwithstanding the foregoing, each Member may disclose Confidential Information: (i) to its directors, employees, consultants, advisors, Affiliates, counsel and accountants on an as-needed basis to the extent such Persons agree to keep such information confidential; and (ii) as required by applicable law. For the avoidance of doubt, it is understood and agreed that Affiliates of CNMK Texas Properties, LLC are required to make public disclosures of material facts and events under U.S.

federal securities law and that nothing contained in this Agreement shall be construed to prevent such Affiliates from making such public disclosures (including, without limitation, public filings or disclosures relating to the Company, the Operating Documents or the business of the Company) as they determine, in their sole and absolute discretion, is required by applicable law or regulations. The term "Confidential Information" shall not include such portions of the Confidential Information as (i) are or become generally available to the public other than as a result of the disclosure by the receiving party; (ii) was known by the receiving party prior to disclosure by the non-disclosing party; or (iii) become available to the receiving party on a non-confidential basis from a source other than the disclosing party (or agent thereof) which is not prohibited from disclosing such Confidential Information to the receiving party by a legal, contractual or fiduciary obligation to the disclosing party.

# ARTICLE III MEMBERS; MEMBERSHIP UNITS

Section 3.1 **Authorization of Membership Units.** The Membership Units shall be classified as Class A Membership Units, Class B Membership Units and Class C Membership Units. The current total number of Membership Units which the Company is authorized to issue is twelve thousand (12,000) Membership Units, of which four thousand (4,000) shall be designated Class A Membership Units, four thousand (4,000) shall be designated Class B Membership Units and four thousand (4,000) shall be designated Class C Membership Units. All of the Class A Membership Units shall be deemed issued to and held by the Class A Member. All of the Class B Membership Units shall be deemed issued to and held by the Class C Membership Units shall be issued upon the occurrence of a Class B Conversion.

## Section 3.2 **Member Meetings**.

- (a) <u>Calling the Meeting</u>. Meetings of the Voting Members may be called by (i) the Board or (ii) by a Voting Member. Only Voting Members shall have the right to call or attend meetings of the Members, or take any action by vote as a meeting or by written consent without a meeting, in all cases to take any action or conduct any business.
- (b) <u>Notice</u>. Written notice stating the place, date and time of the meeting and, in the case of a meeting of the Voting Members not regularly scheduled, describing the purposes for which the meeting is called, shall be delivered not fewer than five (5) days and not more than thirty (30) days before the date of the meeting to each Voting Member, by or at the direction of the Board or the Voting Member calling the meeting, as the case may be. Meetings of the Voting Members may be held at the Company's principal office or at such other place as the Board may designate in the notice for such meeting or the Voting Members may agree upon.
- (c) <u>Participation</u>. Any Voting Member may participate in a meeting of the Voting Members by means of conference telephone or other communications equipment by means of which all Persons participating in the meeting can hear each other, and participation in a meeting by such means shall constitute presence in person at such meeting.

- (d) <u>Vote by Proxy</u>. On any matter that is to be voted on by Voting Members, a Voting Member may vote in person or by proxy, and such proxy may be granted in writing, by means of electronic transmission or as otherwise permitted by applicable law. Every proxy shall be revocable in the discretion of the Voting Member executing it unless otherwise provided in such proxy; <u>provided</u> that such right to revocation shall not invalidate or otherwise affect actions taken under such proxy prior to such revocation.
- (e) <u>Conduct of Business</u>. The business to be conducted at such meeting need not be limited to the purpose described in the notice and can include business to be conducted by Voting Members holding Voting Units; <u>provided</u>, that the appropriate Voting Members shall have been notified of the meeting in accordance with <u>Section 3.2(b)</u>. Attendance of a Voting Member at any meeting shall constitute a waiver of notice of such meeting, except where a Voting Member attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened.
- (f) Quorum; Required Vote. A quorum of any meeting of the Voting Members shall require the presence of the Voting Members holding a majority of the Voting Units held by all Voting Members. Subject to Section 3.3, no action at any meeting may be taken by the Voting Members (i) unless the appropriate quorum is present and (ii) except as otherwise provided in Section 5.3(b), at any meeting at which a quorum is present, without the affirmative vote of Voting Members holding a majority of the Voting Units held by all Voting Members.
- Section 3.3 **Voting and Consents of Members**. Any action requiring the consent or approval of the Members shall be approved solely by the Voting Members in accordance with <u>Section 5.3(b)</u>. To the extent that the Voting Members approve any action pursuant to this Agreement, the Board shall carry out such action in accordance with the approval by the Voting Members. Any action required or permitted to be taken or authorized at a meeting of the Voting Members may be taken or authorized without a meeting if, prior or subsequent to the action, a consent or consents thereto is signed by the holders of the minimum number of Membership Units that would be necessary to authorize or take such action pursuant to <u>Section 5.3(b)</u> and is filed with the records of the Company. The Class C Membership Units shall be non-voting Membership Units with no right to vote on any matter, except as provided in <u>Section 4.1(c)</u>, <u>Section 5.3(c)</u> and <u>Section 12.2</u>.
- Section 3.4 **Consent of Class**. To the extent that any consent, approval or other action by the holder or holders of a class of Membership Units is required or permitted under this Agreement, such consent, approval or other action shall be the consent of, approval or other action by the holders of a majority of the Membership Units of such class and shall be binding on all holders of Membership Units of such class.

### Section 3.5 Securities Laws; Legends.

(a) The Membership Units have been issued pursuant to a claim of exemption from the registration or qualification provisions of federal and state securities laws and, in addition to the other restrictions contained herein, may not be transferred without compliance with the registration or qualification provisions of applicable federal and state securities laws or applicable exemptions therefrom.

(b) If certificates evidencing any Membership Units are issued, each such certificate shall bear such legends as may be required by applicable federal or state laws, or as may be deemed necessary or appropriate by the Board to reflect restrictions upon transfer contemplated herein.

# ARTICLE IV CAPITAL CONTRIBUTIONS

## Section 4.1 **Capital Contributions of Members.**

- (a) The Class A Member has made (or will promptly make, at the request of the Board, a contribution to the capital of the Company in the amount of twenty million dollars (\$20,000,000) (the "*Initial Class A Contribution*").
- (b) The Class B Member has made (or will promptly make, at the request of the Board, a contribution to the capital of the Company in the amount of twenty million dollars (\$20,000,000) (the "*Initial Class B Contribution*").
- (c) The aggregate Initial Class A Contribution and the Initial Class B Contribution shall each be made as of the Effective Date in cash, along with and the contribution of mutually agreed costs and expenses incurred on behalf of the Company prior to the Effective Date.
- (d) In the event that the Board approves an additional Facility or Facilities each Capital Member shall be required to make subsequent capital contributions in such amounts and on such dates as may be required by the Board, provided that following a Company Mitchell Trigger Event or Cinemark Mitchell Trigger Event, subsequent mandatory capital contributions shall require the consent of the Class C Member. In the event of mandatory capital call, each Capital Member shall be required to contribute, pro rata, the aggregate amount of the capital call (a "Subsequent Capital Contribution"). In the event that any Capital Member fails to make its Initial Capital Contribution or any Subsequent Capital Contribution within thirty (30) days of the date set forth in a notice by the Board (or within thirty (30) days of the Effective Date with respect to an Initial Capital Contribution) (in each case, a "Capital Contribution Default"), the non-defaulting Member may elect one of the following:
  - ( i ) such non-defaulting Member may contribute the amount of the defaulting Member's Capital Contribution, with a corresponding adjustment to the Membership Interest held by the Members based on the aggregate amount of capital contributed by each Member; or
  - (ii) the non-defaulting Member may contribute the amount of the defaulting Member's Capital Contribution in the form of a loan evidenced by a capital note in the form of Exhibit E (a "Capital Note").

The issuance of any additional Membership Units to the non-defaulting Member in the event of a Capital Contribution Default will be based on a valuation equal to the lesser of the post-money valuation following the latest Capital Contribution or at the non-defaulting Member's option, the fair market value of the Membership Units as determined by a third party appraiser selected by the non-defaulting Member and approved by the defaulting Member (whose approval shall not be unreasonably withheld, delayed or conditioned).

Section 4.2 **No Interest on Contributions; No Withdrawal**. No interest shall accrue on any Capital Contribution and no Member shall have the right to withdraw or to be repaid any capital contributed by such Member except as otherwise specifically provided in this Agreement and permitted by applicable law.

## ARTICLE V MANAGEMENT

## Section 5.1 Management of the Company; Board.

- (a) <u>General</u>. Except as provided for herein or by the Business Code, the management, control and operation of the Company shall be vested exclusively in the Board. The Board, acting as a body pursuant to this Agreement, shall constitute a "manager" for purposes of the Business Code. No Manager, in such capacity, acting singly or with any other Manager, shall have any authority or right to represent, make public statements on behalf of, advocate or take positions on any matter on behalf of, act on behalf of or bind the Company other than by exercising the Manager's voting power as a member of the Board, unless specifically authorized by the Board in each instance.
  - (b) Number of Managers; Composition.
    - (i) Except as otherwise provided herein in the event of a Capital Contribution Default, a Class B Default or a Class B Conversion, the voting members of the Board shall be comprised of four (4) voting members who shall be divided into two (2) classes as described below:
      - (A) Two (2) Class A Managers, who shall be appointed by the Class A Member (the " *Class A Managers*"); and
      - (B) Two (2) Class B Managers, who shall be appointed by the Class B Member and shall initially include Lee Roy Mitchell (the "*Class B Managers*").
        - (ii) The Board as of the Effective Date is set forth on Exhibit B.
    - (iii) On the death, resignation or removal of any Class A Manager, the Class A Member may appoint an individual representative of the Class A Member to replace such Class A Manager. Subject to Section 5.1(b) (iv), on the death, resignation or removal of any Class B Manager, the Class B Member may appoint an individual representative of the Class B Member to replace such Class B Manager.
    - (iv) In the event of a Class B Default or a Class B Conversion, (1) there shall only be one class of Managers, each of whom shall be appointed by the Class A Member, and (2) the Class A Member shall have the sole right to remove any Manager and fill any vacancy resulting from the death, resignation or removal of any Manager or any increase in the number of Managers.

- (c) Qualifications for Service. Managers need not be residents of the State of Texas.
- (d) <u>Election; Tenure</u>. The Class A Managers shall be appointed by the Class A Member and, except as otherwise provided in this Agreement, the Class B Managers shall be appointed by the Class B Member; provided, however, that in the event that Lee Roy Mitchell no longer serves as a Class B Manager due to death, disability, retirement, or any other reason, before the first Facility has been in operation for a full year, Gary Witherspoon shall automatically be appointed as the successor Class B Manager, without any action required by the Class B Member. Each Manager shall hold office until a qualified successor has been appointed or until the Manager's death, resignation, or removal, if sooner. Except as otherwise provided in this Agreement, any Manager may be removed at any time by the Member appointing such Manager in which case, such Member shall promptly appoint a qualified successor Manager of the same class as the removed Manager.
- (e) Quorum; Proxy; Voting; Meetings. The Board shall meet at least quarterly during each Fiscal Year. A majority of the then serving Managers, either present (in person or by teleconference), shall constitute a quorum for purposes of any meeting of the Board, except as otherwise provided by law or this Agreement; provided, however, a Class A Manager must be present to constitute a quorum. Unless otherwise expressly required by the non-waivable provisions of the Business Code or this Agreement, the act or decision of Managers, holding at least fifty-one percent (51%) of the total voting power of the Board, shall be the act or decision, as applicable, of the Board. Meetings of the Managers may be held at such time or times and such place or places as shall be determined from time to time by the Managers, provided that notice of such meeting(s) is sent to all Managers in advance thereof. Meeting dates for regular meetings may be determined in advance by the Board. Special meetings of the Board may be called by the Class A Member, the Class B Member, the Chair of the Board, the Secretary of the Company or any two (2) Managers on at least twenty-four (24) hours' notice to the Board. To the extent practicable, the notice should state the purpose of and business to be transacted at, such meeting. Attendance of a Manager at a meeting constitutes a waiver of notice of the meeting, except where a Manager attends a meeting for the express purpose of objecting to the transaction of any business on the ground that the meeting is not lawfully called or convened. A Manager who is present at a meeting of the Board at which action on any matter is taken shall be presumed to have assented to the action unless his or her dissent is entered in the minutes of the meeting or unless he or she files his or her written dissent to the action with the secretary of the meeting before the adjournment thereof, or delivers the dissent to the Company immediately after the adjournment of the meeting. A Manager who votes in favor of the action waives his or her right to dissent. Each Manager shall be entitled to one (1) vote on the Board.
- (f) <u>Attendance at Meetings</u>. Attendance at any meeting of the Board may be in person, by telephonic or by any other electronic means such that each attendee may be heard by, and may hear, each other attendee.

- (g) Action by Written Consent. Any action required or permitted to be taken or authorized at a meeting of the Board may be taken or authorized without a meeting if, either (i) at least twenty-four (24) hours prior notice of such action is provided to the Board and prior or subsequent to the action, a consent or consents thereto is signed by the minimum number of Managers that would be necessary to authorize or take such action, and is filed with the records of the Company or (ii) a consent or consents to such action is signed by all of the Managers.
- (h) Conduct of Business; Record of Action. The Board shall appoint the Chair of the Board; provided, however, in the event Gary Witherspoon replaces Lee Roy Mitchell as a Class B Manager, pursuant to Section 5.1(d), then the individual who is currently serving as Vice Chair shall be appointed as Chair of the Board, and Gary Witherspoon shall be appointed the Vice Chair. The Chair of the Board shall preside over all meetings and the Secretary of the Company shall act as secretary of all meetings of the Board, but in the absence of the Secretary, the Chair may appoint any person to act as secretary of the meeting. The Chair shall determine the order of the business and the procedure at any meeting of the Board. Written minutes of the business transacted at any Board meeting and meetings of any committees thereof shall be made and circulated to each Manager for his or her approval. Such written minutes, together with any and all written consents executed by the Managers pursuant to Section 5.1(g) shall be maintained as part of the records of the Company pursuant to Section 5.4. The initial Chair is set forth on Exhibit B.
- (i) <u>Compensation; Expense Reimbursement</u>. Managers shall receive no compensation for their services, including, without limitation, their attendance at meetings of the Board, except as may be approved by unanimous vote of the Board. Managers shall be reimbursed for necessary, documented and reasonable authorized expenses incurred on behalf of the Company.
- (j) <u>Resignation</u>. Any Manager may resign at any time by giving at least thirty (30) days' prior written notice to the other Manager(s) of the Company. The resignation of any Manager shall take effect on the thirtieth (30th) day following receipt of such notice or at such later time as shall be specified in the notice; and, unless otherwise specified in the notice, the acceptance of the resignation shall not be necessary to make it effective.
- (k) <u>Filling Vacancies</u>. Any vacancy resulting from the death, resignation or removal of a Class A Manager may be filled by the Class A Member. Subject to Section 5.1(b)(iv), any vacancy resulting from the death, resignation or removal of a Class B Manager may be filled by the Class B Member. The Class A Managers may be executive officers of the Class A Member or its Affiliates. The Class B Managers may be executive officers of the Class B Member or its Affiliates. Subject to <u>Section 5.1(b)(iv)</u>, any vacancy occurring on the Board as a result of an increase in the number of Managers constituting the Board may be filled by the mutual consent of the Class A Member and the Class B Member.

## Section 5.2 **Authority of the Board.**

(a) Subject to Section 5.3 and except as otherwise provided in this Section 5.2(a), the Board, by a majority vote of the Board, shall have the sole power on behalf and in the name of the Company to carry out any and all of the objects and purposes of the Company and to perform all acts which they may, in their discretion, deem necessary or desirable.

(b) Subject to Section 5.3, the Board may delegate to any Person or Persons all or any of the powers, rights, privileges, duties and discretion vested in it in this ARTICLE V, and such delegation may be made upon such terms and conditions as the Board shall determine; except that no such delegation shall modify the obligations, liabilities or responsibilities of the Board under the Business Code and under this Agreement.

## Section 5.3 **Restrictions on Authority of the Board.**

- (a) Notwithstanding anything to the contrary herein, except as provided in <u>Section 5.3(b)</u>, the following activities of the Company shall require the approval of all of the Class A Managers and, so long as neither a Class B Default nor a Class B Conversion has occurred, all of the Class B Managers. In the event of a Class B Default or a Class B Conversion, the following activities may be approved solely by the consent of a majority of the Class A Managers. Such activities shall include the following:
  - (i) authorizing or issuing additional Membership Units or any equity in, or any options, warrants or other rights to acquire any equity in, the Company or requiring any Member to make any Capital Contribution;
  - (ii) any conversion of the Company to another form of entity or other material corporate reorganization of the Company;
  - (iii) any change in, or addition to, the designation of any rights, powers and/or preferences of any class of Membership Interests;
  - (iv) any amendment or modification to this Agreement, the Management Services Agreement or the Theater Services Agreement;
    - (v) the appointment of any Officers pursuant to Section 5.6;
  - (vi) the incurrence of any indebtedness (other than pursuant to a Capital Note) or entering into or consummating any other financing transaction or capital lease or creating any Lien against the assets of the Company, in any case for an amount in excess of \$100,000;
  - (vii) incurring any costs, expenses, debts, liabilities, obligations or otherwise that result in a variance of greater than five percent (5%) of any category for the entire fiscal year in the Approved Budget as previously approved by the Board;
    - (viii) distributing any Distributable Cash to the Members other than as provided in this Agreement;
  - (ix) approving the design of the Facility, including without limitation, the motion picture theater auditoriums incorporated in any Facility;

- (x) forming any subsidiary, joint venture or other affiliated entity or authorizing any existing or future affiliate of the Company to operate the business of the Company in a new location or making any investment in any non-controlled entity in excess of \$100,000;
- (xi) approving any acquisition (by merger or otherwise), leasing or licensing by the Company or sale not in the ordinary course of business of any assets of the Company and the locations thereof including site selection for Facilities and decisions to lease or purchase Facility locations;
  - (xii) entering into, waiving any rights under, modifying, or terminating any Material Contract;
- (xiii) engaging any professional advisors (e.g., attorneys (including in connection with real estate, liability and employment matters), financial advisors, auditors, etc.) for, or on behalf of, the Company for a cost in excess of \$25,000 for any such professional advisor;
- (xiv) approving the Company's business plan, Approved Budget, capital projects and expenditures, financial statements, and tax returns, and any material change in the Approved Budget, accounting or tax practices, procedures or policies of the Company;
- (xv) entering into any agreement or transaction with any Member or any officer, director, employee, Affiliate or relative of any Member or any Manager or their respective Affiliates;
- (xvi) commencing, settling, or withdrawal from any suit, action or legal, administrative or arbitral proceeding in excess of \$50,000 or where equitable relief or injunctive relief is included as part of such suit or settlement;
- (xvii) filing any documents with any regulatory agency, including the Internal Revenue Service, other than the permitted filings specified in the Management Services Agreement;
  - (xviii) any change in the size of the Board;
- (xix) terminating Management Co. or appointing a new third party to provide the services under the Management Services Agreement or terminating or amending the Management Services Agreement;
- (xx) terminating Consultant or appointing a new third party to provide the services under the Theater Services Agreement or terminating or amending the Theater Services Agreement;
  - (xxi) adopting or amending any other form of incentive plan; and

- (xxii) taking any action or authorizing, approving or entering into any binding agreement with respect to any of the foregoing.
- (b) Notwithstanding anything to the contrary herein, the following activities of the Company shall require the approval of each of the Voting Members (provided that if a Class B Default or Class B Conversion has occurred, then only the approval of the Class A Member shall be required):
  - (i) the Company entering into any transactions (other than a Class A Change in Control, a Drag-Along Sale or a Qualified Purchase Offer) which could reasonably be expected to result in a Change in Control;
    - (ii) any amendment or repeal of this Agreement other than as provided in <u>Section 12.2</u>;
  - (iii) any recapitalization, reorganization winding up liquidation or dissolution of the Company, including any distribution of assets pursuant to a plan of dissolution or the filing of any petition under any bankruptcy or other insolvency law; and
  - (iv) engaging the Company in any business other than the business of the Company described in Section 2.7 of this Agreement.
- (c) Notwithstanding anything to the contrary herein, upon the occurrence of a Company Mitchell Trigger Event or a Cinemark Mitchell Trigger Event:
  - (i) no additional Facilities shall be approved by the Board without Class C Member approval; and
  - (ii) the Company shall not enter into any agreement or arrangement or transaction with the Class A Member or any of its affiliates, directors, managers, employees or Affiliates that is not in the ordinary course of business, armslength and on commercially reasonable terms, without Class C Member approval.
- Section 5.4 **Books and Records**. The Board shall keep or cause to be kept at the address of the Company (or at such other place as the Board shall advise the Members in writing) full and accurate books and records of the Company. A Member, on written request stating the purpose therefor, may examine at any reasonable time, for any proper purpose and at such Member's expense, the records required to be kept under this <u>Section 5.4</u> and other information regarding the business, affairs, and financial condition of the Company as is just and reasonable for such Member to examine.
- Section 5.5 **Committees**. The Board shall have the power to create one (1) or more committees, each of which shall have such purposes, functions, duties and authority as the Board may determine (each, a "*Committee*" and, collectively, the "*Committees*"). The composition and membership of each Committee shall be as determined by the Board. All Committee members shall serve one (1) year terms, unless otherwise determined by the Board.

Section 5.6 **Officers**. The Board shall have the power to appoint one (1) or more officers of the Company, who shall be responsible for the day-to-day operations of the Company, subject to the direction and control of the Board (each, an "Officer" and, collectively, the "Officers"). The Company shall have such number of Officers as the Board shall determine. No Officer need be a resident of the State of Texas. Officers are not "managers," as that term is used in the Business Code. Any Officers so designated shall have the authority to perform such duties as the Board delegates to them. The Board may assign titles to particular Officers. Each Officer holds office until his successor has been duly designated and qualified or until his death or until he resigns or has been removed in the manner hereinafter provided. Any number of offices may be held by the same person. Officers shall not be considered employees of the Company. Any Officer may resign as such at any time. Resignations are to be made in writing and take effect at the time specified therein, or if no time be specified, at the time of its receipt by the Board. The acceptance of a resignation is not necessary to make it effective, unless expressly so provided in the resignation. Any Officer may be removed as such, either with or without cause, by the Board whenever in their judgment the best interests of the Company will be served thereby. Any vacancy occurring in any office of the Company shall be filled by the Board.

Section 5.7 **Management Services Agreement.** Concurrently with the execution and delivery of this Agreement, the Company shall enter into the Management Services Agreement with Management Co. pursuant to which Management Co. shall provide the development, management and staffing services described in the Management Services Agreement.

Section 5.8 **Theater Services Agreement.** Concurrently with the execution and delivery of this Agreement, the Company shall enter into the Theater Services Agreement with Consultant pursuant to which Consultant shall provide the theater services described in the Theater Services Agreement. The Parties hereto acknowledge and agree that so long as the Theater Services Agreement remains in effect, the ticket prices charged for admission to any motion picture at any Facility shall be determined exclusively by Consultant.

## ARTICLE VI TRANSFERS

### Section 6.1 **Conditions to Transfers.**

(a) Restrictions on Transfers of Membership Units. No Member shall Transfer any interest in any Membership Units except: (i) a Transfer by a Capital Member to a Permitted Transferee as expressly set forth in Section 6.1(b) hereof; (ii) with the prior written consent of the Board (which consent may be withheld for any reason or for no reason); or (iii) as expressly permitted by this Agreement with respect to a Change in Control of the Company, a Drag-Along Sale pursuant to Section 10.1 or the exercise of the Class A Member Call Option pursuant to Section 10.2 (each of (i), (ii) or (iii) above, a "Permitted Transfer"). Any purported Transfer of Membership Units in violation of this Agreement shall be null and void. The Company shall not be required (a) to transfer on its books Membership Units which have been Transferred in violation of any of the provisions set forth in this Agreement or (b) to treat as an owner of such Membership Units, to afford the right to vote or to pay distributions to, any Transferee to whom Membership Units have been Transferred (or purported to have been Transferred) in violation of the provisions set forth in this Agreement.

- (b) Permitted Transfers. Membership Units may be Transferred as follows without the Board's approval (each such transferee being a "Permitted Transferee"): (i) by a Member or Permitted Transferee to the Company; (ii) by a Capital Member or any Permitted Transferee to any member, partner or shareholder of such Capital Member or Permitted Transferee and, if such member, partner or shareholder is a corporation, limited liability company or partnership, such member may transfer such Capital Units to the ultimate holders of the equity interest in such member, or to a liquidating trust for their benefit; (iii) by a Capital Member to any Person which such Capital Member or any of its Affiliates, directly or indirectly, has the sole power to direct or cause the direction of the management and policies of such Person, whether through the ownership of Capital Units, by contract or otherwise; (iv) by a Capital Member to any of its Affiliates; (v) by a Capital Member to the remaining Capital Member or (vi) by the Class B Member or Management Parent to the Mitchell Foundation. For the avoidance of doubt, no Capital Member may transfer any portion of its Capital Units to a competitor of the remaining Capital Member without such remaining Member's prior written consent.
- (c) <u>Rights of First Offer</u>. Each Capital Member will have a right of first offer (the "*ROFO*") if the other Capital Member (the "*Offering Member*") proposes to Transfer, directly or indirectly, all or any portion of the Membership Units owned by it (the "*Offered Units*") other than pursuant to a Permitted Transfer or a Drag-Along Sale. Each time the Offering Member proposes to so Transfer any Offered Units, the Offering Member will first offer to sell the Offered Units to the other Capital Member in accordance with the provisions of this <u>Section 6.1(c)</u>.
  - (i) The Offering Member will give written notice (the "Offering Member Notice") to the Company and the other Capital Member stating the Offering Member's intention to Transfer the Offered Units and specify the Membership Interest to be Transferred as the Offered Units and the material terms and conditions, including price, pursuant to which the Offering Member proposes to Transfer the Offered Units.
  - (ii) The Offering Member Notice will constitute the Offering Member's offer to Transfer the Offered Units to the other Capital Member, which offer will be irrevocable for a period of 60 days (the "ROFO Notice Period").
  - (iii) By delivering the Offering Member Notice, the Offering Member will be deemed to represent and warrant to the Company and the other Capital Member that: (1) the Offering Member has full right, title and interest in and to the Offered Units; (2) the Offering Member has all necessary power and authority and has taken all necessary action to sell such Offered Units as contemplated by this Section 6.1(c); and (3) the Offered Units will be Transferred free and clear of any and all Liens other than restrictions on transfer arising under state or federal securities laws or under this Agreement.

- (iv) Upon receipt of the Offering Member Notice, the other Capital Member will have until the expiration of the ROFO Notice Period to offer to purchase all (but not less than all) of the Offered Units by delivering a written notice (a "ROFO Acceptance Notice") to the Offering Member and the Company stating that it offers to purchase such Offered Units on the terms specified in the Offering Member Notice. Any ROFO Acceptance Notice so delivered will be binding upon delivery and irrevocable by the applicable Capital Member. A Capital Member that does not deliver a ROFO Offer Notice during the ROFO Notice Period will be deemed to have waived all such Capital Member's rights to purchase the Offered Units under this Section 6.1(c) (iv). If the other Capital Member timely delivers a ROFO Acceptance Notice, such Capital Member (the "Purchasing Member") will be entitled to purchase the Offered Units.
- (v) If the other Capital Member fails to deliver a ROFO Acceptance Notice in accordance with Section 6.1(c)(iv), the Offering Member may, during the 90-day period following the expiration of the ROFO Notice Period (which period may be extended for a reasonable time to the extent reasonably necessary to obtain any governmental approvals) (the "Waived ROFO Transfer Period"), and subject to compliance with the provisions of Section 6.1(d), Transfer all of the Offered Units to an Independent Third Party at a higher price and otherwise on terms and conditions no more favorable to the Independent Third Party than those specified in the Offering Member Notice. Any noncash consideration offered by an Independent Third Party will be valued at its fair market value, as reasonably agreed by the Offering Member and the other Capital Member, and failing such agreement, as determined by an independent third party appraiser selected by the Offering Member and reasonably acceptable to the other Capital Member (the cost for which third party appraiser will be paid by the Offering Member). Any Independent Third Party to whom Offered Units are Transferred in accordance with this Section 6.1(c)(v) will become a Substitute Member if the Offering Member so designates, in a written instrument delivered to the Company and the other Capital Member, and such Independent Third Party executes a joinder to this Agreement as required by Section 6.3. If the Offering Member does not Transfer the Offered Units within such period or, if such Transfer is not consummated within the Waived ROFO Transfer Period, the Offered Units in question will once again become subject to the preceding provisions of this Section 6.1(c) and the Offering Member will no longer be permitted to Transfer such Offered Units without again fully complying with such provisions.
- (vi) The Offering Member and the Purchasing Member will take all actions as may be reasonably necessary to consummate any sale to the Purchasing Member contemplated by this <u>Section 6.1(c)</u>, including entering into agreements and delivering certificates and instruments and consents as may be deemed necessary or appropriate by the Company.

- (vii) At the closing of any sale and purchase to a Purchasing Member pursuant to this <u>Section 6.1(c)</u>, the Offering Member will deliver to the Purchasing Member an Assignment of the Offered Units to be sold (if any), accompanied by any other evidence of transfer and all necessary transfer taxes paid and stamps affixed, if necessary, against receipt of the purchase price therefor from such Purchasing Member by wire transfer of immediately available funds into an account specified by the Offering Member (or such other method of payment as may be agreed upon by the Offering Member).
- (d) <u>Rights of First Refusal</u>. Each Capital Member will have a right of first refusal (the "*ROFR*") if the other Capital Member (the "*Transferring Member*") proposes to Transfer, directly or indirectly, all or any portion of the Membership Units owned by it (the "*Subject Units*") other than pursuant to a Permitted Transfer or a Drag-Along Sale. Each time the Transferring Member proposes to so Transfer any Subject Units, the Transferring Member will first offer to sell the Subject Units to the other Capital Member in accordance with the provisions of this <u>Section 6.1(d)</u>.
  - (i) The Transferring Member will give written notice (the "Transfer Notice") to the Company and the other Capital Member stating the Transferring Member's intention to Transfer the Subject Units and specify the Membership Interest to be Transferred as the Subject Units, the identity of the proposed purchaser of the Subject Units (the "Proposed Purchaser") and a complete description of all terms and conditions of the proposed sale, including price and type of consideration, pursuant to which the Transferring Member proposes to Transfer the Subject Units.
  - (ii) The Transfer Notice will constitute the Transferring Member's offer to Transfer the Subject Units to the other Capital Member, which offer will be irrevocable for a period of 60 days (the "*ROFR Notice Period*").
  - (iii) By delivering the Transfer Notice, the Transferring Member will be deemed to represent and warrant to the Company and the other Capital Member that: (1) the Transferring Member has full right, title and interest in and to the Subject Units; (2) the Transferring Member has all necessary power and authority and has taken all necessary action to sell such Subject Units as contemplated by this Section 6.1(d); and (3) the Subject Units will be Transferred free and clear of any and all Liens other than restrictions on transfer arising under state or federal securities laws or under this Agreement.
- (iv) Upon receipt of the Transfer Notice, the other Capital Member will have until the expiration of the ROFR Notice Period to offer to purchase all (but not less than all) of the Subject Units by delivering a written notice (a "ROFR Acceptance Notice") to the Transferring Member and the Company stating that it offers to purchase such Subject Units on the terms specified in the Transfer Notice. Any ROFR Acceptance Notice so delivered will be binding upon delivery and irrevocable by the applicable Capital Member. A Capital Member that does not deliver a ROFR Acceptance Notice during the ROFR Notice Period will be deemed

to have waived all such Capital Member's rights to purchase the Subject Units under this <u>Section 6.1(d)(iv)</u>. If the other Capital Member timely delivers a ROFR Acceptance Notice, such Capital Member (the "*Exercising Member*") will be entitled to purchase the Subject Units.

(v) If the other Capital Member fails to deliver a ROFR Acceptance Notice in accordance with Section 7.1(d)(iv), the Transferring Member may, during the 90-day period following the expiration of the ROFR Notice Period (which period may be extended for a reasonable time to the extent reasonably necessary to obtain any governmental approvals) (the "Waived ROFR Transfer Period"), and subject to compliance with the provisions of Section 6.1(c), Transfer all of the Subject Units to the Proposed Purchaser on terms and conditions no more favorable to the Proposed Purchaser than those specified in the Transfer Notice. Any noncash consideration offered by the Proposed Purchaser will be valued at its fair market value, as reasonably agreed by the Transferring Member and treasonably acceptable to the other Capital Member (the cost for which third party appraiser selected by the Transferring Member). Any Proposed Purchaser to whom Subject Units are Transferred in accordance with this Section 6.1(d)(v) will become a Substitute Member if the Transferring Member so designates, in a written instrument delivered to the Company and the other Capital Member, and such Proposed Purchaser executes a joinder to this Agreement as required by Section 6.3. If the Transferring Member does not Transfer the Subject Units within such period or, if such Transfer is not consummated within the Waived ROFR Transfer Period, the Subject Units in question will once again become subject to the preceding provisions of this Section 6.1(d) and the Transferring Member will no longer be permitted to Transfer such Subject Units without again fully complying with such provisions.

(vi) The Transferring Member and each Exercising Member will take all actions as may be reasonably necessary to consummate any sale to the Exercising Member contemplated by this Section 6.1(d), including entering into agreements and delivering certificates and instruments and consents as may be deemed necessary or appropriate by the Company.

(vii) At the closing of any sale and purchase pursuant to this <u>Section 6.1(d)</u>, the Transferring Member will deliver to the Exercising Member an Assignment of the Subject Units to be sold (if any), accompanied by any other evidence of transfer and all necessary transfer taxes paid and stamps affixed, if necessary, against receipt of the purchase price therefor from such Exercising Member by wire transfer of immediately available funds into an account specified by the Transferring Member (or such other method of payment as may be agreed upon by the Transferring Member).

Section 6.2 **Effect of Membership Units in Hands of the Transferee**. Membership Units that are Transferred pursuant to this <u>ARTICLE VI</u> shall thereafter continue to be subject to all restrictions on Transfer contained in this Agreement except as otherwise provided in this <u>ARTICLE VI</u>. Without limiting the generality of the foregoing, the Transferee must comply with the provisions of this <u>ARTICLE VI</u> if such Transferee shall propose to Transfer any such Membership Units, as if such Transferee were a Member. If any Membership Units are Transferred to a Transferee that is not admitted as a Member pursuant to <u>Section 6.1(c)(v)</u>, <u>Section 6.1(d)(v)</u> and <u>Section 6.5</u>, then the Transferee shall issue an irrevocable proxy to the

Offering Member or Transferring Member, as the case may be, to vote such Transferred Units with respect to all matters submitted to the Voting Members. The right of first offer under <u>Section 6.1(c)</u> and the right of first refusal under <u>Section 6.1(d)</u> does not inure to the benefit of any Transferee of Membership Units and may not be exercised by any Transferee of Membership Units.

Section 6.3 **Joinder**. Any Person (including any Permitted Transferee) to whom Membership Units are to be Transferred shall execute and deliver, as a condition to such Transfer, all documents deemed reasonably necessary by the Company, in consultation with its counsel, to evidence such party's joinder in, acceptance of, and agreement with, the obligations with respect to the Membership Units contained in this Agreement. Any subsequent holder of the Membership Units shall execute and deliver to the Class A Member a blank Assignment of Membership Units pursuant to Section 10.3 as a condition to the transfer of the Membership Units. Each such transferor of Membership Units shall, prior to the Transfer, cause the Transferee thereof to so execute and deliver such documents.

Section 6.4 **Imposition of Restrictions**. Membership Units that are Transferred shall thereafter continue to be subject to all restrictions and obligations imposed by this Agreement with respect to Membership Units and Transfers thereof.

Section 6.5 **Limitation on Admission**. Notwithstanding anything to the contrary that may be expressed or implied in this Agreement, upon a Transfer of Membership Units permitted pursuant to this <u>ARTICLE VI</u>, no Transferee of such Membership Units (including a Permitted Transferee other than an Affiliate or the Mitchell Foundation) shall be admitted into the Company as an additional or Substituted Member or have any rights to participate in the management of the business and affairs of the Company as a Member without the prior express written consent of the Board which consent may be granted or withheld in the sole discretion of the Board for any reason or for no reason. A Transferee of a Membership Unit not admitted into the Company as an additional or Substituted Member shall not be considered a voting Member.

Section 6.6 **Other Limitations**. Notwithstanding anything in this <u>ARTICLE VI</u> to the contrary: (a) no Transfer of Class A Membership Units shall be made by the Class A Member to a Transferee engaged in a Class B Competing Business without the consent of the Class B Member or Class C Member, as applicable, unless such Transfer is pursuant to a Drag-Along Sale resulting from a Class A Change in Control; (b) no Transfer of Class B Membership Units or Class C Membership Units shall be made by the Class B Member or Class C Member, respectively, to a Transferee engaged in a Class A Competing Business without the consent of the Class A Member; and (c) no Transfer of Membership Units shall be permitted, and no Transferee of Membership Units shall be admitted to the Company as a Member, if, (x) such Transfer is not expressly permitted pursuant to <u>Section 6.1(a)</u> or (y) in the opinion of the Board, such transfer or admission alone or in conjunction with one or more other transfers or admissions would: (i) result in a violation of applicable securities laws, (ii) result in the Company being taxable as a corporation, (iii) if unknowingly waived by the Board, cause a termination of the Company under Section 708(b) (1)(B) or any other section of the Code, (iv) result in the Company becoming a "publicly traded partnership" within the meaning of Sections 469(k)(2) and 7704(b) of the Code, (v) be an event which would constitute a violation or breach

(or, with the giving of notice or passage of time, would constitute a violation or breach) of any law, regulation, ordinance, agreement or instrument by which the Company, or any of its properties or assets, is bound, or (vi) require the Company to register under the Investment Company Act of 1940, as amended.

# ARTICLE VII CAPITAL ACCOUNTS; TAX ALLOCATIONS; TAX MATTERS

#### Section 7.1 Capital Accounts.

- (a) A separate Capital Account shall be established for each Member on the books of the Company on the date on which such Member first makes its Capital Contribution and shall be maintained in the manner set out in the definition of Capital Account. The foregoing provisions and other provisions of this Agreement relating to the maintenance of Capital Accounts are intended to comply with Section 1.704-1(b) of the Treasury Regulations and shall be interpreted and applied in a manner consistent with such Treasury Regulations. Upon the Transfer of any Membership Units in accordance with this Agreement, the Capital Account of the transferor Member that is attributable to the transferred Membership Units will be carried over to the transferee Member.
- (b) Any loans made by any Member to the Company shall not be deemed to result in an increase (or decrease) in the Capital Contribution, Capital Account or Membership Units of such Member; rather, such loans shall constitute an indebtedness of the Company to the Member making such loan.
- (c) Except as otherwise provided in this Agreement or as may be approved by the Board acting in its sole and absolute discretion, no Member shall demand or receive a return of its Capital Contribution. Under circumstances requiring a return of any Capital Contribution, no Member shall have the right to receive property other than cash, except as may be specifically provided herein or as may be determined by the Board acting in its reasonable discretion.
- (d) Except as expressly otherwise provided in this Agreement and/or in a separate services or employment agreement or as may be determined by the Board acting in its sole and absolute discretion, no Member shall receive any interest, salary or draw with respect to its Capital Contribution or its Capital Account or for services rendered on behalf of the Company or otherwise in its capacity as a Member.
- (e) The Members shall not be personally liable for the debts, liabilities, contracts or any other obligations of the Company. No Member shall be required to lend any funds to the Company.

#### Section 7.2 Allocations of Profits and Losses.

- (a) <u>Profits</u>. Subject to and after making any special allocations pursuant to <u>Section 7.4</u>, any remaining Profits for each fiscal year shall be allocated among the Members in the following manner:
  - (i) First. First, among the Members in proportion to, and to the extent of, Losses previously allocated to each Member that have not been offset by allocations of Profits pursuant to this Section 7.1(a)(i); and
  - (ii) Second. Thereafter, to the Members pro rata in proportion to their respective Membership Interests.
- (b) <u>Losses</u>. Subject to and after making any special allocations pursuant to <u>Section 7.3</u>, any remaining Loss for each fiscal year shall be allocated among the Members in proportion to their respective Membership Interests.

Notwithstanding the foregoing, no Member shall receive an allocation of Loss pursuant to this <u>Section 7.2(b)</u> which would cause such Member to have an Adjusted Capital Account Deficit at the end of any fiscal year. Loss not allocated to a Member pursuant to this <u>Section 7.2(b)</u> shall be allocated to the other Members pro rata in proportion to their respective Membership Interests to the extent an Adjusted Capital Account Deficit balance would not be created or increased with respect to the other Members. If no Member may receive an additional allocation of Losses pursuant to this <u>Section 7.2(b)</u> such additional Losses shall be allocated to the Member or Members who bear the economic risk of loss pursuant the Treasury Regulations, as determined by the Board.

# Section 7.3 **Special Allocations**. The following special allocations shall be made in the following order:

- (a) <u>Minimum Gain Chargeback</u>. Notwithstanding any other provision of this Agreement, if there is a net decrease in Company Minimum Gain or in any Member Minimum Gain during any taxable year or other period, prior to any other allocation pursuant hereto, the Members shall be specially allocated items of Profit for such year (and, if necessary, subsequent years) in an amount and manner required by Treasury Regulations Section 1.704-2(f) or 1 .704-2(i)(4). The items to be so allocated shall be determined in accordance with Treasury Regulations Section 1.704-2. This <u>Section 6.3(a)</u> is intended to comply with the minimum gain chargeback requirements of Treasury Regulations <u>Section 7.3(a)</u>, and will be interpreted consistently with the Treasury Regulations and will be subject to all exceptions provided therein.
- (b) Qualified Income Offset. Any Member who unexpectedly receives an adjustment, allocation or distribution described in Treasury Regulations Section 1.704-l(b)(2)(ii)(d)(4), (5) or (6) which causes or increases an Adjusted Capital Account Deficit shall be allocated items of income or gain in an amount and manner sufficient to eliminate, to the extent required by such Treasury Regulations, the Adjusted Capital Account Deficit of the Member as quickly as possible. This Section 7.3(b) is intended to constitute a "qualified income offset" within the meaning of Treasury Regulations Section 1.704-1 (b)(2)(ii)(d), will be interpreted consistently with the Treasury Regulations and will be subject to all exceptions provided therein.

- (c) <u>Gross Income Allocation</u>. Each Member who has a deficit Capital Account at the end of any Company taxable year that is in excess of the amount the Member is obligated to restore pursuant to any provision of this Agreement, including any amount that it is deemed to be obligated to restore under Treasury Regulations Section 1.704-2(g)(l) and Section 1.704- 2(i)(5), will be specially allocated items of Company income and gain in the amount of the excess as quickly as possible.
- (d) <u>Nonrecourse Deductions</u>. Nonrecourse Deductions for any taxable year or other period will be specially allocated among the Members pro rata in proportion to their respective Membership Interests in the Company.
- (e) <u>Member Nonrecourse Deductions</u>. Any Member Nonrecourse Deductions for any taxable year or other period will be allocated to the Member who bears the economic risk of loss with respect to the Member Nonrecourse Debt to which such Member Nonrecourse Deductions are attributable in accordance with principles under Treasury Regulations Section 1.704-2(i).
- (f) <u>Code Section 754 Adjustments</u>. To the extent an adjustment to the adjusted tax basis of any Company asset under Sections 734(b) or 743(b) of the Code is required to be taken into account in determining Capital Accounts under Treasury Regulations Section 1.704-1(b)(2)(iv)(m), the amount of the adjustment to the Capital Accounts will be treated as an item of gain (if the adjustment increases the basis of the asset) or loss (if the adjustment decreases the basis), and the gain or loss will be specially allocated to the Members in a manner consistent with the manner in which their Capital Accounts are required to be adjusted under Treasury Regulations Section 1.704-1(b)(2)(iv) (m).
- (g) <u>Curative Allocations</u>. The "Regulatory Allocations" consist of allocations made to a Member (or predecessor) under <u>Section 7.3(a)</u> through <u>Section 7.3(j)</u>. Notwithstanding any other provision of this <u>Section 7.3</u> other items of income, gain, loss and deduction will be allocated among the Members so that, to the extent possible, the net amount of those allocations of other items and the Regulatory Allocations to each Member will be equal to the net amount that would have been allocated to the Member if the Regulatory Allocations had not occurred. The Board shall take into account future Regulatory Allocations under <u>Section 7.3(a)</u> that, although not yet made, are likely to offset Regulatory Allocations under <u>Section 7.3(d)</u> and <u>Section 7.3(e)</u>.
- (h) Compliance with Section 704(c). In accordance with Section 704(c) of the Code and the applicable Treasury Regulations thereunder, income, gain, loss and deduction with respect to any Company property for which its adjusted tax basis differs from its Book Basis will, solely for tax purposes, be allocated among the Members so as to take account of any variation between the adjusted tax basis of such property to the Company for federal income tax purposes and the Book Basis of such Property. Any elections or other decisions relating to allocations under this Section 7.3(h) will be made in any manner which the Board shall determine is consistent with Section 704(c) of the Code and the Treasury Regulations promulgated thereunder and reasonably reflects the purpose and intention of this Agreement. Allocations under this Section 7.3(h) are solely for purposes of federal, state and local taxes and will not affect, or in any way be taken into account in computing, any Member's Capital Account or share of Profits, Losses or other items or distributions under any provision of this Agreement.

- (i) <u>Sale of Assets</u>. Notwithstanding <u>Section 7.2</u>, any income, gain, loss or deduction of the Company resulting from the sale, exchange or other disposition of Company assets shall be allocated as follows: (i) first to each Member in the minimum amount required to cause each Member's positive Capital Account balance to be in the same ratio as their respective Membership Interests, in proportion to such required amounts, and (ii) thereafter, to the Members in proportion to their Membership Interests.
- (j) Intent of Allocations. The parties intend that the foregoing allocation provisions of this Section 7.3 shall produce final Capital Account balances of the Members that will permit liquidating distributions under Section 8.3 to be made in a manner identical to the priorities set forth in Section 7.9. To the extent that the allocation provisions of this Section 7.3 would fail to produce such final Capital Account balances, (i) such provisions shall be amended by the Board if and to the extent necessary to produce such result and (ii) Net Profits and Net Losses of the Company for prior open years (or items of gross income, gain, loss and deduction of the Company for such years) shall be reallocated by the Board among the Members to the extent it is not possible to achieve such result with allocations of items of, income (including gross income and gain), deduction and loss for the current year and future years. This Section 7.3 shall control notwithstanding any other provision of this Agreement or the reallocation or adjustment of taxable income, taxable loss or items thereof by the Internal Revenue Service or any other taxing authority.
- (k) <u>Member Acknowledgment</u>. The Members agree to be bound by the provisions of this <u>Section 7.3</u> in reporting their shares of Company income and loss for income tax purposes.

#### Section 7.4 Other Allocation Rules.

- (a) For purposes of determining the Profits, Losses, or any other items allocable to any period, Profits, Losses, and any such other items shall be determined on a daily, monthly, or other basis, as determined by the Tax Matters Member, subject to the prior approval of the Board, using any permissible method under Code Section 706 and the Treasury Regulations promulgated pursuant thereto.
- (b) Notwithstanding any other provision of this Agreement, no Member shall have any obligation to restore any deficit balance in such Member's Capital Account.

#### Section 7.5 Tax Allocations.

(a) For federal, state and local income tax purposes, each item of income, gain, loss, deduction and credit of the Company shall be allocated among the Members as nearly as possible in the same manner as the corresponding item of income, expense, gain or loss is allocated pursuant to the other provisions of this <u>ARTICLE VII</u>. It is intended that the Capital Accounts will be maintained at all times in accordance with Section 704 of the Code and applicable Treasury Regulations promulgated pursuant thereto, and that the provisions hereof relating to the Capital Accounts be interpreted in a manner consistent therewith. The Tax Matters Member, subject to the prior approval of the Board, shall be authorized to make appropriate amendments to the allocations of items pursuant to this <u>Section 7.5</u> in its discretion

to comply with Section 704 of the Code or applicable Treasury Regulations promulgated pursuant thereto; <u>provided that</u> no such change shall have an adverse effect upon the amount distributable to any Member hereunder.

- (b) Notwithstanding anything else contained in this <u>ARTICLE VII</u>, if any Member has a deficit Capital Account for any fiscal period as a result of any adjustment of the type described in Treasury Regulations Section 1.704-1(b)(2)(ii)(d)(4) through (6), then the Company's income and gain will be specially allocated to such Member in an amount and manner sufficient to eliminate such deficit as quickly as possible. Any special allocation of items of income or gain pursuant to this paragraph shall be taken into account in computing subsequent allocations pursuant to this <u>ARTICLE VII</u> so that the cumulative net amount of all items allocated to each Member shall, to the extent possible, be equal to the amount that would have been allocated to such Member if there had never been any allocation pursuant to this paragraph.
- Section 7.6 **Company Tax Returns** . The Tax Matters Member, subject to the prior approval of the Board, shall cause to be prepared and timely filed all tax returns required to be filed for the Company.
- Section 7.7 **Tax Year and Accounting Matters** . The taxable year of the Company shall be the Fiscal Year of the Company. The Company shall adopt such methods of accounting and file its tax returns on the methods of accounting as determined by the Board.
- Section 7.8 **Tax Elections**. The Tax Matters Member, in the exercise of its reasonable discretion, may cause the Company to make or revoke all tax elections provided for under the Code.
- Section 7.9 **Distributions**. Distributable Cash shall be determined by the Board. Distributable Cash, if any, may be distributed in the following order and priority, as determined in the discretion of the Board:
- (a) First, on or before April 1 of each year, to the Members in an aggregate amount equal to the Tax Distributions for the preceding fiscal year. Subject to applicable law, distributions of property other than cash may be made in the discretion of the Board. Distributions of property shall be valued at the fair market value therein as determined by the Board, less any encumbrances.
- (b) Second, on or before October 1 of each year, an additional Tax Distribution to each Member in an amount equal to the positive difference, if any, between the Tax Distribution made pursuant to Section 7.10 and the amount of Tax Distribution calculated on the basis of the actual tax liability of such Member for the preceding fiscal year if the actual tax liability exceeds the estimated tax liability by ten percent (10%) or more.
- (c) Thereafter, any remaining Distributable Cash shall be distributed to the Members pro rata (in accordance with their respective Membership Interests).

(d) <u>Set-Off.</u> The Company may set-off against and reduce any distributions payable under this <u>ARTICLE VII</u> by any bona fide amounts actually owing from a Member to the Company.

Section 7.10 **Tax Distributions**. To the extent that any Member receives an allocation of taxable income pursuant to Section 7.2 or Section 7.3 but is not otherwise entitled to receive a distribution pursuant to Section 7.9 sufficient to pay, or to permit such Member (including the direct and indirect owners of any entity treated as a partnership for federal income tax purposes) to pay, the federal, state and local income taxes attributable to such allocation, a distribution for the payment of such taxes (a "Tax Distribution") for each applicable period shall be made, prior to making any distribution pursuant to Section 7.9, to such Member in an amount equal to (a) the amount of taxable Net Profits allocated (and, items of taxable income and gain specially allocated) to such Member for the current period pursuant to Section 7.2 and Section 7.3 (determined for the current period as if all distributions for such period were already made pursuant to Section 7.9 without giving effect to this Section 7.10, multiplied by (b) the Assumed Tax Rate. Any Tax Distributions to a Member under this Section 7.10 shall be treated as a preliminary distribution of future amounts due to such Member, and any future distributions otherwise due to such Member hereunder shall be adjusted so that, to the greatest extent possible, all distributions are according to the priorities otherwise set forth in Section 7.9(a). Tax Distributions shall be determined with regard to the allocation related to the basis adjustments pursuant to Code Section 743(b).

#### Section 7.11 Tax Matters Member.

- (a) With respect to tax years beginning on or before December 31, 2017, the Class B Member shall be the "tax matters partner" of the Company pursuant to Code Section 6231(a)(7) (the "*Tax Matters Partner*") and with respect to tax years beginning after December 31, 2017, the Class B Member, or such other qualified Person designated by the Class B Member, shall serve as the "partnership representative" within the meaning of the Revised Partnership Audit Procedures (the "*Partnership Representative*"). The Tax Matters Partner or Partnership Representative, as applicable, shall be referred to herein as the "*Tax Matters Member*." The Tax Matters Member is specifically directed and authorized to take whatever steps it, in its discretion, deems necessary or desirable to perfect such designation, including filing any forms or documents with the Internal Revenue Service and taking such other action as may from time to time be required to perfect such designation under the Treasury Regulations.
- (b) In the event of a Class B Default or Class B Conversion, the Class A Member shall become the Tax Matters Member to the extent permitted by applicable law.
  - (c) The Tax Matters Member shall be subject to the following provisions.
    - (i) The Tax Matters Member shall represent the Company with regard to any tax matters initiated by the Internal Revenue Service or any state, local, or foreign taxing authority.

- (ii) With the Consent of the Board, the Tax Matters Member shall have full authority to act on behalf of the Members and the Company, as provided by applicable law, with respect to any tax audit, investigation, or other proceeding brought by any taxing authority, including administrative settlement and judicial review (a "Tax Proceeding") and, subject to applicable law, the Members and the Company shall be bound by the actions taken by the Tax Matters Member in such capacity. With the consent of the Board, the Tax Matters Member may: (A) execute any agreement or other document relating to, or affecting, any tax matter, (B) make (or decline to make) any election under the Revised Partnership Audit Procedures or any similar provisions of applicable law, and (C) take any action and execute and file any statement or form on behalf of the Company that applicable law permits or requires, in each case in accordance with applicable law, provided, however, the Tax Matters Member shall not have any right to settle or compromise any Tax Proceeding that would have a disproportionate adverse effect on a Member without obtaining the prior written consent of the Member (such consent not to be unreasonably withheld, conditioned, or delayed). The Members acknowledge and agree that it is the intention of the Members to minimize any obligations of the Company and its Members to pay taxes, interest, and penalties in connection with any audit of the Company, including by means of elections or alternatives available under the Revised Partnership Audit Procedures.
  - (iii) Each Member, if it so requests, may participate in a non-binding manner in all Tax Proceedings.
- (iv) With the consent of the Board the Tax Matters Member may employ experienced tax counsel to represent the Company in connection with any Tax Proceeding.
- ( v ) The Tax Matters Member shall keep the Members reasonably informed regarding tax matters related to the Company.
- (vi) The relationship of the Tax Matters Member to the Members is that of a fiduciary, and the Tax Matters Member has a fiduciary obligation to perform its duties as Tax Matters Member in such manner as will serve the best interests of the Company and all of the Company's Members.

- (vii) All costs and expenses incurred in connection with any Tax Proceeding or otherwise incurred by the Tax Matters Member in performing its duties shall be borne by the Company, subject to reimbursement in the reasonable discretion of the Board to ensure that each Member bears its allocable share of such expenses. If the Tax Matters Member incurs expenses in connection with tax matters not affecting each of the Members, then the Company may seek reimbursement from, or charge such expenses to the Capital Accounts of, those Members on whose behalf such expenses were incurred. Each Member shall be responsible for any costs incurred by the Member with respect to any Tax Proceeding against the Member, even though the Tax Proceeding may relate to the Company.
- (viii) To the fullest extent permitted by applicable law, but subject to the limitations and exclusions of Section 11.3, the Company agrees to indemnify the Tax Matters Member and its agents and save and hold them harmless, from and in respect to all (A) fees, costs and expenses in connection with or resulting from any claim, action, or demand against the Tax Matters Member or the Company that arise out of or in any way relate to the Tax Matters Member's status as Tax Matters Member for the Company, and (B) all such claims, actions, and demands and any losses or damages therefrom, including amounts paid in settlement or compromise of any such claim, action, or demand.
- (d) Each Person (herein called a "*Pass-Thru Member*") that holds or controls an interest as a Member on behalf of, or for the benefit of another Person or Persons, or which Pass-Thru Member is beneficially owned (directly or indirectly) by another Person or Persons shall, within thirty (30) days following receipt from the Tax Matters Member of any notice, demand, request for information or similar document, convey such notice or other document in writing to all holders of beneficial interests in the Company holding such interests through such Pass-Thru Member.
- (e) If the Company incurs or is required to pay any liability for any taxes, interest, penalties, other additions to tax, or any related costs and expenses (including reasonable accounting and attorney's fees) of any kind or nature that may be sustained or suffered by the Company or the Members (including any costs incurred in connection with a Tax Proceeding) for any taxable year (a "Company Tax Liability"):
  - (i) With the consent of the Board the Tax Matters Member may cause each Member to pay its allocable share of such Company Tax Liability, as reasonably determined by the Tax Matters Member, and each such Member hereby agrees to pay that amount to the Company and such amount shall not be treated as a Capital Contribution for purposes of any provision of this Agreement that affects distributions to the Members;

- (ii) Any amount not paid by a Member at the time reasonably requested by the Tax Matters Member shall accrue interest at the rate set by the Board (not to exceed the maximum rate permitted by law) until paid, and that Member shall also be liable to the Company for any damages resulting from a delay in making such payment beyond the date such payment is reasonably requested by the Tax Matters Member, and for this purpose the fact that the Company could have paid this amount with other funds shall not be taken into account in determining such damages; and
- (iii) Without reduction in a Member's obligation under paragraphs (d)(i) and (d)(ii), any amount paid by the Company that is attributable to a Member, as determined by the Tax Matters Member in its reasonable discretion with the consent of the Board, and that is not paid by that Member pursuant to paragraphs (d)(i) and (d) (ii)\_may be treated (A) as a distribution to (and shall reduce amounts otherwise distributable to) that Member or (B) in such other manner as determined by the Tax Matters Member in its reasonable discretion.
- (f) Each Member hereby agrees to indemnify and hold harmless the other Members from and against any liability (including any liability for taxes) with respect to income attributable to, or distributions or other payments to, such Member.
- Proceeding, the reduction or satisfaction of any Company Tax Liability, to enable the Company to satisfy any applicable tax reporting or compliance requirements, to make any tax election, to qualify for an exception from or reduced rate of tax or other tax benefit, to be relieved of liability for any tax, or any other action taken by the Tax Matters Member regardless of whether such requirement, tax benefit, or tax liability existed on the date such Member was admitted to the Company. The Members shall promptly provide upon the Tax Matters Member's request therefor, such information as the Tax Matters Member may reasonably request, and shall take such action as reasonably requested by the Tax Matters Member, including executing and filing returns, forms, or other statements, providing information about the Member, or any other action reasonably requested by the Tax Matters Member. If a Member fails to file or provide any such returns, forms, statements, or other information requested by the Tax Matters Member such Member will be required to indemnify the Company and the other Members for the share of any tax deficiency paid or payable by the Company that is due to such failure (as reasonably determined by the Board).
- (h) Notwithstanding any other provision of this Agreement, the obligations of a Member pursuant to this Section 7.11 (i) may be enforced by legal process or any other lawful means, including, without limitation, by offsetting any unpaid obligations against amounts otherwise distributable to the Member, (ii) will survive indefinitely with respect to any taxes withheld by the Company, or paid by the Company or the Members, that relate to the period during which such Person was actually a Member, regardless of whether such taxes are assessed, withheld or otherwise paid during such period, (iii) will survive the transfer by that Member of its interest, the withdrawal of that Member, or that Member otherwise ceasing to be a Member of the Company, (iv) will survive the dissolution, liquidation, winding up, and termination of the

Company, and (v) will remain binding on the Members for as long a period of time as is necessary to resolve with the Internal Revenue Service or any other taxing authority any and all matters regarding the taxation of the Company or the Members. If any Member withdraws or otherwise disposes of their Membership Interests, they shall keep the Company advised of their contact information until released in writing by the Company from such obligation.

- (i) References to a Member in this <u>Section 7.11</u> include those Persons who or which are (i) former Members or (ii) transferees or assignees of a Member or former Member.
- (j) The Members covenant and agree to amend this Agreement in order to reflect the effect of any applicable Treasury Regulations or other guidance promulgated with respect to a tax matters partner or partnership representative.

# ARTICLE VIII TERM; DISSOLUTION

#### Section 8.1 **Term; Termination.**

- (a) The term of the Company commenced on the date specified in the Certificate and shall continue until the Company is dissolved pursuant to this Agreement.
  - (b) This Agreement may be terminated under the following circumstances:
    - (i) By the Class A Member, upon a material breach of this Agreement by the Class B Member, which breach is not cured within ten (10) days (for any monetary breach) or sixty (60) days (for any non-monetary breach) after receipt of written notice of such breach;
    - (ii) By the Class B Member, upon a material breach of this Agreement by the Class A Member, which breach is not cured within ten (10) days (for any monetary breach) or sixty (60) days (for any non-monetary breach) after receipt of written notice of such breach;
    - (iii) By the Class A Member, upon a material breach of the Management Services Agreement by Management Co., which breach is not cured within the cure period provided under the Management Services Agreement;
    - (iv) By the Class B Member, upon a material breach of the Theater Services Agreement by the Company, which breach is not cured within the cure period provided under the Theater Services Agreement;
      - (v) By the mutual agreement of the Class A Member and the Class B Member; or

(vi) By either the Class A Member or the Class B Member in the event of a winding up, liquidation or dissolution of the other Member or the filing of any petition under any bankruptcy or other insolvency law by or against the other Member or assignment for the benefit of creditors of the other Member.

Section 8.2 **Liquidation of Company**. Upon termination of this Agreement pursuant to <u>Section 8.1</u>, the Company's business shall be liquidated in an orderly manner. Subject to <u>Section 5.3(b)</u>, the Board may act as the liquidator and wind up the affairs pursuant to this Agreement. Upon the election of the Board, the Board may approve one or more liquidators to act as the liquidator in carrying out such liquidation, subject to <u>Section 5.3(a)</u>. In performing its duties, the liquidator is authorized to sell, distribute, exchange or otherwise dispose of the assets of the Company in any reasonable manner that the liquidator shall determine to be in the best interest of the Members, subject to <u>Section 5.3(a)</u>.

Section 8.3 **Distribution Upon Dissolution of the Company.** Upon dissolution of the Company, subject to Section 5.3(a), the liquidator winding up the affairs of the Company shall determine in its discretion which assets of the Company shall be sold and which assets of the Company shall be retained for distribution in kind to the Members. The value of assets to be distributed in kind shall be reasonably determined by the liquidator in good faith. Subject to the Business Code, after all liabilities of the Company have been satisfied or duly provided for, the remaining assets of the Company shall be distributed to the Members in accordance with the Members' respective positive Capital Account balances after giving effect to all contributions, distributions and allocations for all periods.

# ARTICLE IX REPRESENTATION, WARRANTIES AND COVENANTS

- Section 9.1 **Representations, Warranties and Covenants of the Members**. Each Member hereby represents, warrants and covenants to the Company and the other Member that:
- (a) it is duly organized, validly existing and in good standing under the laws of its state of organization, and is registered as a foreign corporation in each jurisdiction in which it conducts business;
- (b) it has full power and authority to execute and deliver the Operating Documents to which it is a party and to consummate the transactions contemplated by such documents, the execution, delivery and performance of the Operating Documents by such Member has been approved by all necessary action of such Member and no other proceedings on the part of such Member are necessary to approve and authorize the execution and delivery of the Operating Documents and the consummation of the transactions contemplated hereby;
- (c) it is aware that the Membership Units have not been registered under any federal or state securities laws and acknowledges that it has acquired its Membership Units for its own account for investment purposes only and not with a view to the distribution or resale thereof, in whole or in part, and agrees that it will not Transfer, or offer to Transfer, all or any portion of its Membership Units in any manner that would violate or cause the Company to violate any federal or state securities laws;

- (d) each Operating Document to which it is a party has been duly executed and delivered by such Member and, assuming due authorization, execution and delivery hereof by the other parties thereto, constitutes a valid and binding agreement of such Member enforceable against it in accordance with its terms;
- (e) other than any reports required to be filed by Affiliates of the Class A Member under applicable securities laws, it is not required to submit any notice, report or other filing with any governmental entity in connection with the execution or delivery of the Operating Documents or the consummation of the transactions contemplated thereby, and no waiver, consent, authorization or approval of any governmental entity is required to be obtained by such Member in connection with its execution and delivery of the Operating Documents or the transactions contemplated thereby;
- (f) it is aware of the income tax consequences of this Agreement and the transactions contemplated hereby, agrees to be bound by the provisions of this Agreement in reporting its share of Company Profit and Loss for income tax purposes, and is aware that there may not be any Distributable Cash available for distribution;
- (g) it shall comply in all material respects with all applicable laws, rules and regulations of any governmental agency, entity or authority in connection with the execution, delivery and performance of the Operating Documents and the transactions contemplated thereby; and
- (h) it will abide by the Company's conflict of interest policy as adopted by the Board, and as may be amended from time to time.

# ARTICLE X ADDITIONAL AGREEMENTS

# Section 10.1 **Drag-Along Sale.**

(a) Required Sale of Membership Units. If at any time (i) the Class A Member approves a sale to a third party that is not an Affiliate of the Class A Member of its Membership Units and such third party proposes to acquire the Company in a Company Sale Transaction or (ii) the Class A Parent approves a Class A Change in Control to a third party that is not an Affiliate of the Class A Member which would include a Company Sale Transaction (in either case, a "Drag-Along Sale"), then, without any further action or approval by the Managers, all other Members (the "Dragged Members") shall consent to and raise no objections against the Drag-Along Sale or the process by which the Drag-Along Sale is undertaken and the Dragged Members, if requested by the Class A Member, shall sell (or otherwise Transfer) each such Dragged Member's respective Membership Units (or any portion thereof if requested), on the same terms and conditions approved by and applicable to the Class A Member and as set forth in this ARTICLE X and provided that the purchase price for the Membership Units of the applicable Dragged Member is not less than the Call Purchase Price that would be applicable to the Membership Units (or portion thereof) of the applicable Dragged Member pursuant to the Class A Member Call Option. Each Dragged Member shall promptly take all actions deemed necessary or desirable (in the sole judgment of the Class A Member) in connection with, and to

facilitate the consummation of, the Drag-Along Sale, including the execution of all agreements and instruments as requested by, and on the same terms applicable to, the Class A Member. Without limiting the foregoing, (i) if the Drag-Along Sale is structured as a merger, consolidation, joint venture, sale of all or substantially all assets of the Company or similar transaction, then each Dragged Member shall vote in favor of such transaction and waive any dissenters' rights, appraisal rights or similar rights in connection with such merger, consolidation or sale and (ii) if the Drag-Along Sale is structured as a sale or exchange of Membership Units, each Dragged Member shall agree to sell or exchange its Membership Units on the terms and conditions approved by the Class A Member and upon which the Class A Member agrees to sell or exchange its Class A Membership Units. The Company shall notify the Dragged Members in writing not less than thirty (30) days prior to the proposed consummation of a Drag-Along Sale; provided, however, that the Dragged Members agree not to directly or indirectly (without the prior written consent of the Company) disclose to any other Person (other than to such Dragged Member's legal counsel in confidence, as otherwise necessary to protect such Dragged Member's rights under this Agreement or as otherwise required by law) any information related to such potential Drag-Along Sale. Notwithstanding anything in this Section 10.1(a) to the contrary, the Class A Member may not invoke the right to require a Drag-Along Sale to a purchaser engaged in a Class B Competing Business unless such Drag-Along Sale is related to a Class A Change in Control. In the event of any Drag-Along Sale, the proceeds will be distributed to the Members pro rata (in accordance with their respective Membership Interests).

- (b) <u>Exceptions</u>. <u>Notwithstanding</u> the foregoing, neither a Class A Member nor a Class B Member or Class C Member (if an Affiliate of such Class B Member or Class C Member has not managed the Facility for the prior 6 months preceding a Drag-Along Sale) will be required to comply with the provisions of <u>Section 10.1(a)</u> above in connection with any Drag-Along Sale, unless:
- (i) any representations and warranties to be made by such Member in connection with the Drag-Along Sale are limited to representations and warranties related to authority, ownership and the ability to convey title to such Membership Interest, including, but not limited to, representations and warranties that (A) the Member holds all right, title and interest in and to the Membership Interest such Member purports to hold, free and clear of all liens and encumbrances, (B) the obligations of the Member in connection with the transaction have been duly authorized, (C) the documents to be entered into by the Member have been duly executed by the Member and delivered to the acquirer and are enforceable against the Member in accordance with their respective terms; and (D) neither the execution and delivery of documents to be entered into in connection with the transaction, nor the performance of the Member's obligations thereunder, will cause a breach or violation of the terms of any agreement, law or judgment, order or decree of any court or governmental agency;
- (ii) the Member shall not be liable for the inaccuracy of any representation or warranty made by any other Person in connection with the Drag-Along Sale;
- (iii) the liability for indemnification, if any, of such Member in the Drag-Along Sale and for the inaccuracy of any representations and warranties is several and not joint with any other Person and is pro rata in proportion to, and does not exceed, the amount of consideration paid to such Member in connection with such Drag-Along Sale; and

(iv) if any Member is given an option as to the form and amount of consideration to be received as a result of the Drag-Along Sale, (i) all Members will receive the same economic considerations and (ii) if the consideration consists of cash and non-cash components, the Class B Member (or Class C Member as applicable) shall receive cash.

# Section 10.2 Class A Member Call Option.

- (a) <u>Right to Purchase Membership Units</u>. At any time after the occurrence of a Call Option Trigger, the Class A Member shall have the right (the "*Class A Member Call Option*") to elect to purchase all Membership Units then outstanding (the "*Option Units*") for a purchase price equal to the Call Purchase Price; provided that if the Call Option Trigger is a Class B Default, the Call Purchase Price shall be reduced by fifty percent (50%).
- (b) <u>Procedure.</u> If the Class A Member wishes to exercise its right to purchase Option Units pursuant to this <u>Section 10.2</u>, the Class A Member shall deliver to the holders of the Option Units any time after the occurrence of a Call Option Trigger, a written notice (the "*Call Option Notice*") specifying the applicable Call Option Trigger, the applicable Call Purchase Price and the closing date for the purchase of the Option Units (the "*Call Right Closing Date*"), which such Call Right Closing Date shall be no earlier than 30 days from the date of the Call Option Notice. At the closing of any purchase consummated pursuant to this <u>Section 10.2</u>, each holder of Option Units shall represent and warrant to the Class A Member that (i) such holder has full right, title and interest in and to its Option Units, (ii) such holder has all the necessary power and authority and has taken all necessary action to sell such Option Units as contemplated by this <u>Section 10.2</u>, and (iii) the Option Units are free and clear of any and all Liens. Each holder of Option Units shall take all actions as may be reasonably necessary to consummate the sale contemplated by this <u>Section 10.2</u>, including without limitation, entering into agreements and delivering certificates, instruments and consents as may be deemed necessary or appropriate.
- Section 10.3 **Assignment of Option Units**. Each holder of Option Units and any subsequent holder of the Option Units shall execute and deliver to the Class A Member a blank Assignment of Membership Units in the form of Exhibit F to enable the Class A Member to exercise its rights under Section 10.1 and Section 10.2 without requiring additional signatures on the part of any holder of Option Units or any subsequent holder of the Option Units.
- Section 10.4 **Class B Default**. In the event of a Class B Default (defined below), all outstanding Class B Membership Units shall automatically convert into Class C Membership Units as provided in <u>Section 10.5</u>. A "Class B Default" shall have occurred in any of the following instances:
  - (a) A Capital Contribution Default by the Class B Member.
- (b) Any Transfer by the Class B Member (including, without limitation, an Involuntary Transfer) made in violation of this Agreement.

- (c) A material breach by the Class B Member under this Agreement that is not cured within ten (10) days (for any monetary breach) or sixty (60) days (for any non-monetary breach that the Class B Member makes reasonable efforts to cure during such cure period) after receipt of written notice from the Class A Member of such breach.
- (d) A material breach by Management Co. under the Management Services Agreement that is not cured within the cure period provided in the Management Services Agreement.
- (e) Management Co. terminates the Management Agreement without cause under <u>Section 10.2(b)</u> of the Management Agreement and another company owned by the Class B Member that has operating capabilities has not been contracted prior to the end of the applicable transition period; provided however, that if Management Co. terminates the Management Agreement as a result of termination of the Theater Services Agreement, this shall not be a Class B Default.
- (f) A Class B Change in Control that is not approved by the Class A Member (which such approval may be withheld for any reason).
- (g) The Class B Member (or its direct or indirect parent), Management Co. or Management Parent winds up, liquidates or dissolves or commences any proceeding to take such action.
- (h) A petition under bankruptcy or other insolvency laws is filed against the Class B Member (or its direct or indirect parent), Management Co. or Management Parent or the Class B Member (or its direct or indirect parent), Management Co. or Management Parent makes an assignment for the benefit of its creditors.
- Section 10.5 Class B Conversion. In the event that (a) a Class B Default occurs, (b) a Company Mitchell Trigger Event or Cinemark Mitchell Trigger Event occurs or (c) the Class B Membership Units are Transferred to any Person other than a Permitted Transferee, then unless otherwise expressly waived in writing by the Class A Member in its sole discretion within five days after the Class A Member receives notice of a Class B Default, a Company Mitchell Trigger Event, a Cinemark Mitchell Trigger Event or Transfer of Class B Membership Interests, all outstanding Class B Membership Units shall automatically convert into the same number of Class C Membership Units without further action by the Class B Member. Upon the conversion of the Class B Membership Units into Class C Membership Units, (i) all voting or consent rights of the Class B Member shall cease, (ii) the Class A Member shall have the right to elect the entire Board, (iii) the Class A Member (or Class A Managers, as applicable) shall have the right to make all decisions enumerated in Section 5.3, and (iv) the holder of the Class C Membership Units shall have such rights and obligations with respect to the Class C Membership Units converted to Class C Membership Units shall be the Capital Account attributable to such Class C Membership Units.

#### Section 10.6 Sale of Class C Membership Units.

- (a) <u>Class B or Class C Offer Election</u>. If the Class A Member has not delivered a Call Option Notice within nine (9) months after the occurrence of a Company Mitchell Trigger Event or a Cinemark Mitchell Trigger Event, the Class B Member or Class C Member, as applicable (the "CB Member") will have the right for ninety (90) days after the expiration of such ninety (90) day period (the "Offer Election Period") to offer to sell all Class B Membership Units, or Class C Membership Units, as applicable (the "CB Units"), to the Class A Member for the CB Purchase Price (a "CB Offer"). The CB Offer shall be evidenced by a written notice (the "CB Offer Notice") to be delivered to the Class A Member during the Offer Election Period and setting forth the CB Proposed Valuation and CB Purchase Price as determined by the CB Member. The CB Offer Notice will constitute the CB Member's offer to Transfer the CB Units to the Class A Member, which offer will be irrevocable for a period of 30 days (the "CB Notice Period").
- (b) <u>Purchase of CB Units</u>. The Class A Member may accept the CB Offer by delivering a written notice (a "*CB Acceptance Notice*") to the CB Member stating that it accepts the CB Offer to sell the CB Units for the CB Purchase Price specified in the CB Offer Notice.
  - (i) By delivering the CB Offer Notice, the CB Member will be deemed to represent and warrant to the Company and the Class A Member that: (1) the CB Member has full right, title and interest in and to the CB Units; (2) the CB Member has all necessary power and authority and has taken all necessary action to sell such CB Units as contemplated by this Section 10.6; and (3) the CB Units will be Transferred free and clear of any and all Liens other than restrictions on transfer arising under state or federal securities laws or under this Agreement.
  - (ii) If the Class A Member delivers a CB Acceptance Notice on or before the expiration of the CB Notice Period, the CB Member and the Class A Member will take all actions as may be reasonably necessary to consummate any sale to the Class A Member contemplated by this <u>Section 10.6</u>, including entering into agreements and delivering certificates and instruments and consents as may be deemed necessary or appropriate by the Company.
  - (iii) The closing date for the purchase of the CB Units shall be specified by the Class A Member in the CB Acceptance Notice and shall be a date not more than thirty (30) days after the date of the CB Acceptance Notice. At the closing of any sale to the Class A Member pursuant to this Section 10.6, the CB Member will deliver to the Class A Member an Assignment of the CB Units to be sold, accompanied by any other evidence of transfer and all necessary transfer taxes paid and stamps affixed, if necessary, against receipt of the purchase price therefor from the Class A Member by wire transfer of immediately available funds into an account specified by the CB Member (or such other method of payment as may be agreed upon by the CB Member).

#### Section 10.7 Sale of the Company.

- (a) <u>Sale Election</u>. If (i) the Class A Member has not delivered a Call Option Notice within nine (9) months after the occurrence of a Company Mitchell Trigger Event or a Cinemark Mitchell Trigger Event, (ii) the CB Member has delivered a CB Offer Notice during the CB Notice Period and (iii) the Class A Member has not accepted the CB Offer pursuant to Section 10.6, then the CB Member will have the right (the "Sale Right") during the six (6) month period following the expiration of the CB Notice Period (the "Sale Election Period") to elect to cause the Company to solicit offers from prospective third party purchasers to buy the Company (each, a "Purchase Offer"). If the CB Member elects to exercise the Sale Right, the CB Member will notify the Class A Member, in writing of any such election (a "Sale Election Notice"). Upon receipt of a Sale Election Notice from the CB Member, the Class A Member will proceed to cause the Board to commence the process to solicit a Purchase Offer pursuant to Section 10.7(b) (the "Sale Process").
- (b) <u>Sale Process</u>. Within 10 days after receipt of a Sale Election Notice to commence a Sale Process, the CB Member and the Class A Member shall jointly retain (or cause the Company to retain) either Peter J. Solomon Company or Barclays Bank as the investment banker to conduct the Sale Process. If the CB Member and the Class A Member are unable to agree on which of the two investment banks to retain, the investment bank offering the most favorable economic terms to the Company shall be selected. The CB Member and the Class A Member shall cause the Board to prepare an information package on the Company that includes (i) a form purchase agreement reasonably approved by the Board (to be executed by any such prospective third party purchaser), and (ii) the other material terms and conditions on which the Class A Member and the CB Member are prepared to sell the Company.
- (c) <u>Purchase Offer</u>. Unless otherwise mutually agreed by the Class A Member and the CB Member, any Purchase Offer must be a Qualified Purchase Offer. If the Company obtains a Qualified Purchase Offer within six (6) months after delivery of a Sale Election Notice to commence a Sale Process, and the CB Member determines that the Company should accept such Qualified Purchase Offer, the CB Member will provide written notice (a "*Company Sale Notice*") to the Class A Member of its desire to accept such Qualified Purchase Offer. Upon receipt of the Company Sale Notice, the Board shall consummate a sale of the entire Company on the terms and subject to the conditions set forth in the applicable Qualified Purchase Offer.

# ARTICLE XI CONFLICTS, EXCULPATION, INDEMNIFICATION

Section 11.1 **Conflicts**. The Company and the Class B Member acknowledge and agree that except as otherwise expressly provided herein, nothing herein shall be deemed to restrict Affiliates of the Class A Member from owning, acquiring and operating entertainment facilities similar to the Facilities, including entertainment facilities with movie theaters, bowling, games, rides, restaurants, and bars and in no event shall the Company or the Class B Member be deemed to have any interest in any entertainment facility owned, acquired, or operated by Affiliates of the Class A Member. The Company and the Class A Member acknowledge and agree that except as otherwise expressly provided herein, nothing herein shall be deemed to restrict the Affiliates of the Class B Member from owning, acquiring and operating any

entertainment facilities that do not contain a movie theater. For a period of 10 years from the Effective Date, except for facilities in existence as of the Effective Date, (i) neither the Class A Member nor the Class B Member nor their respective Affiliates shall own, acquire or operate entertainment facilities similar to the Facilities within a radius of eight (8) miles from any Facility developed and owned by the Company, (ii) neither the Class A member nor any of its Affiliates shall own, acquire or operate a movie theatre within three (3) miles from any Facility owned by the Company or (iii) neither the Class B Member or its Affiliates shall own, acquire or operate a family entertainment center incorporating bowling, games, restaurant, bar and rides within eight (8) miles from any Facility owned by the Company.

### Section 11.2 Limitation of Liability; Exculpation.

- (a) The debts, obligations and liabilities of the Company, whether arising in contract, tort or otherwise, shall be solely the debts, obligations and liabilities of the Company, and no Member or Manager shall be obligated personally for any such debt, obligation or liability of the Company solely by reason of being a Member or Manager.
- (b) No Member, Manager or Officer of the Company shall be liable to the Company or to any Member for any action (or omission to act) taken with respect to the Company so long as such Member, Manager or Officer: (i) acted in good faith and in a manner such Person reasonably believed to be in the best interests of the Company; (ii) was neither grossly negligent nor engaged in fraud or willful malfeasance; (iii) did not breach this Agreement in any material respect; and (iv) did not knowingly violate any material law. A Member, Manager or an Officer shall be fully protected and justified with respect to any action or omission taken or suffered by such Person in good faith if such action or omission was taken or suffered in reliance upon and in accordance with the opinion or advice as to matters of law, of legal counsel or, as to matters of accounting, of accountants selected with reasonable care so long as such Member, Manager or Officer: (i) acted in good faith and in a manner such Person reasonably believed to be in the best interests of the Company; (ii) was neither grossly negligent nor engaged in fraud or willful malfeasance; (iii) did not breach this Agreement in any material respect; and (iv) did not knowingly violate any material law. A failure to observe any formalities or requirements of this Agreement, the Certificate or the Business Code shall not be grounds for imposing personal liability on any Members, Manager or Officer for liabilities of the Company.

# Section 11.3 Indemnification.

(a) The Company shall indemnify any Person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that such Person is or was a Member, Manager, Affiliate of a Member or Officer or agent of the Company, or is or was serving at the request of the Company as a manager, director, officer or agent of another limited liability company, corporation, partnership, joint venture, trust or other entity or enterprise (each, an "Indemnitee"), against any loss, damage, liability or expense (including attorneys' fees, costs of investigation and amount paid in settlement) incurred by or imposed upon the Indemnitee in connection with any action, suit or proceeding, unless it shall have been finally adjudicated that the Indemnitee: (i) did not act in good faith and in a manner that such Indemnitee reasonably believed to be in the best interest of the Company; (ii) was either grossly

negligent or engaged in willful malfeasance; (iii) breached this Agreement in any material respect; or (iv) knowingly violated any material law. Notwithstanding the foregoing, no indemnification shall be payable hereunder to any Indemnitee in respect of any action in which such Indemnitee is a plaintiff, other than an action for indemnification under this <u>Section 11.3</u>. The Board shall have the discretion to extend the protection of this Section 11.3 to employees of the Company or of any Subsidiary who are not officers.

- (b) The Company shall pay the expenses incurred by an Indemnitee in defending any action, suit or proceeding, or in opposing any claim arising in connection with any potential or threatened action, suit or proceeding, in each case for which indemnification may be sought pursuant to this <u>ARTICLE XI</u>, in advance of the final disposition thereof, upon receipt of a written undertaking by such Indemnitee to repay such payment if it shall be determined by a court of final jurisdiction that such Indemnitee is not entitled to indemnification therefor as provided herein.
- (c) The rights to indemnification and advancement of expenses conferred in this <u>ARTICLE XI</u> shall not be exclusive of any other right which any Indemnitee may have or hereafter acquire under any law, statute, rule, regulation, charter document, bylaw, contract or agreement and shall inure to the benefit of the executors, administrators, personal representatives, successors and permitted assigns of each such Indemnitee.
- (d) Recourse by an Indemnite for indemnity under this <u>ARTICLE XI</u> shall be only against the Company as an entity, and no Member shall by reason of being a Member be liable for the Company's obligations under this <u>ARTICLE XI</u>.
- (e) The Company may purchase and maintain insurance on behalf of any Person who is or was a Member, Manager, Affiliate of a Member or Manager, officer, agent or employee of the Company, or is or was serving at the request of the Company as a manager, director, officer, employee or agent of another Person, against any liability asserted against such Person and incurred by such Person in any such capacity, or arising out of such Person's status as such, whether or not such Person would be entitled to indemnity against such liability under the provisions of this ARTICLE XI.
- (f) Rights and benefits conferred on an Indemnitee under this <u>ARTICLE XI</u> shall be considered a contract right and shall not be retroactively abrogated or restricted without the written consent of the Indemnitee affected by the proposed abrogation or restriction.

# ARTICLE XII MISCELLANEOUS PROVISIONS

Section 12.1 **Notices.** All notices, requests and other communications to any party hereunder shall be in writing (including a telecopy or similar writing) and shall be given to such party at its address, electronic mail address or telecopy number set forth on Exhibit A hereto or such other address, electronic mail address or telecopy number as such party may hereafter specify, in writing, for such purpose by notice to the Board (if such party is a Manager) or to all the Members (if such party is a Member). Each such notice, request or other communication shall be effective (i) if given by telecopy, when such telecopy is transmitted to the telecopy

number specified pursuant to this <u>Section 12.1</u>; (ii) if given by electronic mail, when such electronic mail is transmitted to the electronic mail address specified pursuant to this <u>Section 12.1</u>; or (iii) if given by any other means, when delivered at the address specified pursuant to this <u>Section 12.1</u>.

Section 12.2 **Amendment**. This Agreement may not be altered, amended or repealed without the approval of (i) the Class A Member and (ii) the holders of a majority of the combined Class B Membership Units and Class C Membership Units voting together as a single class. Upon any change in the Membership Units of the Members resulting from (i) the assignment or transfer of any Membership Units or (ii) the issuance of any new Membership Units, the Board shall be authorized to and shall amend Exhibit A to reflect any change in the Membership Units of any Member. Notwithstanding any provision of this Agreement to the contrary, without the approval of each affected Member, no amendment to this Agreement shall be adopted that would (a) increase the liability of a Member beyond the liability of such Member expressly set forth in this Agreement or otherwise adversely modify or affect the limited liability of such Member, (b) disproportionately decrease the interest in the Company of any Member (other than as a result of the issuance of additional Membership Units or as otherwise provided in this Agreement), (c) adversely change the method of distributions or allocations made under ARTICLE VII hereof to any Member, or (d) reduce the Capital Account of any Member other than as contemplated in this Agreement.

Section 12.3 **Jurisdiction; Waiver of Jury Trial**. To the fullest extent permitted by applicable law, the Managers and each Member hereby agree that any claim, action or proceeding by any Member seeking any relief whatsoever against any Indemnified Party based on, arising out of or in connection with this Agreement or the Company's affairs shall be brought only in State or Federal courts located in Dallas or Collin County, Texas, and not in any other State or Federal court in the United States of America or any court in any other country. The Managers and each Member acknowledge that, in the event of any breach of this provision, such Indemnified Parties have no adequate remedy at law and shall be entitled to injunctive relief to enforce the terms of this Agreement. **EACH MEMBER HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.** 

Section 12.4 **Headings**. The headings used in this Agreement are for convenience only and do not constitute matter to be construed in the interpretation of this Agreement.

Section 12.5 **Governing Law**; **Severability**. This Agreement shall be governed by and construed in accordance with the laws of the State of Texas. In particular, it shall be construed to the maximum extent possible to comply with all of the terms and conditions of the Business Code. If, nevertheless, it shall be determined by a court of competent jurisdiction that any provision or wording of this Agreement shall be invalid or unenforceable under the Business Code or other applicable law, such invalidity or unenforceability shall not invalidate the entire Agreement. In that case, this Agreement shall be construed so as to limit any term or provision so as to make it enforceable or valid within the requirements of applicable law, and, in the event such term or provision cannot be so limited, this Agreement shall be construed to omit such invalid or unenforceable provisions.

Section 12.6 **Successors; Counterparts.** This Agreement (i) shall be binding as to the executors, administrators, estates, heirs and legal successors of the Members and (ii) may be executed in several counterparts with the same effect as if the parties executing the several counterparts had all executed one counterpart. Signatures to this Agreement which are distributed to the Members via facsimile or other electronic means (including .pdf) shall have the same effect as if distributed in original form to all Members.

Section 12.7 **Entire Agreement**. This Agreement, including all exhibits and schedules hereto, constitutes the entire agreement of the Members hereto with respect to the subject matter hereof and supersedes any prior or contemporaneous oral and written understandings or agreements. All exhibits and schedules that are annexed or attached to this Agreement are expressly made a part of this Agreement as fully as though completely set forth herein and all references to this Agreement herein or in any of such exhibits and schedules shall be deemed to refer to, and include, all such exhibits and schedules.

Section 12.8 **No Waiver**. No course of dealing between the Company and any Members, and no delay by the Company in exercising any right, power or remedy shall operate as a waiver or otherwise prejudice the exercise by the Company of that right, power or remedy against that or any other Member.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the undersigned has hereto set their hand as of the day and year first above written.

# CLASS A MEMBER: CNMK TEXAS PROPERTIES, LLC

By: /s/ Michael Cavalier

Name: Michael Cavalier

Title: EVP – General Counsel

# CLASS B MEMBER: AWSR INVESTMENTS, LLC

By: /s/ Gary Witherspoon
Name: Gary Witherspoon

Title: President

Signature Page to
Limited Liability Company Agreement of FE Concepts, LLC

IN WITNESS WHEREOF, the undersigned have hereto set their hand as of the day and year first above written

<u>MANAGERS</u>	
/s/ W. Mark Moore	
W. MARK MOORE	
/s/ Lee Roy Mitchell	
LEE ROY MITCHELL	
/s/ Mark Zoradi	
MARK ZORADI	
/s/ Sean Gamble	
SEAN GAMBLE	

Signature Page to Limited Liability Company Agreement of FE Concepts, LLC

# **EXHIBIT A**

# **MEMBERS**

# **MEMBERSHIP INTEREST**

# **CLASS A MEMBER**

CNMK Texas Properties, LLC 50% 3900 Dallas Parkway, Suite 500

Plano, Texas 75093

Attn: Michael Cavalier, General Counsel E-mail: MCavalier@cinemark.com

# **CLASS B MEMBER**

AWSR Investments, LLC 50%

12400 Coit Road, Suite 800 Dallas, Texas 75251 Attn: Gary Witherspoon E-mail: gw@cbcap.com

Dated: April 20, 2018

# **EXHIBIT B**

# INITIAL APPOINTMENTS

Board	
Class B Manager; Board Chair:	Lee Roy Mitchell
Class A Manager; Vice Chair:	Mark Zoradi
Class A Manager:	Sean Gamble
Class B Manager:	Mark Moore

#### EXHIBIT C

#### FORM OF MANAGEMENT SERVICES AGREEMENT

THIS MANAGEMENT SERVICES AGREEMENT (this "Agreement") is dated and made effective as of April 20, 2018 (the "Effective Date"), by and between FE Concepts, LLC, a Texas limited liability company (the "Company") and Cinema Operations, L.L.C., a Texas limited liability company (together with any of its subsidiaries from time to time providing Services hereunder, the "Manager"). The Manager and the Company are sometimes referred to in this Agreement, collectively, as the "Parties" and each individually as a "Party." Defined terms used in this Agreement but not otherwise defined herein have the meanings ascribed to them on Schedule I hereto.

#### **RECITALS**

- A. The Manager is an affiliate of AWSR Investments, LLC, a Texas limited liability company ("CB Entity").
- B. Cinemark USA, Inc., a Texas corporation ("Cinemark") is an affiliate of CNMK Texas Properties, LLC, a Texas limited liability company ("CNK Entity").
- C. CNK Entity and CB Entity are parties to that certain Limited Liability Company Agreement, dated as of April 20, 2018 (the "*LLC Agreement*"), pursuant to which CB Entity and CNK Entity have formed the Company to own and operate a certain family entertainment facility or facilities branded *Strike* + *Reel* incorporating games, rides, bowling, restaurants, bars and motion picture theater auditoriums (the "*Facilities*").
- D. The Manager has or will use commercially reasonable efforts to obtain the qualified, experienced and necessary personnel to manage each Facility and the Manager also has or will use commercially reasonable efforts to obtain the qualified personnel necessary to provide accounting and administrative services in connection with the operation of a Facility.
- E. The LLC Agreement provides in part that the Company shall engage the Manager to provide the Services pursuant to this Agreement.
- F. The Company desires to retain the Manager as the exclusive provider of the Services with respect to any Facility and the Manager is willing to provide such services, upon the terms and subject to the conditions set forth in this Agreement.
- **NOW**, **THEREFORE**, in consideration of the mutual promises and covenants contained in this Agreement and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties, intending to be legally bound, agree as follows:

# ARTICLE I CONSTRUCTION AND DEVELOPMENT SERVICES

1.1 Provision of Construction and Development Services. Subject to Article III below, the Company hereby engages the Manager to provide the construction and development services described in Schedule II hereto (collectively, the "Development Services") in connection with the acquisition, construction and development of a Facility. Subject to Article III below, the

Manager is hereby vested with full authority to take all such actions, do all such things and execute and enter into on behalf of the Company all such contracts, agreements, leases, licenses, instruments, commitments, undertakings and understandings (collectively, "Contracts") relating to such activities listed on Schedule II except as specified in Section 3.1.

- 1.2 Standards for Provision of Development Services.
- (a) The Manager shall devote such time and personnel as the Manager deems appropriate to provide the Development Services. The Manager shall not be required to spend its full time and attention to provide the Development Services.
- (b) The Company will provide such cooperation as is needed from time to time in connection with the provision of the Development Services hereunder. In connection with the provision of the Development Services hereunder, the Manager shall obtain and maintain at the Company's expense, for itself or on behalf of the Company, as the case may be, all licenses, permits and authorizations from any Governmental Authorities which are necessary for the construction and development of a Facility. The Company shall cooperate with the Manager in obtaining all consents, authorizations, approvals, permits and licenses necessary for the construction and development of a Facility. The Manager shall use commercially reasonable efforts to be at all times qualified and licensed to do business and be in good standing in all jurisdictions where such qualification or license is required to enable the Manager to provide the Development Services as contemplated by the terms of this Agreement.
- (c) The Manager shall conduct all activities and duties hereunder in good faith and shall use its commercially reasonable efforts to obtain terms for the supply of goods and services to a Facility and the Business comparable with the terms that the Manager and its subsidiaries and Affiliates are able to obtain for the Manager Related Facilities, provided, however, that the Manager is not required to purchase such goods and services on the same terms as Manager purchases similar goods and services for Manager Related Facilities. In connection with the construction and development of a Facility, the Manager may purchase necessary goods, supplies and services from or through the Manager or any of its Affiliates so long as such prices are comparable with the prices and terms of goods, supplies and services of like quality available from non-affiliated third parties in an arm's length transaction.
- 1.3 Theater Services. The Manager acknowledges that the Company has entered into the Theater Services Agreement with Cinemark pursuant to which Cinemark shall provide the Theater Services, which includes certain expertise regarding the construction and design of the motion picture theater auditoriums and the selection of certain technologies related thereto (the "Theater Construction Expertise"). The Theater Construction Expertise shall be provided exclusively by Cinemark and shall not be construed to constitute Development Services provided by the Manager. The Manager shall coordinate the provision of Development Services under this Agreement with the provision of the Theater Construction Expertise by Cinemark, including the implementation of any recommendations or plans resulting therefrom.

### ARTICLE II MANAGEMENT SERVICES

2.1 Provision of Management Services. Subject to Article III below, the Company hereby engages the Manager to manage a Facility and the business conducted at a Facility (the "Business") and to provide the Management Services in accordance with this Agreement, and the

Manager hereby agrees to manage a Facility and the Business and to provide the Management Services in accordance with this Agreement. Subject to Article III below, the Manager is hereby vested with full authority and responsibility for the day-to-day management of a Facility and the Business and is empowered to take all such actions, do all such things and execute and enter into all such Contracts in this regard, on behalf of and in the name of the Company, as the Manager may, in its reasonable discretion, deem necessary or desirable for such purpose; subject to the provisions of Article III. For the avoidance of doubt, it is agreed that the Manager shall have the responsibility and authority to make any and all decisions with respect to the day-to-day, ordinary course of business operations of a Facility and the Business, subject to the provisions of Article III.

#### 2.2 Standards for Provision of Management Services; Cooperation.

- (a) The Manager shall devote such time and personnel as the Manager deems appropriate to provide the Management Services that are comparable to those employed by the Manager in connection with the operation of the Manager Related Facilities. The Manager shall not be required to spend its full time and attention to provide the Management Services.
- (b) The Company will provide such cooperation as is needed from time to time in connection with the provision of the Management Services hereunder. In connection with the provision of the Management Services hereunder, the Manager shall use commercially reasonable efforts to obtain and maintain at the Company's expense, for itself or on behalf of the Company, as the case may be, all licenses, permits and authorizations from any Governmental Authorities which are necessary for the operation of the Business in the manner contemplated by this Agreement, including, but not limited to, licenses to serve food and alcoholic beverages. The Company shall cooperate with the Manager in obtaining all consents, authorizations, approvals, permits and licenses necessary to operate the Business. The Manager shall use commercially reasonable efforts to be qualified and licensed to do business and be in good standing in all jurisdictions where such qualification or license is required to provide Management Services as contemplated by the terms of this Agreement.
- (c) The Manager shall conduct all activities and duties hereunder in good faith and shall use commercially reasonable efforts to obtain terms for the supply of goods and services to and by the Facilities and the Business comparable with the terms that the Manager and its subsidiaries and Affiliates are able to obtain for the Manager Related Facilities. In the management of the Facilities, the Manager may purchase necessary goods, supplies and services from or through the Manager or any of its Affiliates so long as such prices are comparable with the prices and terms of goods, supplies and services of like quality available from non-affiliated third parties in an arm's length transaction.
- (d) As part of the Management Services, the Manager shall provide "point of sale" systems necessary to effectuate the retail transactions incident to the Business for the Facilities that are the same as the point of sale software systems used by the Manager in the operations of the Manager Related Facilities.

- (e) The Manager shall use commercially reasonable efforts to obtain all software, permits, licenses, sublicenses, authorizations, intellectual property and other intangibles and rights necessary to operate the Facilities and perform under this Agreement, including any third party software, applications and/or licenses. The Manager shall use commercially reasonable efforts to obtain any necessary consents and/or provide documentation to evidence the foregoing upon the Company's request.
- 2.3 Compliance with Leases. In providing the Management Services, to the extent a Facility is subject to a Lease (as defined below), the Manager shall use commercially reasonable efforts to comply in all material respects with, and not violate in any material respect, the terms of any real property lease associated with the Facilities (each such lease, a "Lease" and collectively, the "Leases").
- 2 . 4 Compliance with Laws. The Manager shall use commercially reasonable efforts to cause the Facilities to comply in all material respects with any Law affecting the Facilities and issued by any Governmental Authority having jurisdiction thereof. Further, the Manager shall not unreasonably refuse or delay compliance with any specific instructions that are given by the Company and that are reasonably necessary to cause the Facilities to comply with such Laws. The Company shall cooperate with the Manager in complying with such Laws, including paying all costs of such compliance.
- 2.5 Theater Services. The Manager acknowledges that the Company has entered into the Theater Services Agreement with Cinemark pursuant to which Cinemark shall provide the Theater Services. The Theater Services shall be provided exclusively by Cinemark and shall not be construed to constitute Management Services provided by the Manager. The Manager shall consult with Cinemark to receive Cinemark's recommendations with respect to the Theater Services to be provided by Cinemark under the Theater Services Agreement. The Manager shall coordinate providing its Services under this Agreement with the Theater Services to be provided by Cinemark under the Theater Services Agreement. The Manager shall cause the Company to pay all fees and expenses payable to Cinemark under the Theater Services Agreement.
- 2.6 Internet Websites. The Manager will create, or cause to be created, an Internet website for the Company listing each Facility on such website and providing on such website the ability for customers to purchase Facility admissions, prepaid cards and movie tickets for each Facility.

### ARTICLE III LIMITATIONS ON MANAGER AUTHORITY

- 3 . 1 Actions Requiring Prior Written Consent of the Company. Notwithstanding anything to the contrary contained in this Agreement, the Manager shall not, unless specifically authorized by the Company in writing, take any of the following actions with respect to any of the Company or any of its assets or properties, including the Facilities:
- (a) incur or commit to incur on behalf of the Company any capital expenditure (or series of related capital expenditures) that are, in the aggregate, more than five percent (5%) over amounts included in an Approved Budget; <u>provided, however</u>, that nothing in this Agreement shall be construed to prohibit or prevent the Manager from making emergency repairs necessary to prevent (i) damage to or destruction of the assets, properties or premises of the Facilities or (ii) protect the safety of the Facilities employees, customers or other invitees;

- (b) subject to Section 6.3, incur or commit to incur on behalf of the Company any expenditure(s) that would result in the aggregate of expenditures in all accounting categories, (other than Excluded Expenditures), exceeding an Approved Budget by more than five percent (5%) for each Facility for such entire Fiscal Year. Notwithstanding the preceding sentence, prior written consent will not be required if an expenditure for the Business exceeds the amount set forth in the Approved Budget during an interim reporting period if the Manager reasonably determines that such expenditure will not result in the aggregate of expenditures in all accounting categories for the Business exceeding those set forth in an Approved Budget for such entire Fiscal Year by more than five percent (5%) in the aggregate for a Facility;
- (c) enter into on behalf of the Company any transaction or Contract that is neither contemplated by this Agreement nor reasonably related to the Services being provided hereunder;
- (d) enter into on behalf of the Company any Contract in excess of \$50,000 per annum that cannot be terminated by the Company without premium or penalty on less than thirty (30) days prior written notice;
- (e) incur, guarantee, grant any security interest in respect of, or otherwise become liable for, any indebtedness for borrowed money, letters of credit, notes, bonds, mortgages or similar Contracts or lease obligations that are required to be treated as capitalized leases in accordance with generally accepted accounting principles in the United States ("GAAP");
  - (f) agree to subject the assets of the Company to any other lien or encumbrance, other than Permitted Liens; or
- (g) change any policy regarding the prices charged for tickets for any of the motion pictures exhibited in a Facility without the prior approval of the Board.

In the event that the authorization of the Company is required under this <u>Section 3.1</u> in respect of any action to be taken or not to be taken hereunder or any amount to be incurred or not to be incurred hereunder, the Manager shall request such authorization in writing as provided in <u>Section 12.5</u>.

3.2 Company Approval. For the purpose of this Agreement, any consent or approval required or permitted to be given by the Company shall require the approval of the Board (including at least one Board member not affiliated with the Manager) or, the approval of an executive officer of the Company not affiliated with the Manager to the extent that specific authority to grant such consent or approval has been expressly delegated to such officer by the Board, subject to any additional approval required by the Members in accordance with the LLC Agreement. Any notice by the Manager to the Company shall be given in accordance with Section 12.5.

#### ARTICLE IV STAFFING SERVICES

- 4.1 Provision of Staffing Services. The Company hereby engages the Manager to provide the Staffing Services at a Facility in accordance with the terms of this Agreement, and the Manager hereby agrees to provide the Staffing Services at a Facility in accordance with the terms of this Agreement.
- 4.2 Standards for the Provision of Staffing Services; Cooperation. The Manager shall devote such time and personnel as the Manager deems appropriate to provide the Staffing Services. The Manager shall not be required to spend its full time and attention to provide the Staffing Services. Nothing in this Agreement shall prevent or restrict the Manager from terminating the employment or engagement of any of its employees, agency workers, consultants, contractors or other workers at any time or from taking any other actions with respect to its employees, agency workers, consultants, contractors or other workers, in its sole and absolute discretion.

#### 4.3 Manager as Employer.

- (a) The Parties hereto agree that the Manager will use commercially reasonable efforts to recruit, employ, train and supervise any and all employees necessary to enable the Manager to provide the Services hereunder, that such employees will at all times be employees of the Manager, a Manager Subsidiary or Manager Affiliate, and that the Manager will be responsible for (i) making all determinations related to such employees, including determinations with respect to wages, salaries, fringe benefits and other compensation, employment duration, the assignment of duties and the negotiation and settlement of labor disputes, (ii) making all payments in connection with the compensation of such employees, collecting and remitting any and all payroll taxes or other withholdings in connection therewith and filing any and all tax returns as are required with respect to such taxes, and (iii) procuring and maintaining adequate workers' compensation insurance as may be required by Law (it being understood that all expenses (including expenses relating to any of the foregoing) associated with the employment of Facility-level employees shall be Company Expenses).
- (b) In staffing each Facility, the Manager will use its commercially reasonable efforts to employ persons in such numbers and of such skill levels as are comparable in number and skill level to the persons the Manager employs in connection with the operation of the Manager Related Facilities.
- (c) The Manager acknowledges and agrees that, in any and all matters related to a Facility and the Business, it shall comply in all material respects with the Fair Labor Standards Act, the Occupational Safety and Health Act and any other law, rule or regulation relating to employment or human health and safety in connection with the Staffing Services. The Company acknowledges that employees at a Facility may be employed by a subsidiary or Affiliate of Manager.
- (d) The Parties hereto agree that the Company shall have no liability to any of the employees hired by the Manager, except for reimbursing the Manager for Company Expenses and otherwise as specifically set forth in this Agreement.

### ARTICLE V FINANCIAL INFORMATION, ACCESS AND CONSULTATION

- 5.1 Maintenance of Books and Records. The Manager shall keep full and materially accurate books of account and such other records in accordance with generally accepted accounting principles reflecting the Company operations of a Facility showing all revenue and expenses of the Company from a Facility during the term of this Agreement. The Manager's books and records for the Company will correspond with the Manager's fiscal year. Such Company records and accounts shall be prepared and maintained for each Facility in accordance with good business practices and in accordance with the method the Manager prepares such records and accounts for the Manager Related Facilities, and shall be sufficiently detailed to enable the statements of operations, balance sheets and statements of cash flows for the Company to be prepared from time to time for each Facility. For the avoidance of doubt, this Section 5.1 shall not require the Manager to provide any books and records of any Person other than the Company, including, but not limited to, any books and records of any Company Affiliates, Company Subsidiaries, Manager Affiliates, or Manager Subsidiaries.
- 5.2 Monthly and Quarterly Financial Reporting. The Manager shall within thirty (30) days after the end of each Fiscal Month ending after the Effective Date furnish to the Company a monthly Facility profit and loss statement of the Company in the form attached as Exhibit A (which monthly Facility profit and loss statement shall also be provided in a spreadsheet format), a trial balance reflecting all ledger account balances for the Company for each Facility, and a summary of all pro-rata Company costs and expenses allocated to a Facility by Manager for such period. Within forty-five (45) days after the end of each Fiscal Quarter ending after the Effective Date, unless such Fiscal Quarter ends on the last day of a Fiscal Year, in which case within sixty (60) days after the end of such Fiscal Quarter, furnish to the Company a consolidated unaudited balance sheet, income statement, statement of comprehensive income, statement of Member's equity and statement of cash flows for the Company, in each case in accordance with U.S. GAAP, which, for the avoidance of doubt, shall not require any footnotes. For the avoidance of doubt, this Section 5.2 shall not require the Manager to provide any financial records of any Person other than the Company, including, but not limited to, any financial records of any Company Affiliates, Company Subsidiaries, Manager Affiliates, or Manager Subsidiaries.
- 5.3 Inspection Rights. The representatives of the Company shall be entitled at all reasonable times and with reasonable prior notice to the Manager to inspect and obtain copies of all documents, records and accounts of the Company under the control of the Manager relating to a Facility and/or the Business and to discuss the financial statements and performance of a Facility with appropriate personnel of the Manager. The frequency, manner and duration of such inspections and discussions shall be conducted at times and dates mutually agreed by the Manager and the Company and without undue hindrance to the proper conduct of the operations of the Business or the activities of the Manager and its employees, agents and independent contractors. Any expenses incurred by the Parties in connection with any such inspections shall be the sole cost and expense of the Company. For the avoidance of doubt, this Section 5.3 shall only provide the Company with the right to inspect any documents, records and accounts of the Company Subsidiaries, Manager Affiliates, or Manager Subsidiaries.

- 5.4 Year-End Financials. The Manager shall cause the Company's accountants to prepare unaudited full year end financial statements of the Company for any such Fiscal Year (i.e., balance sheets, statements of income, statements of comprehensive income, statements of cash flows and statements of changes in Member's equity), which, for the avoidance of doubt, shall not require any footnotes, for presentation to the Company within sixty (60) days after the end of each Fiscal Year. The Company will also provide full year end audited financial statements (with related footnotes) of the Company within 180 days after the end of each Fiscal Year. In connection with the preparation of any such Company financial statement or any related audit or review thereof, the Manager shall make the Company books and records of a Facility available for such purposes in a timely manner. Any such financial statement preparation, audit or review shall be at the sole cost and expense of the Company. The costs and expenses associated with the full year end audited financial statements shall be billed directly to the Company. For the avoidance of doubt, this Section 5.4 shall only require the preparation of unaudited and audited financial statements for the Company, and not for any other Person, including, but not limited to any Company Affiliates, Company Subsidiaries, Manager Affiliates, or Manager Subsidiaries.
- 5.5 Notice of Claims, Proceedings and Material Damage. The Manager shall notify the Company in writing in a prompt and timely manner of receipt by the Manager of any written threat of, or, upon becoming aware thereof, the actual commencement or resolution of, any claims or Proceedings of the Company involving a Facility, or of any material damage, or material repair needs, of a Facility that the Company would be liable for.

# ARTICLE VI BUDGETS

- 6.1 Draft Annual Budgets. Prior to the Effective Date, the Manager and the Company shall agree upon an initial Annual Budget for the remainder of the Fiscal Year following the Effective Date in the form attached hereto as Exhibit B. With respect to each Fiscal Year thereafter during the term of this Agreement, the Manager shall prepare and provide to the Company, no later than the fifteenth (15th) day of December, a proposed draft budget for such Fiscal Year relating to each Facility in the form attached hereto as Exhibit B. The Manager shall consult with Cinemark when preparing the draft budget for line items relating to the motion picture theater auditoriums, including film rentals, in-theatre advertising and projector repair and maintenance.
- 6.2 Approved Annual Budgets. The Company shall consider and comment upon the proposed draft budget prepared by the Manager with respect to each Facility pursuant to Section 6.1, and the Manager shall from time to time amend and revise such draft budget in accordance with any reasonable written requests of the Company mutually agreed to by the Company and the Manager (each such budget including the initial Annual Budget referred to in Section 6.1, as so amended from time to time, an "Approved Budget"). If the Manager submits a proposed Annual Budget or an amendment to the Annual Budget for the Company's approval, and the Company does not request any changes or make any comments with respect to such Annual Budget, within thirty (30) days of submission, then such proposed Annual Budget shall be deemed the Approved Budget. Until such time as the proposed budget is approved by the Company, the last Approved Budget shall be the Approved Budget for such year except that (i) any portion of the proposed budget which is specifically approved or not in dispute shall be deemed approved and effective and (ii) during any interim period the Manager may reasonably exceed the Approved Budget for the prior Fiscal Year for (a) taxes, utility charges and other

items not within the Manager's reasonable control and (b) increases in contract services, personnel and other costs to the extent required to maintain the same level of services for the attendance volumes anticipated at a Facility provided during the previous Fiscal Year if the costs of such items has increased above the Approved Budget for the prior Fiscal Year. In the event the Parties are unable to agree on an Approved Budget for a particular Fiscal Year within ninety (90) days of the beginning of such Fiscal Year, Manager shall operate a Facility related to the disputed portion of the proposal based upon the last approved Annual Budget.

6.3 Operation in Accordance with Approved Budgets. The Parties acknowledge and agree that Facility operating results (e.g., Facility attendance, revenues, operating expense, net income and cash flows) are subject to significant variance within any budgeted Fiscal Year and that no draft budget or Approved Budget can be expected to accurately predict such variances. Consequently, the Parties understand that actual interim operating results may vary from the Approved Budget. The Manager shall have the authority to make those expenditures that have been approved in the Approved Budget and which do not exceed the Approved Budget by more than five percent (5%) in the aggregate with respect to any Facility; provided that, notwithstanding the foregoing, the Manager may reasonably exceed the Approved Budget and make additional expenditures as Company Expenses for (a) taxes, utility charges and other items not within the Manager's reasonable control, and (b) increases in contract services, personnel and other costs, including but not limited to film rentals, food and beverage costs, game and retail, and salaries and wages (the expenses listed in (a) and (b), the "Excluded Expenditures"), to the extent required to maintain the same level of services for the attendance volumes anticipated at a Facility, if the costs of such items have increased above the Approved Budget. If a condition of an emergency nature should exist which precludes consultation with and approval by the Company and requires that immediate repairs be made for the preservation and protection of the Facility or to protect the safety of the Facility employees, customers or other invitees, however, the Manager shall be authorized to take all actions and to make the necessary expenditure to repair and correct such condition, regardless of whether any provision has been made in the Approved Budget for such expenditure. Cash Expenditures made by the Manager in connection with any such emergency expenditure shall be paid as a Company Expense. Notwithstanding the foregoing, the Manager shall use commercially reasonable efforts to minimize expenditures not in the Approved Budget regardless of the Company's pre-approval of such expenditure. The Manager shall use commercially reasonable efforts to ensure that the actual costs of maintaining and operating a Facility shall not exceed the Approved Budget by more than five percent (5%) with respect to any Facility. All expenses shall be charged to the proper account in accordance with the Approved Budget. Expenses may be reasonably classified or reclassified in excess of the annual budgeted amount for an accounting category, provided that the Manager shall use commercially reasonable efforts to ensure the aggregate of expenditures in all accounting categories (other than Excluded Categories) does not exceed the Approved Budget by more than five percent (5%). If the Manager determines that any interim variances are reflective of expected operating results for the remainder of the Fiscal Year and that such variances may result in the Approved Budget for a Facility for such entire Fiscal Year exceeding those set forth in an Approved Budget for such Facility by more than five percent (5%) for such entire Fiscal Year (other than Excluded Expenditures), including any contingency with respect thereto provided for in the Approved Budget for the entire Fiscal Year, then the Manager may notify the Company and request that the Approved Budget for the Fiscal Year be amended in accordance with Section 6.2. Notwithstanding anything to the contrary in this Agreement, the Company's sole remedy for any breach by the Manager of this Article VI shall be termination of the Agreement in accordance with Section 10.2.

# ARTICLE VII MANAGEMENT COMPENSATION AND PAYMENT OF COSTS AND EXPENSES

- 7.1 Management Compensation. The Company shall pay to the Manager the compensation set forth on Exhibit E as follows:
  - (a) Payments under Tier 1 shall be paid within thirty (30) days of the opening of a new Facility.
- (b) Payments under Tier 2, if any, shall be paid annually within fifteen (15) days after delivery by the Manager to the Company of the unaudited full year end financial statements.
- (c) Payments under Tier 3 shall be paid upon the occurrence of a Trigger Event concurrently with payment to the CB Entity for such Trigger Event. For purposes of payments under Tier 3, "*Trigger Event*" shall mean (i) a Company Sale Transaction (as such term is defined in the LLC Agreement), or (ii) a transfer under a Call Option Notice, a CB Offer Notice, or a Sale Election Notice (as each such term is defined in the LLC Agreement).
- 7.2 Company Expenses. The Company shall be responsible for the payment of all costs and expenses incurred in connection with the construction, development and operation of a Facility or the operation of the Business, whether such costs or expenses are incurred directly by the Company or incurred by the Manager or its Affiliates in connection with the provision of the Services hereunder, including, without limitation, (a) the Facility's pro-rata shares of any Facility-level costs and expenses incurred on behalf of a Facility as part of any arrangements entered into by the Manager on behalf of a Facility and Manager Related Facilities, (b) Cash Expenditures and (c) all expenses and liabilities relating to workers compensation insurance and workers compensation claims by Facility-level employees (all such costs and expenses for which the Company are so responsible being referred to, collectively, as the "Company Expenses"). The Manager shall have the right to utilize its corporate employees and assets in connection with the operation of a Facility as required under this Agreement. When doing so, the Manager shall be reimbursed for all non-Facility level costs and expenses incurred by the Manager or its Affiliates allocable to work done for the Company, as set forth on Exhibit F ("Manager Overhead Costs"), including the salaries, benefits, bonuses and other compensation of home-office personnel who render Services and all office and overhead expenses that are incurred other than at the Facility -level. Without limiting the generality of the foregoing, Company Expenses shall include, without limitation, (i) Cash Expenditures and all of the costs and expenses of the type described on Schedule IV hereto, (ii) any costs and expenses in connection with any third party consents, permits or licenses relating to any Facility that are related to the Services, and (iii) any capital expenditures with respect to a Facility but only to the extent any such costs have been approved in advance and in writing by the Board. To the extent that the Manager corporate employees are required to, or requested to by the Company, to perform services beyond those contemplated herein (i.e., assisting with tax audits, managing litigation, etc.), then the Company and the Manager shall meet and consult in good faith to determine a fair and reasonable expense allocation for such services which will be fully reimbursable as Company Expenses. Amounts due to the Manager under this Section 7.2 shall be paid within thirty (30) days of receipt by the Company of an invoice for such amounts.

- 7.3 Company Account; Payment of Company Expenses; Quarterly Statements.
- (a) All monies advanced by the Company to fund Company Expenses and all cash from operations of each Facility shall be deposited in separate bank accounts of the Company (the "Company Account") over which the Manager, as agent for the Company, shall have joint access and control with the Company. The Manager is hereby authorized to pay from the Company Account, on behalf of the Company, all Company Expenses and to use funds in the Company Account to reimburse the Manager for any Company Expenses previously paid by the Manager. Checks or other documents of withdrawal drawn upon the Company Account by the Manager shall be signed by representatives of the Manager as agent for the Company. The Parties hereby acknowledge and agree that the Manager will manage the finances of the Business on a combined Facility basis and shall be entitled to use cash flow generated from one Facility to fund operating expenses and losses of any other Facility.
- (b) Concurrently with the delivery of the quarterly financial statements required by Section 5.2, the Manager shall issue to the Company a statement (the "Quarterly Statement") setting forth in reasonable detail all payments made from the Company Account in respect of Company Expenses with respect to each Facility and on a combined basis during such Fiscal Quarter. In the event that the Company reasonably and in good faith questions or disputes any element of any Quarterly Statement, written notice thereof shall be provided by the Company to the Manager within thirty (30) days of the receipt of the Quarterly Statement by the Company, and the Parties shall thereafter use commercially reasonable efforts to resolve to their mutual satisfaction any such questions and disputes during the twenty (20) days following delivery of such written notice by the Company. In the event the Parties are unable to resolve any such questions or disputes, the unresolved matters may be submitted to arbitration pursuant to Section 12.14.

# ARTICLE VIII INTELLECTUAL PROPERTY

- 8.1 Reservation of Rights. Any Facility subject to this Agreement shall be operated, marketed and branded as a "Strike + Reel" Facility at all times while the Services are being provided under this Agreement with respect to such Facility. The Manager confirms and acknowledges that the Company is and shall be at all times, the exclusive owner of all rights, title and interest in and to the Company Trademarks, including the goodwill associated therewith, (ii) the Manager shall not contest or challenge the Company's rights in such Company Trademarks at any time, (iii) nothing herein shall restrict the use of the Company Trademarks by the Company and the Company Subsidiaries or licensees at any time or in any manner, (iv) the Manager shall not acquire any ownership right in the Company Trademarks by virtue of said marks' use pursuant to this Agreement, and (v) any goodwill derived from the use of the Company Trademarks in connection with a Facility shall inure to the exclusive benefit of the Company.
- 8.2 Quality Control. The Manager acknowledges and agrees that: (i) all use of the Company Trademarks under this Agreement shall be, at all times, subject to the Company's reasonable quality control and (ii) to protect the valuable goodwill associated with the Company Trademarks, the Manager shall assure that the Company Trademarks are used solely in a manner consistent with the quality of the goods and services offered under the Company Trademarks. Upon receiving notice from the Company that the quality of the goods and services offered under

the Company Trademarks does not meet the Company's acceptable quality standards, the Manager shall promptly take all remedial action to assure that goods and services offered under the Company Trademarks shall meet the quality standards that are acceptable to the Company's reasonable satisfaction.

8 . 3 No Implied Rights. Nothing in this Agreement provides or implies any right for the Manager to use the Company Trademarks except as otherwise provided for in this <u>Article VIII</u>.

# ARTICLE IX INDEMNITY

- 9.1 Indemnification by the Company. The Company shall indemnify and hold harmless the Manager and its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (and the members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons) (such indemnified Persons, the "Manager Indemnified Persons") from, against and in respect of any and all claims, Proceedings, obligations, liabilities, liens, encumbrances, losses, damages, assessments, fines, penalties, taxes, fees, costs (including costs of investigation, defense and enforcement of this Agreement, including attorneys' fees and disbursements), expenses or amounts paid in settlement (in each case, including reasonable attorneys' and experts fees and expenses) (collectively, "Losses"), in the case of clauses (b), (c), and (d) below, whether or not involving a Third Party Claim, to the extent such Losses are incurred or suffered by the Manager Indemnified Persons as a result of, arising out of or directly or indirectly relating to:
- (a) any claim or Proceeding made, initiated or threatened by any Person other than a Party (a "Third Party Claim") that relates to the provision of the Services under this Agreement, the management and operation of a Facility or the Business, or the consequences of the Manager's compliance with the terms of this Agreement or any directions provided by the Company relating to the operation of the Business, other than to the extent such Losses are caused by the bad faith, gross negligence or willful misconduct of the Manager or any of its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (or the shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons);
  - (b) all Company Expenses;
  - (c) all Manager Overhead Costs;
  - (d) any breach of this Agreement by the Company;
  - (e) any potential regulatory issues related to the affiliation of the Company and the CNK Entity; and
- (f) any potential action brought by a shareholder of Cinemark and/or the CNK Entity arising from claims related to conflict of interest.
- 9.2 Indemnification by the Manager. The Manager shall indemnify and hold harmless the Company, the Company Subsidiaries and the members, managers, officers, directors, employees and other agents, representatives and Affiliates thereof (and the members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons)

(such indemnified Persons, the "Company Indemnified Persons") from, against and in respect of any and all Losses to the extent such Losses are incurred or suffered by the Manager Indemnified Persons as a result of, arising out of or directly or indirectly relating to:

- (a) any Third Party Claim that relates to the provision of the Services under this Agreement or the management and operation of a Facility or the Business, in each case, to the extent such Losses are caused by the bad faith, gross negligence or willful misconduct of the Manager or any of its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (or the shareholders, members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons); or
- (b) any Third Party Claim brought by or on behalf of any of the Manager's employees, including Third Party Claims alleging age discrimination, sex discrimination or sexual harassment; <u>provided</u> that this clause (b) shall not be deemed to (i) provide for indemnification to the Company Indemnified Persons for any cost relating to employees of the Manager that would otherwise constitute a Company Expense (for example, compensation of Facility-level employees in the ordinary course of business or workers compensation -related expenses) or (ii) reduce or modify the obligations of the Manager or the Company pursuant to <u>Section 9.3</u>.
- (c) any Third Party Claim alleging Manager has violated the Fair Labor Standards Act, the Occupational Safety and Health Act with respect to the operations of a Facility or any other rule related to employment or human health and safety in connection with the Staffing Services and the Occupational Safety and Health Act related to the construction of a Facility only if such Occupational Safety and Health Act violation is caused by the bad faith, gross negligence or willful misconduct of the Manager.

#### 9.3 Insurance.

- (a) The Manager shall maintain all appropriate insurance coverage and fidelity bonds for a Facility, and the maintenance of such insurance coverage shall be consistent with Manager's practices for the Manager Related Facilities including the minimum coverages and amounts outlined in Exhibit C. All costs and expenses of maintaining insurance in connection with the Business and the operation of a Facility shall be Company Expenses.
- (b) The Manager shall notify the Company concurrently with notification to the Company's insurance carrier of any casualty, disaster, loss, damage or injury occurring at a Facility that is reported by the Manager to an insurance carrier, and otherwise promptly notify the Company of any actual claim or loss not occurring in the Company's ordinary course of business. The Manager shall not use a Facility, or permit the same to be used, for any purpose which will make void or voidable any of such insurance policies, and shall not keep or allow to be kept at a Facility any material, machinery, equipment, substance or other thing which may make void or voidable any of such insurance policies. The Manager shall, upon the request of the Company, assist the Company in any reasonable manner, as The Company may request, in the settlement of any claim under any of such insurance policies.
- (c) The amount of Losses for which indemnification is available under <u>Section 9.1</u> and <u>9.2</u> shall be calculated net of any amounts actually recovered by the Company Indemnified Persons or the Manager Indemnified Persons, as the case may be, under insurance policies with respect to such Losses. In the event that any indemnifying person makes any

payment under <u>Section 9.1</u> or <u>9.2</u> in respect of any Losses, such indemnifying person shall be subrogated, to the extent of such payment, to the rights of such indemnified person against any insurer or other third Person with respect to such Losses.

# ARTICLE X TERM AND TERMINATION

- 10.1 Term of Agreement. This Agreement shall apply to each Facility acquired, owned or operated by the Company and shall commence on the Effective Date and shall continue until the tenth (10th) anniversary of the Effective Date, unless terminated earlier or extended pursuant to Section 10.2 (the term of this Agreement, as so modified is referred to herein as the "*Term*").
  - 10.2 Modification of Term; Termination.
- (a) The Company shall have the right at any time to terminate this Agreement for Cause by giving fifteen (15) days advance notice in writing to the Manager; and shall have the right to terminate this Agreement without Cause, by giving two-hundred and seventy (270) days advance notice in writing to the Manager; or
- (b) The Manager may terminate this Agreement at any time upon the occurrence of a Company Default by giving fifteen (15) days advance notice in writing to the Company, and shall have the right to terminate this Agreement for any reason, with or without cause, after the first anniversary of the opening of the first Facility to the general public by giving ninety (90) days advance notice in writing to the Company; or
- (c) The Manager may terminate this Agreement at any time if (i) the CB Entity no longer holds Class B Units (as such term is defined in the LLC Agreement) in the Company, (ii) Lee Roy Mitchell no longer serves as an active member of the board of directors of Cinemark Holdings, Inc. for any reason, including without limitation retirement, death or disability, or (iii) breach by Cinemark of the Theater Services Agreement which remains uncured for thirty (30) days following such breach, by giving fifteen (15) days advance notice in writing to the Company; or
  - (d) This Agreement shall automatically terminate upon termination of the Theater Services Agreement.
- 10.3 Commencement and Termination of Services with Respect to Specified Facilities. The provision of the Development Services, Management Services and Staffing Services under this Agreement shall commence with respect to each Facility as may be specified in a notice by the Company with respect to each Facility. After commencement of the Services, the provision of the Services under this Agreement shall continue in respect of such Facilities, unless terminated earlier as expressly provided in Section 10.2, until the fifteenth (15th) calendar day after the Company shall have given the Manager written notice of termination with respect to such Service(s).
- 10.4 Effects of Termination . The termination of this Agreement shall not terminate the rights and obligations of the Parties under Article XI, which rights and obligations shall only terminate upon the termination of the Transition Service Periods. The termination of this Agreement in its entirety, howsoever arising shall be without prejudice to the rights and duties of the Parties accrued prior to termination or during any Transition Service Period. The Manager's

access and control over any Company Account shall terminate upon termination of this Agreement in its entirety, or, if such access is required for the Manager to perform one or more of the Transition Services, upon the termination of the Transition Service Period applicable to such Transition Services. Notwithstanding the foregoing, the covenants, agreements, terms and conditions of this Agreement which expressly or impliedly have effect after termination (including, for the avoidance of doubt, those covenants, agreements, terms and conditions contained in Articles VII, VIII, X XI and XII of this Agreement and all indemnification and confidentiality provisions contained in this Agreement) shall continue to be enforceable notwithstanding termination, including the termination of any Transition Service Period. If this Agreement is terminated for any reason, any amounts accrued but not yet paid to the Manager hereunder shall be due and payable within ninety (90) days after the time of such termination and the Manager shall deliver to the Company hereunder shall be due and payable within ninety (90) days after the time of such termination and the Manager shall deliver to the Company the originals of all books, records, Contracts and all other documents, certificates, permits or instruments relating to a Facility, including documents evidencing termination of the Manager's right to take any action with respect to the Company Account. If possession by the Manager of any such books, records, Contracts, documents, certificates, permits or instruments are necessary for the performance of any Transition Service, Manager shall deliver the same to the Company upon termination of the Transition Service Period applicable to such Transition Services.

### ARTICLE XI TRANSITION SERVICES

- 11.1 Transition Services. If this Agreement terminates for any reason other than Cause, the Manager agrees to provide the services set forth on Exhibit D, and any other mutually agreed services determined to be necessary or appropriate to the Company in the same manner as the Services are required to be provided by this Agreement and to otherwise cooperate with the Company in transitioning the Services to a new manager(s) or in winding down the operations of the Company, as applicable, including but not limited to granting the Company access to and otherwise making available for copies any information related to the Company held by the Manager (collectively, the "Transition Services") for the respective periods set forth on Exhibit D (each such period, a "Transition Service Period"), commencing on the date on which the termination is effective ("Transition Start Date"). The Company agrees to pay the Manager as provided in this Agreement during the Transition Service Period.
- 11.2 Software Use & License. Subject to the terms and conditions of this Agreement, the Manager agrees to continue to provide any software and/or licenses that the Manager provided during the Term that are necessary for the performance of the Transition Services.
  - 11.3 Employees.

(a) Offer of Employment. During a Transition Service Period, the Company, Cinemark and any of Cinemark's Affiliates (any of the foregoing that makes any such offer, the "New Facility Employer") may make offers of employment to any Facility-Level Employees. During the Transition Service Period, the Manager will make any employees of the Manager that work on an annual basis at the Manager's corporate offices and who dedicate at least eighty percent (80%) of their working time to the operation and management of a Facility (the "Facility Corporate Employees") reasonably available to the New Facility Employer for employment interviews. Upon the commencement of the Transition Service Period, the New Facility Employer will have the option, in its sole discretion, to attempt to hire any Facility Corporate

Employee. To the extent such offers of employment are accepted, upon accepting such offers of employment, such Facility Corporate Employees and Facility-Level Employees' employment with the Manager shall terminate. The Manager will assist in the transition of such employees from the Manager to the New Facility Employer. The New Facility Employer may not make offers of employment to any employees who are not Facility-Level Employees or Facility Corporate Employees.

- (b) <u>Severance Expenses</u>. If, upon termination of this Agreement, (i) New Facility Employer does not offer employment to any Facility-Level Employee or Facility Corporate Employee, and (ii) as a direct result of the termination of this Agreement, the Manager terminates the employment of such employee, the Company will be solely responsible for, and shall reimburse the Manager for all costs incurred by the Manager associated with the failure of the New Facility Employer to provide employment to any Facility-Level Employee or Facility Corporate Employee, including, but not limited to, any severance or additional compensation. The Manager shall provide evidence of such severance costs and any other documentation or support reasonably requested by the Company prior to any such payment by the Company.
- 11.4 Financial and Tax Cooperation and Exchange of Information. The Manager shall provide and transition to the Company all financial information relating to a Facility in a manner compatible with the Company's financial reporting system. The Manager and the Company shall provide each other with such assistance as reasonably may be requested by either of them with respect to a Facility in connection with (a) the preparation of any tax return, amended tax return or claim of refund or (b) any audit or other examination by any taxing authority, or any judicial or administrative proceedings relating to liability for taxes or tax reporting by the Company (including preparation of tax reporting information to the members of the Company). The Manager and the Company will make themselves (and their respective employees) available, on a mutually convenient basis, to provide explanations of any documents or information provided under this Section 11.4. Any information obtained under this Section 11.4 shall be kept confidential, except as may be otherwise necessary in connection with the filing of tax returns or claims for refund or in conducting an audit or other proceeding.
- 11.5 Post-Closing Access. Subject to the restrictions of any applicable legal requirement, the Manager, on the one hand, and the Company, on the other hand, shall, following the termination of the Services and upon reasonable notice, afford such other party or parties and their representatives reasonable access (including the right to make photocopies, at the sole cost and expense of the parties requesting such access), to the extent necessary in connection with the Manager's operation of a Facility prior to the Transition Start Date or the Company's operation of a Facility after the Transition Start Date, during normal business hours and in a manner so as not to interfere unreasonably with such other Party, to such other Party's records related solely to a Facility, including upon specific request, a complete copy of the books and records to the extent permitted by applicable legal requirements and to the extent such books and records are under such other Party's or its Affiliates' control.
- 11.6 Further Assurances. The Parties will cooperate reasonably with each other and with their respective representatives in connection with any steps required to be taken as part of their respective obligations under this Article XI, and will (a) furnish upon request to each other such further information; (b) execute and deliver to each other such other documents; and (c) do such other acts and things, including obtain any consents and approvals required (if any), all as the other Party may reasonably request for the purpose of carrying out the intent of this Article XI and the Transition Services.

11.8 No Third Party Beneficiaries. Nothing in this Article XI, express or implied, is intended to confer on any person other than the parties hereto or their respective successors and assigns any rights, remedies, obligations or liabilities under or by reason of this Agreement. Nothing in this Article XI shall create any third party beneficiary rights in any employee or any beneficiary or dependents thereof or in any person other than the Parties to this Agreement.

### ARTICLE XII MISCELLANEOUS

### 12.1 Exclusions and Limitations of Liability.

- (a) All warranties, representations, guarantees, conditions and terms, other than those expressly set out in this Agreement, whether express or implied by statute, common law, trade usage or otherwise and whether written or oral are hereby expressly excluded to the fullest extent permissible by Law.
- (b) None of the Parties shall in any circumstances be liable for any claim, whether arising in contract, tort or otherwise, for consequential, economic, special or other indirect loss of any other Party, including losses calculated by reference to lost profits, contracts, business, goodwill, income, production or accruals of any such other Party (it being understood that if in connection with any Proceeding a Party is required to pay damages to a third party based on consequential, economic, special or other indirect losses of such third party, such liability for such damages shall be deemed to be a direct loss of the Party that is required to pay such amounts to such third party).
- Company acknowledges and agrees that the Manager shall not have any liability to the Company for any Losses made, suffered or incurred by the Company as a result of (i) the Manager's complying with the terms of this Agreement or any directions or authorizations that are provided to the Manager by the Company as specifically contemplated by this Agreement or (ii) if the Manager requests in writing an authorization from the Company under Section 2.3 (together with a reasonably detailed written explanation of the reasons for such request) in order to take a specified action but the Company does not provide such requested authorization, the consequences of not taking the actions that were the subject matter of such authorization request. In addition, the Company acknowledges and agrees that, notwithstanding anything to the contrary contained in this Agreement, the Manager shall not be required to incur any cost or make any expenditure that is not contemplated by the Approved Budget then in effect and shall have no liability to the Company for any Losses made, suffered or incurred by the Company as a result of the Manager's not incurring any such cost or making any such expenditure or failing to take any related actions (including rendering the Services) as a result thereof.
- (d) The Manager, its directors, agents, officers, employees, subsidiaries and Affiliates, as agents of the Company, shall not be liable to the Company or to any other Person for any act or omission committed in the performance of this Agreement unless such act constitutes bad faith, gross negligence, fraud or willful and wanton misconduct. Notwithstanding any other provision in this Agreement, in no event shall the Company make any claims against the Manager or its Affiliates on account of any alleged errors of judgment made in good faith in the development or operation of a Facility in accordance with the terms of this Agreement.

- (e) The Parties accept that the limitations and exclusions set out in this Section 12.1 are reasonable having regard to all the circumstances.
- 12.2 Company Subsidiaries. The Parties acknowledge that the Company may cause one or more Facilities to be owned by a Company Subsidiary. The Manager agrees that the Company may designate one or more Company Subsidiaries to receive the Services to be performed by the Manager with respect to any Facility and all references herein to the Company shall be construed to include any such Company Subsidiary. For the purpose of this Agreement, Company Account may include one or more accounts of a Company Subsidiary designated by the Company to be used with respect to the Services provided by Manager to such Company Subsidiary.
- 12.3 Succession and Assignment; No Third-Party Beneficiaries. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. No Party may assign, delegate, sub-contract or otherwise transfer either this Agreement or any of its rights, interests or obligations hereunder without the prior written approval of the Company (in the case of a transfer by the Manager) or the Manager (in the case of a transfer by the Company); provided, that the Manager shall be entitled to (a) designate one or more Affiliates of the Manager to perform its obligations hereunder and/or (b) sub-contract with third parties the performance of its duties under this Agreement, provided that no such designation or sub-contract arrangement shall relieve any Party of its obligations under this Agreement. Except as expressly provided herein, this Agreement is for the sole benefit of the Parties and nothing in this Agreement (whether expressed or implied) will give or be construed to give any Person, other than the Parties, any legal or equitable rights in connection with this Agreement. Any purported assignment in breach of this Section 12.3 shall be void and confer no rights on the purported assignee.
- 12.4 Transaction Costs. Except as otherwise specified herein, the Company shall bear the costs and expenses in connection with the preparation, negotiation, execution and performance of this Agreement.

### 12.5 Notices and Communications.

(a) All notices, requests, demands, claims and other communications required or permitted to be delivered, given or otherwise provided under this Agreement must be in writing and must be either personally delivered, mailed by first class mail (postage prepaid and return receipt requested), sent by electronic mail with confirmation of transmission by the transmitting equipment or sent by reputable overnight courier service (charges prepaid) to the recipient at the address below indicated:

If to the Company or a Company Subsidiary:

FE Concepts, LLC 12400 Coit Road, Suite 800 Dallas, TX 75251 Attention: Gary Witherspoon

E-mail: gw@cbcap.com

with a copy (which shall not constitute notice) to:

Cinemark USA, Inc. 3900 Dallas Parkway, Suite 500 Plano, Texas 75093

Attention: Michael Cavalier, General Counsel

E Mail: mcavalier@cinemark.com

If to the Manager, to it at:

Cinema Operations, L.L.C. 12400 Coit Road, Suite 800 Dallas, Texas 75251 Attention: W. Mark Moore E-Mail: mmoore@epgusa.com

with a copy (which shall not constitute notice) to:

Cinemark USA, Inc. 3900 Dallas Parkway, Suite 500 Plano, Texas 75093

Attention: Michael Cavalier, General Counsel

E Mail: mcavalier@cinemark.com

- (b) Each of the Parties may specify an address or email address different than that set forth in <u>Section 12.5(a)</u> by giving notice in accordance with <u>Section 12.5(a)</u> to each of the other Parties.
- 12.6 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. In the event that any provision hereof would, under applicable Law, be invalid or unenforceable in any respect, each Party hereto intends that such provision will be construed by modifying or limiting it so as to be valid and enforceable to the maximum extent compatible with, and possible under, applicable Law.
  - 12.7 Amendments, Waivers and Remedies.
- (a) No amendment or waiver of any provision of this Agreement will be valid and binding unless it is in writing and signed, in the case of an amendment, by each of the Parties hereto, or in the case of a waiver, by the Party against whom the waiver is to be effective.
- (b) No waiver by any Party of any breach or violation of, or default under or inaccuracy in, any representation, warranty or covenant hereunder, whether intentional or not, will extend to any prior or subsequent breach or violation of, or default under or inaccuracy in, any such representation, warranty or covenant or affect in any way any rights arising by virtue of any such prior or subsequent occurrence. No delay or omission on the part of any Party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof.

### 12.8 Confidentiality.

- (a) Subject to Section 12.8(b), the Parties shall treat as strictly confidential all information received or obtained as a result of entering into or performing this Agreement which relates to: (i) the provisions of this Agreement, or any document entered into pursuant to this Agreement; (ii) the negotiations relating to this Agreement; (iii) the provision of the Services or the Business; and/or (iv) any of the other Parties, or their respective businesses, operations, assets, liabilities, financial condition or results of operations.
- (b) The restrictions in <u>Section 12.8(a)</u> shall not apply to prevent the use or disclosure of any confidential information if the relevant use or disclosure: (i) is required by applicable Law or any Governmental Authority of competent jurisdiction, subject to the condition that the disclosing Party will provide notice of such requirement to the non-disclosing Parties prior to disclosing any confidential information; (ii) is made to any legal, financial or other adviser, auditor, financing source, shareholder or other Affiliate of the disclosing Party, subject to the condition that the Party making the disclosure shall be responsible for ensuring that such Persons comply with this <u>Section 12.8</u> as if they were parties to this Agreement; (iii) is made to the officers or employees of the disclosing Party or any Affiliate thereof who need to know the information for the purposes of the transactions effected or contemplated by this Agreement, subject to the condition that the Party making the disclosure shall be responsible for ensuring that such Persons comply with this <u>Section 12.8</u> as if they were parties to this Agreement; (iv) is in or becomes part of the public domain other than through an action of any Party; or (v) is approved in writing in advance by the Company (in the case of a disclosure by the Manager) or the Manager (in the case of a disclosure by the Company), as the case may be.
- 12.9 Counterparts. This Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which together will constitute but one and the same instrument.
- 12.10 Entire Agreement. This Agreement constitutes the entire agreement among the Parties hereto with respect to the subject matter hereof and supersedes any and all prior discussions, negotiations, proposals, undertakings, understandings and agreements, whether written or oral, with respect to such subject matter.
- 12.11 No Partnership, Etc.. Nothing in this Agreement or any document referred to in it or any arrangement contemplated by it shall be construed as creating a joint venture or partnership between the Parties for any purpose whatsoever and no Party shall have the power or authority to bind any other Party or impose any obligations on it to the benefit of any third party.
- 12.12 Construction. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement. Except as otherwise explicitly specified to the contrary, (a) references to a Section, Article or Schedule mean a Section or Article of, or Schedule to, this Agreement, (b) the word "including" will be construed as "including without limitation," (c) references to a particular statute or regulation include all rules and regulations thereunder and any predecessor or successor statute, rules or regulation, in each case as amended or otherwise modified from time to time and any reference to a document is to that document as varied, supplemented or replaced from time to time by agreement between the parties thereto, (d) the

headings contained in this Agreement are for convenience purposes only and will not in any way affect the meaning or interpretation of this Agreement, (e) words in the singular or plural form include the plural and singular form, respectively, (f) use of any gender includes the other genders, (g) any reference to writing shall include any modes of reproducing words in a legible and non-transitory form, (h) references to a particular Person include such Person's successors and assigns to the extent not prohibited by this Agreement, and (i) the Schedules and Recitals hereto form part of this Agreement and shall have effect as if set out in full in the body of this Agreement, and any reference to this Agreement includes the Schedules and Recitals hereto.

- 12.13 Governing Law. This Agreement, the rights of the Parties hereunder and all Proceedings arising in whole or in part under or in connection herewith, will be governed by and construed in accordance with the domestic substantive laws of the State of Texas, without giving effect to any choice or conflict of law provision or rule that would cause the application of the laws of any other jurisdiction.
- 12.14 Arbitration. Any claim, dispute or other matter in question between the Parties hereto arising out of or relating to this Agreement, the rights of the Parties hereunder or the breach hereof, shall be decided by arbitration in accordance with the rules of the American Arbitration Association in effect on the Effective Date before three arbitrators; one designated by the Company, one designated by the Manager and the third designated in accordance with the Rules of the American Arbitration Association. Any such arbitration shall be conducted in Dallas, Texas, unless the Parties mutually agree to another location. The arbitrators shall be qualified by education, training or experience as may be appropriate according to the nature of the claim, dispute or other matter in question. The foregoing agreement to arbitrate and any other agreement to arbitrate shall be specifically enforceable under the prevailing arbitration law. The award rendered by the arbitrators shall be final, and judgment may be entered upon it in accordance with applicable Law in any court having jurisdiction thereof. To the extent permitted by law, by agreeing to engage in the arbitration provided for in this Section 12.14, the Parties waive their right to appeal any decision made by the arbitrators. The demand for arbitration shall be made within a reasonable time after the claim, dispute or other matter in question has arisen; and in no event shall it be made after the date when institution of legal or equitable proceedings based on such claim, dispute or other matter in question would be barred by the applicable statute of limitations. All costs and expenses (including reasonable attorneys' fees and costs) incurred in connection with any such arbitration shall be borne in the manner which the arbitrators making the determination shall direct. Notwithstanding the provisions of this Section 12.14, either Party may seek appropriate injunctive relief for any threatened breach.
- 12.15 Affiliate Transactions . Subject to Article III, in connection with the provision of the Management Services under this Agreement, the Manager may purchase necessary goods, supplies, rights and services (a) from or through any of its Affiliates or (b) pursuant to arrangements Manager Related Facilities so long as (i) any such transaction, series of transactions or arrangement is disclosed to the Company and approved by the Board including at least one Board member not affiliated with the Manager, (ii) in respect of any such Affiliate transaction or series of transactions, the terms and conditions thereof are competitive with the prices and terms of goods, supplies, rights and services of like quality available from non -Affiliated third parties in an arm's length transaction, and (iii) in respect of such shared arrangements, the allocations of costs with respect thereto are fair and reasonable and, if applicable, consistent with the methodology used by the Manager and its subsidiaries and Affiliates to allocate similar costs among the Manager Related Facilities.

- 12.16 Authority; Binding Agreement. Each Party represents and warrants that (a) it is a corporation, limited liability company, partnership or other entity, as applicable, duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, (b) it has the requisite power and authority to execute, deliver and perform this Agreement, (c) the execution, delivery and performance by it of this Agreement has been duly and validly authorized by all necessary actions on its part, and (d) this Agreement constitutes the legal, valid and binding obligations of such Party, enforceable against such Party in accordance with its terms.
- 12.17 Further Assurances. Each Party hereto shall further execute and deliver all such other appropriate supplemental agreements and other instruments and take such other action as which are or may be necessary or appropriate to effectuate and carry out the provisions of this Agreement, may be necessary to make this Agreement fully and legally effective, binding and enforceable as between the Parties and as against third parties, or as the other Parties may reasonably request.
- 12.18 Cooperation. The Parties will, and will cause their respective Affiliates to, in good faith, diligently communicate, consult and cooperate with each other to facilitate each other's performance of their respective and joint responsibilities and duties under this Agreement, including, but not limited to, any licensing responsibilities and duties. The Parties will, and will cause their respective Affiliates to, deal with each other on all matters related to this Agreement and otherwise in good faith in an open manner as promptly as possible under the given circumstances. Wherever an approval or consent is required in this Agreement by any Party, such approval or consent shall not be unreasonably withheld, delayed or conditioned. Each Party shall act reasonably with respect to the Party's responsibilities and obligations under this Agreement.
- 12.19 Other Activities. The Company acknowledges that the Manager and its Affiliates are in the business of developing and operating entertainment facilities that may be competitive with a Facility and that the Manager and its Affiliates may in the future be presented with opportunities to develop new entertainment facilities and acquire existing entertainment facilities. Except as may be set forth herein, the Parties agree that nothing in this Agreement shall be construed to restrict the Manager from pursuing opportunities to develop new entertainment facilities and acquire existing entertainment facilities, and the pursuit of those ventures or activities by the Manager or its Affiliates shall not be deemed wrongful or improper, even to the extent they may be competitive with a Facility or might otherwise constitute an opportunity for, the Company. Without limiting the foregoing, the Manager and its Affiliates may pursue concepts, ventures and opportunities on behalf of themselves rather than on behalf of the Company and need not offer, and the Company renounces any interest or expectancy as to participate in, such concepts, ventures and opportunities to the Company. Notwithstanding the foregoing, for a period of five (5) years after the termination of this Agreement neither Manager nor its Affiliates shall own, operate or acquire entertainment facilities similar to a Facility within a eight (8) mile radius from any Facility owned by the Company.

[SIGNATURE PAGES FOLLOW]

written.	
	THE COMPANY:
	FE CONCEPTS, LLC
	By: Name: Title:
	THE MANAGER:
	CINEMA OPERATIONS, L.L.C.
	By: Name: Title:
	C-23

IN WITNESS WHEREOF, each of the undersigned has executed this Management Services Agreement as of the date first above

#### EXHIBIT D

### FORM OF THEATER SERVICES AGREEMENT

THIS THEATER SERVICES AGREEMENT (this "Agreement") is dated and made effective as of April 20, 2018 (the "Effective Date"), by and between FE Concepts, LLC, a Texas limited liability company (the "Company") and CNMK Texas Properties, LLC, a Texas limited liability company ("Cinemark" and, together with any of its subsidiaries or parent from time to time providing Services hereunder, the "Consultant"). The Consultant and the Company are sometimes referred to in this Agreement, collectively, as the "Parties" and each individually as a "Party." Defined terms used in this Agreement but not otherwise defined herein have the meanings ascribed to them on Schedule I hereto.

#### RECITALS

- G. Consultant and AWSR Investments, LLC, a Texas limited liability company, are parties to that certain Limited Liability Company Agreement, dated as of April 20, 2018 (the "*LLC Agreement*"), pursuant to which CB Entity and Consultant have formed the Company to own and operate a certain family entertainment facilities, or facilities, incorporating games, rides, bowling, restaurants, bars and motion picture theater auditoriums (the "*Facilities*").
- H. The Consultant has the qualified, experienced and necessary personnel to consult with the Company regarding the construction and operation of the motion picture theater auditoriums (the "Auditoriums") to comprise a portion of a Facility.
- I. The LLC Agreement provides in part that the Company shall engage the Consultant to provide certain services pursuant to Article I of this Agreement (the "*Theater Services*").
- J. Cinema Management, L.L.C. (the "Manager") is a party to a Management Services Agreement with the Company to provide comprehensive management services regarding the construction, development, operation and staffing of any Facility (collectively, the "Management Services").
- K. The Company desires to retain the Consultant to provide the Theater Services with respect to any Facility and the Consultant is willing to provide such Theater Services, upon the terms and subject to the conditions set forth in this Agreement.
- **NOW**, **THEREFORE**, in consideration of the mutual promises and covenants contained in this Agreement and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties, intending to be legally bound, agree as follows:

### ARTICLE I THEATER SERVICES

- 1.1 Auditorium Construction. The Consultant agrees to consult with the Company and the Manager regarding the construction of the Auditoriums to be constructed at any Facility. The Consultant shall advise the Company and the Manager with respect to:
  - (a) recommendations regarding the number, size and design of the Auditoriums to be located in a Facility;
- ( a ) recommendations regarding architects, engineers, contractors, suppliers and vendors in connection with the construction of the Auditoriums;
  - (b) recommendations regarding furniture, fixtures and equipment to be used in the Auditoriums; and
- ( c ) consultations as may be reasonably requested by the Company or Manager during the construction of the Auditoriums at a Facility.
  - 1.2 Theater Technologies. The Consultant agrees to provide the following services relating to theater technologies:
- (a) select all digital projection, TMS, screen and sound equipment for purchase by the Company or the Manager on the Company's behalf;
  - (b) install, or supervise the installation of, all digital projection and TMS equipment;
  - (c) perform remote support of the digital cinema systems through CSC;
  - (d) manage the warranty and RMA of the digital cinema systems; and
  - (e) perform or supervise annual maintenance, upgrades and emergency support using the Consultant's booth engineers.
- 1.3 Film Booking and Buying. The Consultant agrees to book and license or cause to be licensed on the Company's behalf, films and other attractions for exhibition in the Auditoriums. The Consultant is hereby authorized by the Company to execute film license agreements on behalf of the Company for any Facility.
- 1.4 Training. The Consultant agrees to provide necessary and appropriate training both at a Facility and the Consultant Related Facilities to the Manager's facility-level personnel who will be assigned to conduct the exhibition of films in the Auditoriums, including ticketing personnel, certain concession operations and operators of film projection equipment and any other training necessary to fully plan, construct, equip, and successfully operate the Auditoriums. Training will be conducted at the Consultant's sole election at a Facility or at the Consultant's facilities.

1.5 Theater Operations. The Cooperations including:	nsultant will consult with the Company and the M	Manager regarding the motion picture theater
(a) ticket pricing policie	es;	

- (b) film booking;
- (c) film settlement and settlement of any disputes with the distributors of content booked by the Consultant;
- (d) scheduling of motion picture exhibition times;
- (e) publication of show times;
- (f) management of studio relations and weekly trailer programming;
- (g) management of day-to-day studio relations, in-theater and on-line promotions;
- (h) management of trailer placements;
- (i) management of studio partnership marketing funds as applicable to the films shown at a Facility
- (j) arranging for on-screen advertising;
- (k) recommendations regarding ticketing systems and any third party ticketing vendors;
- ( 1 ) recommendations regarding personnel requirements in connection with motion picture exhibition in the Auditoriums; and
- ( m ) recommendations regarding budget for line items relating to the motion picture theater auditoriums, including without limitation film rental expense, in theatre marketing and projector maintenance and repair.
- 1.6 Exclusive Provider of Theater Services. Unless otherwise expressly agreed by the Consultant with respect to any Facility, the Consultant shall be the exclusive provider of the Theater Services contemplated by this Agreement with respect to each Facility acquired, owned or operated by the Company.
- 1.7 Company Approval. For the purpose of this Agreement, any consent or approval required or permitted to be given by the Company regarding Theater Services shall require the approval of the Board (including either Lee Roy Mitchell or at least one Board member not affiliated with the Consultant) or, the approval of an executive officer of the Company not affiliated with the Consultant to the extent that specific authority to grant such consent or approval has been expressly delegated to such officer by the Board, subject to any additional approval required by the Members in accordance with the LLC Agreement. Any notice by the Consultant to the Company shall be given in accordance with Section 6.5.

### ARTICLE II ACKNOWLEDGEMENTS

- 2.1 Delegation of Duties. Notwithstanding anything to the contrary contained in this Agreement, the Consultant shall have the right to delegate or subcontract its duties under this Agreement to any Person; provided, however, that no such delegation or subcontracting shall relieve the Consultant of liability under this Agreement and further provided that the Consultant shall supervise the activities of such subcontractor, and provided, further that the Consulting Fee continues to be paid to the Consultant.
- 2.2 No Restrictions on the Consultant. The Consultant shall not be required to spend its full time and attention to provide any of the Services. The Consultant shall devote such time and personnel as the Consultant deems appropriate to provide the Services pursuant to this Agreement. The Company acknowledges that the Consultant and its Affiliates are in the business of developing and acquiring motion picture exhibition theaters that may be competitive with a Facility and that the Consultant and its Affiliates may in the future be presented with opportunities to develop new theaters and acquire existing theaters. Except as may be set forth herein, the Parties agree that nothing in this Agreement shall be construed to restrict the Consultant from pursuing opportunities to develop new theaters and acquire existing theaters, and the pursuit of those ventures or activities by the Consultant or its Affiliates shall not be deemed wrongful or improper, even to the extent they are competitive with, or might otherwise constitute an opportunity for, the Company. Without limiting the foregoing, the Consultant and its Affiliates may pursue concepts, ventures and opportunities on behalf of themselves rather than on behalf of the Company and need not offer, and the Company renounces any interest or expectancy as to participate in, such concepts, ventures and opportunities to the Company.
- 2.3 Film Settlement Policies. The Company acknowledges that subject to the next sentence, (i) with respect to booking services, numerous subjective and nonquantifiable matters of judgment go into the decision of which films to book and the terms of such booking and the Consultant, for a variety of reasons, may find it necessary to book films of certain distributors; (ii) separate geographic areas may be unique and involve factors relevant to policies employed in the negotiation and effectuation of film settlement ("Settlement Policies") that are not of equal importance to other geographical areas, (iii) Settlement Policies for one area may not necessarily be evaluated against or compared to the Settlement Policies the Consultant or its Affiliates employ or employed in other geographic areas or markets, and (iv) the Consultant will not be required to employ the same or similar Settlement Policies hereunder as it or its Affiliates follow in other geographic areas or markets or for other theaters owned, leased or managed by the Consultant in the same geographical area; provided that the Consultant shall conduct the Settlement Policies in a good and businesslike manner and in a manner consistent with Consultant Related Facilities. The Company confirms the Consultant has made no representation or covenant as to the Settlement Policies or booking policies it has or will have with respect to any theaters which are owned, leased or managed by the Consultant or its Affiliates or which are to be managed pursuant to the Management Agreement other than that Consultant will employ the same policies and practices in respect thereof on behalf of the Company as it does on behalf of the Consultant's theaters and that the Consultant will act in good faith in respect thereof.

- 2.4 No Restriction on Payment of Consulting Fee. The Company agrees not to directly or indirectly create or otherwise cause or suffer to exist or become effective any consensual encumbrance or the restriction on the ability of the Company to pay to the Consultant any fees or other amounts payable to the Consultant under this Agreement.
- 2.5 Management Agreement. The Consultant acknowledges that the Company has entered into the Management Agreement with the Manager pursuant to which the Manager shall provide the Management Services. The Company hereby authorizes the Consultant to coordinate providing the Theater Services with the Manager. The Consultant is authorized to rely on instructions from the Manager and advise the Manager's personnel regarding the various recommendations to be provided by the Consultant under this Agreement. The Company authorizes and directs the Manager to pay all Consulting Fees to the Consultant, and reimburse the Consultant for all Consultant Expenses incurred by the Consultant, as provided in this Agreement.

### ARTICLE III CONSULTING FEE

- 3 . 1 Consulting Fees. As compensation for its services under this Agreement, the Company will pay Consultant fees (the "Consulting Fees") as set forth on Schedule II hereto. Consultant shall invoice the Company monthly in arrears for Consulting Fees due with respect to the Theater Services performed by Consultant each month during the term of this Agreement. All invoices shall be due and payable within thirty days after receipt by the Company.
- 3.2 Consulting Expenses. The Company shall reimburse the Consultant for its out-of-pocket expenses incurred in providing the Theater Services, including, but not limited to travel expenses to any Facility or proposed Facility site, any Company Trademark related fees or expenses advanced by the Consultant in its reasonable discretion and such other expenses as may be reasonably incurred by the Consultant as set forth on Schedule III hereto ("Consultant Expenses").

## ARTICLE IV INDEMNITY

4.1 Indemnification by the Company. The Company shall indemnify and hold harmless the Consultant and its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (and the members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons) (such indemnified Persons, the "Consultant Indemnified Persons") from, against and in respect of any and all claims, Proceedings, obligations, liabilities, liens, encumbrances, losses, damages, assessments, fines, penalties, taxes, fees, costs (including costs of investigation, defense and enforcement of this Agreement, including attorneys' fees and disbursements), expenses or amounts paid in settlement (in each case, including reasonable attorneys' and experts fees and expenses) (collectively, "Losses"), in the case of clauses (b) and (c) below, whether or not involving a Third Party Claim, to the extent such Losses are incurred or suffered by the Consultant Indemnified Persons as a result of, arising out of or directly or indirectly relating to:

- (a) any claim or Proceeding made, initiated or threatened by any Person other than a Party (a "Third Party Claim") that relates to the provision of the Services under this Agreement, the management and operation of a Facility or the Business, or the consequences of the Consultant's compliance with the terms of this Agreement or any directions provided by the Company or the Manager relating to the operation of the Business, other than to the extent such Losses are caused by the bad faith, gross negligence or willful misconduct of the Consultant or any of its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (or the shareholders, members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons);
  - (b) all Consultant Expenses;
  - (c) any breach of this Agreement by the Company.
- 4 . 2 Indemnification by the Consultant. The Consultant shall indemnify and hold harmless the Company, the Company Subsidiaries and the members, managers, officers, directors, employees and other agents, representatives and Affiliates thereof (and the members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons) (such indemnified Persons, the "Company Indemnified Persons") from, against and in respect of any and all Losses to the extent such Losses are incurred or suffered by the Company Indemnified Persons as a result of, arising out of or directly or indirectly relating to any Third Party Claim that relates to the provision of the Theater Services under this Agreement to the extent such Losses are caused by the bad faith, gross negligence or willful misconduct of the Consultant or any of its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (or the shareholders, members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons);
- 4.3 Insurance Recoveries. The amount of Losses for which indemnification is available under Section 4.1 and 4.2 shall be calculated net of any amounts actually recovered by the Company Indemnified Persons or the Consultant Indemnified Persons, as the case may be, under insurance policies with respect to such Losses. In the event that any indemnifying person makes any payment under Section 4.1 or 4.2 in respect of any Losses, such indemnifying person shall be subrogated, to the extent of such payment, to the rights of such indemnified person against any insurer or other third Person with respect to such Losses.

# ARTICLE V TERM AND TERMINATION

- 5 . 1 Term of Agreement. This Agreement shall commence on the Effective Date and shall continue until the tenth (10<sup>th</sup>) anniversary of the Effective Date, unless terminated earlier or extended pursuant to Section 5.2 (the term of this Agreement, as so modified is referred to herein as the "Term").
  - 5.2 Modification of Term; Termination.
- (a) The Consultant shall have the right at any time by giving (i) fifteen (15) days advance notice in writing to the Company to terminate this Agreement for non-payment under this Agreement if such non-payment is not resolved within fifteen (15) business days following

written notice or (ii) after the first Facility has been in operation for a full year, two-hundred and seventy (270) days advance notice to terminate this Agreement, for any reason, with or without cause; or

- (b) The Company may terminate this Agreement at any time for Cause by giving fifteen (15) days advance notice in writing to the Consultant; provided that the Company shall not be permitted to terminate this Agreement without Cause without the prior written consent of the Consultant at any time prior to the expiration of the Term; or
- (c) This Agreement shall automatically terminate upon termination of the Management Agreement and termination of the Transition Services Period (if applicable).
- 5.3 Commencement and Termination of Specified Theater Services. The provision of the Theater Services under this Agreement shall commence with respect to each Facility on such date prior to the acquisition of the site for such Facility (or acquisition of any existing Facility) as the Company may specify. After commencement of the Theater Services, the provision of the Theater Services under this Agreement shall continue in respect of such Facilities, unless terminated earlier as expressly provided in Section 5.2.
- 5.4 Effects of Termination. The termination of this Agreement in its entirety (or the termination of any of the Theater Services with respect to the Facilities), howsoever arising, shall be without prejudice to the rights and duties of the Parties accrued prior to termination. The covenants, agreements, terms and conditions of this Agreement which expressly or impliedly have effect after termination shall continue to be enforceable notwithstanding termination including all indemnification and confidentiality provisions contained in this Agreement. If this Agreement (or the termination of any of the Theater Services with respect to the Facilities) is terminated for any reason, any amounts accrued but not yet paid to the Consultant hereunder shall be due and payable within thirty (30) days after the time of such termination.

### ARTICLE VI MISCELLANEOUS

- 6.1 Exclusions and Limitations of Liability.
- (a) All warranties, representations, guarantees, conditions and terms, other than those expressly set out in this Agreement, whether express or implied by statute, common law, trade usage or otherwise and whether written or oral are hereby expressly excluded to the fullest extent permissible by Law.
- (b) None of the Parties shall in any circumstances be liable for any claim, whether arising in contract, tort or otherwise, for consequential, economic, special or other indirect loss of any other Party, including losses calculated by reference to lost profits, contracts, business, goodwill, income, production or accruals of any such other Party (it being understood that if in connection with any Proceeding a Party is required to pay damages to a third party based on consequential, economic, special or other indirect losses of such third party, such liability for such damages shall be deemed to be a direct loss of the Party that is required to pay such amounts to such third party).

- (c) The Company acknowledges and agrees that the Consultant shall not have any liability to the Company for any Losses made, suffered or incurred by the Company as a result of (i) the Consultant's complying with the terms of this Agreement or any directions or authorizations that are provided to the Consultant by the Company or the Manager as specifically contemplated by this Agreement or (ii) if the Consultant requests in writing an authorization from the Company or the Manager (together with a reasonably detailed written explanation of the reasons for such request) in order to take a specified action but the Company or the Manager does not provide such requested authorization, the consequences of not taking the actions that were the subject matter of such authorization request.
- (d) The Consultant, its directors, agents, officers, employees, subsidiaries and Affiliates, as agents of the Company, shall not be liable to the Company or to any other Person for any act or omission committed in the performance of this Agreement unless such act constitutes bad faith, gross negligence, fraud or willful and wanton misconduct. Notwithstanding any other provision in this Agreement, in no event shall the Company make any claims against the Consultant or its Affiliates on account of any alleged errors of judgment made in good faith in the development or operation of the Auditoriums in accordance with the terms of this Agreement.
- (e) The Parties accept that the limitations and exclusions set out in this <u>Section 6.1</u> are reasonable having regard to all the circumstances.
- 6.2 Company Subsidiaries. The Parties acknowledge that the Company may cause one or more Facilities to be owned by a Company Subsidiary. The Consultant agrees that the Company may designate one or more Company Subsidiaries to receive the Theater Services to be performed by the Consultant with respect to any Facility and all references herein to the Company shall be construed to include any such Company Subsidiary; provided however, that the Company and any Company Subsidiary receiving any Theater Services from the Consultant shall be jointly and severally liable for the Consulting Fee, Consultant Expenses and indemnification obligations payable under this Agreement.
- 6.3 Succession and Assignment; No Third-Party Beneficiaries. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. No Party may assign, delegate, sub-contract or otherwise transfer either this Agreement or any of its rights, interests or obligations hereunder without the prior written approval of the Company (in the case of a transfer by the Consultant) or the Consultant (in the case of a transfer by the Company); provided, that the Consultant shall be entitled to (a) designate one or more Affiliates of the Consultant to perform its obligations hereunder and/or (b) sub-contract with third parties (other than a Class B Competing Business) the performance of its duties with respect Theater Services provided that no such designation or sub-contract arrangement shall relieve any Party of its obligations under this Agreement; provided, further, that the Consultant shall not be entitled to assign this Agreement to a Class B Competing Business without the Company's and Manager's prior written consent (except where such assignment is a result of the change in Control of Consultant). The Manager is intended to be a third party beneficiary of this Section 6.3. Except as expressly provided herein, this Agreement is for the sole benefit of the Parties and nothing in this Agreement (whether expressed or implied) will give or be construed to give any Person, other than the Parties, any legal or equitable rights in connection with this Agreement. Any purported assignment in breach of this Section 6.3 shall be void and confer no rights on the purported assignee.

6.4 Transaction Costs. Except as otherwise specified herein, the Company shall bear the costs and expenses in connection with the preparation, negotiation, execution and performance of this Agreement.

#### 6.5 Notices and Communications.

(a) All notices, requests, demands, claims and other communications required or permitted to be delivered, given or otherwise provided under this Agreement must be in writing and must be either personally delivered, mailed by first class mail (postage prepaid and return receipt requested), sent by electronic mail with confirmation of transmission by the transmitting equipment or sent by reputable overnight courier service (charges prepaid) to the recipient at the address below indicated:

If to the Company or a Company Subsidiary:

FE Concepts, LLC 12400 Coit Road, Suite 800 Dallas, TX 75251 Attention: Gary Witherspoon E-mail: gw@cbcap.com

If to the Consultant:

CNMK Texas Properties, LLC 3900 Dallas Parkway Plano, Texas 75093 Attention: Michael Cavalier, General Counsel E Mail: mcavalier@cinemark.com

- (b) Each of the Parties may specify an address or email address different than that set forth in Section 6.5(a) by giving notice in accordance with Section 6.5 (a) to the other Parties.
- 6.6 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. In the event that any provision hereof would, under applicable Law, be invalid or unenforceable in any respect, each Party hereto intends that such provision will be construed by modifying or limiting it so as to be valid and enforceable to the maximum extent compatible with, and possible under, applicable Law.
  - 6.7 Amendments, Waivers and Remedies.
- (a) No amendment or waiver of any provision of this Agreement will be valid and binding unless it is in writing and signed, in the case of an amendment, by each of the Parties hereto, or in the case of a waiver, by the Party against whom the waiver is to be effective; provided, that Section 6.3 may not be amended without the consent of the Manager.

(b) No waiver by any Party of any breach or violation of, or default under or inaccuracy in, any representation, warranty or covenant hereunder, whether intentional or not, will extend to any prior or subsequent breach or violation of, or default under or inaccuracy in, any such representation, warranty or covenant or affect in any way any rights arising by virtue of any such prior or subsequent occurrence. No delay or omission on the part of any Party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof.

### 6.8 Confidentiality.

- (a) Subject to Section 6.8(b), the Parties shall treat as strictly confidential all information received or obtained as a result of entering into or performing this Agreement which relates to: (i) the provisions of this Agreement, or any document entered into pursuant to this Agreement; (ii) the negotiations relating to this Agreement; (iii) the provision of the Theater Services or the Business; and/or (iv) any of the other Parties, or their respective businesses, operations, assets, liabilities, financial condition or results of operations.
- (b) The restrictions in Section 6.8(a) shall not apply to prevent the use or disclosure of any confidential information if the relevant use or disclosure: (i) is required by applicable Law or any Governmental Authority of competent jurisdiction, subject to the condition that the disclosing Party will provide notice of such requirement to the non-disclosing Parties prior to disclosing any confidential information; (ii) is made to any legal, financial or other adviser, auditor, financing source, shareholder or other Affiliate of the disclosing Party, subject to the condition that the Party making the disclosure shall be responsible for ensuring that such Persons comply with this Section 6.8 as if they were parties to this Agreement; (iii) is made to the officers or employees of the disclosing Party or any Affiliate thereof who need to know the information for the purposes of the transactions effected or contemplated by this Agreement, subject to the condition that the Party making the disclosure shall be responsible for ensuring that such Persons comply with this Section 6.8 as if they were parties to this Agreement; (iv) is in or becomes part of the public domain other than through an action of any Party; or (v) is approved in writing in advance by the Company (in the case of a disclosure by the Consultant) or the Consultant (in the case of a disclosure by the Company), as the case may be. For the avoidance of doubt, it is understood and agreed that the Consultant and/or its Affiliates are required to make public disclosures of material facts and events under U.S. federal securities law and that nothing contained in this Agreement shall be construed to prevent the Consultant or such Affiliates from making such public disclosures (including, without limitation, public filings or disclosures relating to this Agreement, the Company, or the Business of the Company) as they determine, in their sole and absolute discretion, is required by applicable law or regulations.
- 6.9 Counterparts. This Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which together will constitute but one and the same instrument.
- 6.10 Entire Agreement. This Agreement constitutes the entire agreement among the Parties hereto with respect to the subject matter hereof and supersedes any and all prior discussions, negotiations, proposals, undertakings, understandings and agreements, whether written or oral, with respect to such subject matter.

- 6.11 No Partnership, Etc.. Nothing in this Agreement or any document referred to in it or any arrangement contemplated by it shall be construed as creating a joint venture or partnership between the Parties for any purpose whatsoever and no Party shall have the power or authority to bind any other Party or impose any obligations on it to the benefit of any third party.
- 6.12 Construction. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement. Except as otherwise explicitly specified to the contrary, (a) references to a Section, Article or Schedule mean a Section or Article of, or Schedule to, this Agreement, (b) the word "including" will be construed as "including without limitation," (c) references to a particular statute or regulation include all rules and regulations thereunder and any predecessor or successor statute, rules or regulation, in each case as amended or otherwise modified from time to time and any reference to a document is to that document as varied, supplemented or replaced from time to time by agreement between the parties thereto, (d) the headings contained in this Agreement are for convenience purposes only and will not in any way affect the meaning or interpretation of this Agreement, (e) words in the singular or plural form include the plural and singular form, respectively, (f) use of any gender includes the other genders, (g) any reference to writing shall include any modes of reproducing words in a legible and non-transitory form, (h) references to a particular Person include such Person's successors and assigns to the extent not prohibited by this Agreement, and (i) the Schedules and Recitals hereto form part of this Agreement and shall have effect as if set out in full in the body of this Agreement, and any reference to this Agreement includes the Schedules and Recitals hereto.
- 6.13 Governing Law. This Agreement, the rights of the Parties hereunder and all Proceedings arising in whole or in part under or in connection herewith, will be governed by and construed in accordance with the domestic substantive laws of the State of Texas, without giving effect to any choice or conflict of law provision or rule that would cause the application of the laws of any other jurisdiction.
- 6.14 Arbitration. Any claim, dispute or other matter in question between the Parties hereto arising out of or relating to this Agreement, the rights of the Parties hereunder or the breach hereof, shall be decided by arbitration in accordance with the rules of the American Arbitration Association in effect on the Effective Date before three arbitrators; one designated by the Company, one designated by the Consultant and the third designated in accordance with the Rules of the American Arbitration Association. Any such arbitration shall be conducted in Dallas, Texas, unless the Parties mutually agree to another location. The arbitrators shall be qualified by education, training or experience as may be appropriate according to the nature of the claim, dispute or other matter in question. The foregoing agreement to arbitrate and any other agreement to arbitrate shall be specifically enforceable under the prevailing arbitration law. The award rendered by the arbitrators shall be final, and judgment may be entered upon it in accordance with applicable Law in any court having jurisdiction thereof. To the extent permitted by law, by agreeing to engage in the arbitration provided for in this Section 6.14, the Parties waive their right to appeal any decision made by the arbitrators. The demand for arbitration shall be made within a reasonable time after the claim, dispute or other matter in question has arisen; and in no event shall it be made after the date when institution of legal or equitable proceedings

based on such claim, dispute or other matter in question would be barred by the applicable statute of limitations. All costs and expenses (including reasonable attorneys' fees and costs) incurred in connection with any such arbitration shall be borne in the manner which the arbitrators making the determination shall direct. Notwithstanding the provisions of this Section 6.14, either Party may seek appropriate injunctive relief for any threatened breach.

- 6.15 Affiliate Transactions. In connection with the provision of the Services under this Agreement, the Consultant may purchase necessary goods, supplies, rights and services (a) from or through any of its Affiliates or (b) pursuant to arrangements Consultant Related Facilities so long as (i) any such transaction, series of transactions or arrangement is disclosed to the Company and approved by the Board including at least one Board member not affiliated with the Consultant, (ii) in respect of any such Affiliate transaction or series of transactions, the terms and conditions thereof are competitive with the prices and terms of goods, supplies, rights and services of like quality available from non-Affiliated third parties in an arm's length transaction, and (iii) in respect of such shared arrangements, the allocations of costs with respect thereto are fair and reasonable and, if applicable, consistent with the methodology used by the Consultant and its subsidiaries and Affiliates to allocate similar costs among the Consultant Related Facilities.
- 6.16 Authority; Binding Agreement. Each Party represents and warrants that (a) it is a corporation, limited liability company, partnership or other entity, as applicable, duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, (b) it has the requisite power and authority to execute, deliver and perform this Agreement, (c) the execution, delivery and performance by it of this Agreement has been duly and validly authorized by all necessary actions on its part, and (d) this Agreement constitutes the legal, valid and binding obligations of such Party, enforceable against such Party in accordance with its terms.
- 6.17 Further Assurances. Each Party hereto shall further execute and deliver all such other appropriate supplemental agreements and other instruments and take such other action as may be necessary to make this Agreement fully and legally effective, binding and enforceable as between the Parties and as against third parties, or as the other Parties may reasonably request.

[SIGNATURE PAGES FOLLOW]

written.	IN WITNESS WHEREOF, each of the undersigned has executed this Theater Services Agreement as of the date first above
	THE COMPANY:
	FE CONCEPTS, LLC
	By: Name:
	Title:
	THE CONSULTANT:
	CNMK TEXAS PROPERTIES, LLC
	By: Name:
	Title:
	D-13

## **EXHIBIT E**

## FORM OF CAPITAL NOTE

\$[]	Dallas, Texas	[], 201
is 12400 Coit Road, Suite 800,	EIVED, the undersigned [Joint Venture], LLC, a Texas limited liability Comparaballas, Texas 75251, promises to pay to the order of [] (" <i>Lender</i> "), whos and []/100 (\$[]), with interest as specified herein, at [], or at such ag.	se address is [], on demand the
	his "Note") evidences a loan by Lender made to Borrower pursuant to Section wer and is subject to the following additional provisions, terms and conditions:	
demand or it no demand is made any part of the unpaid principal	e entire unpaid principal balance hereof plus all accrued but unpaid interest the prior to [], 20 on [], 20 Borrower may at any time a labalance of this Note without premium or penalty. All principal payments t being repaid to the date of payment.	and from time to time prepay all or
2. Interest.		
maturity (whether by accelerat maximum nonusurious interest	orrower agrees to pay interest in respect of the unpaid principal amount of the ion or otherwise) at a rate per annum equal to the lesser of (i) []% per annum rate permitted by applicable law (the " <i>Maximum Rate</i> "). Overdue principal such amounts are overdue at a rate per annum equal to the lesser of (i) the Base	num (the "Base Rate") and (ii) the shall bear interest, before and after
, ,	terest on the principal of this Note shall accrue from and including the datalated on the basis of a 360-day year consisting of twelve 30-day months.	e hereof and shall be payable on
noon (Dallas, Texas time) on the funds. Whenever any payment date thereof shall be extended and be payable at the applical collection expenses, if any, the balance hereof. As used in the	payments under this Note shall be made without defense, setoff or counterclared date when due and shall be made in lawful money of the United States of at to be made under this Note shall be stated to be or otherwise due on a day the other next succeeding Business Day, and, with respect to payments of principal pole rate during such extension. Each payment received by Lender shall be noted to the payment of accrued but unpaid interest due hereunder, and then to the sex Note, "Business Day" shall mean any day of the week except Saturday, Sex as are authorized or required by law to close.	America in immediately available hat is not a Business Day, the due al, interest thereon shall accrue on applied first to late charges and a reduction of the unpaid principal
	ъ.	

- 3. **Remedies**. Lender may at any time declare the entire unpaid principal balance of this Note, together with all accrued but unpaid interest thereon, immediately due and payable. Upon such demand, the entire unpaid principal balance of this Note, together with all accrued but unpaid interest thereon, shall automatically and immediately become due and payable, and thereafter Lender may proceed to enforce payment of the same and to exercise any and all of the rights and remedies afforded herein as well as all other rights and remedies possessed by Lender by law or otherwise. The entire unpaid principal balance of, and all accrued and unpaid interest on, this Note shall immediately be due and payable upon the occurrence of Borrower's (a) commencement of a voluntary case under Title 11 of the United States Code, or (b) filing an answer or other pleading admitting or failing to deny the material allegations of a petition filed against Borrower commencing an involuntary case under said Title 11 or failure to controvert timely the material allegations of such petition.
- 4. **No Waiver by Lender**. No failure or delay on the part of Lender in exercising any right, power or privilege hereunder and no course of dealing between Borrower and Lender shall operate as a waiver thereof, nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.
- 5. Waivers. Except as otherwise expressly provided for herein, the makers, signers, sureties, guarantors and endorsers of this Note severally waive notice of acceptance of this Note, notice of extension of credit, demand, presentment, notice of presentment, notice of dishonor, notice of intent to demand or accelerate payment hereof, notice of demand, notice of acceleration, diligence in collecting, grace, notice and protest, and agree to one or more renewals or extensions for any period or periods of time, partial payments and releases or substitutions of security, in whole or in part, with or without notice, before or after maturity. If this Note shall be collected by legal proceedings or through a bankruptcy court, or shall be placed in the hands of an attorney for collection after default or maturity, Borrower agrees to pay all costs of collection, including reasonable attorneys' fees.
- 6. **Limitation on Interest**. Notwithstanding any other provision of this Note, interest on the indebtedness evidenced by this Note is expressly limited so that in no contingency or event whatsoever, whether by acceleration of the maturity of this Note or otherwise, shall the interest contracted for, charged or received by Lender exceed the maximum amount permissible under applicable law. If from any circumstances whatsoever fulfillment of any provisions of this Note or of any other document evidencing, securing or pertaining to the indebtedness evidenced hereby, at the time performance of such provision shall be due, shall involve transcending the limit of validity prescribed by law, then, *ipso facto*, the obligation to be fulfilled shall be reduced to the limit of such validity, and if from any such circumstances Lender shall ever receive anything of value as interest or deemed interest by applicable law under this Note or any other document evidencing, securing or pertaining to the indebtedness evidenced hereby or otherwise an amount that would exceed the highest lawful rate, such amount that would be excessive interest shall be applied to the reduction of the principal amount owing under this Note or on account of any other indebtedness of Borrower to Lender, and not to the payment of interest, or if such excessive interest exceeds the unpaid balance of principal of this Note and such other indebtedness, such excess shall be refunded to Borrower. In determining whether or not the interest paid or payable with respect to any indebtedness of Borrower to Lender, under any

specific contingency, exceeds the highest lawful rate, Borrower and Lender shall, to the maximum extent permitted by applicable law, (a) characterize any non-principal payment as an expense, fee or premium rather than as interest, (b) exclude voluntary prepayments and the effects thereof, (c) amortize, prorate, allocate and spread the total amount of interest throughout the full term of such indebtedness so that the actual rate of interest on account of such indebtedness does not exceed the maximum amount permitted by applicable law, and/or (d) allocate interest between portions of such indebtedness, to the end that no such portion shall bear interest at a rate greater than that permitted by applicable law. The terms and provisions of this paragraph shall control and supersede every other conflicting provision of this Note and all other agreements between Borrower and Lender.

- 7. **Negative Covenants**. So long as this note or any portion hereof remains outstanding and unpaid, including principal and accrued unpaid interest hereon, Borrower shall not, without prior approval of Lender:
- (a) incur any indebtedness or enter into or consummate any other financing transaction or capital lease or create any lien against the assets of the Company, in any case senior to or on party with the indebtedness evidenced by this Note; or
  - (b) make any distributions to the members or equity holder of Borrower.
- 8. **Choice of Law; Venue**. This Note and the validity and enforceability hereof shall be governed by and construed in accordance with the laws of the State of Texas other than the conflicts of laws rules thereof. Exclusive venue for any action under this Note shall be the state and federal courts located in Collin County or Dallas County, Texas.

IN WITNESS WHEREOF, the undersigned has executed this Note as of the date first above written.

	CEPTS, LI mited liabil	any		
By:				
By: Name:				
Title:				
:-3				

## **EXHIBIT F**

## ASSIGNMENT OF MEMBERSHIP UNITS

FOR	VALUE	RECEIVED,	the	undersigned	does	hereby	sell,	assign	and	transfer	unto
							_ all of	the Membe	rship Un	its in FE Co	ncepts,
LLC standing in	the undersign										
			orney to 1	transfer the said	d Member	rship Units	on the b	ooks of sai	d limited	d liability co	mpany
with full power of	substitution	in the premises.									
Dated:											
					OI DED	OE MEMD	EDCIII	LIMITO			
				HOLDER OF MEMBERSHIP UNITS							
				D							
				B <sub>i</sub>	y. ame:						
					itle:						
				11	ille.						
				_							
				Sı	ouse						

#### MANAGEMENT SERVICES AGREEMENT

THIS MANAGEMENT SERVICES AGREEMENT (this "Agreement") is dated and made effective as of April 20, 2018 (the "Effective Date"), by and between FE Concepts, LLC, a Texas limited liability company (the "Company") and Cinema Operations, L.L.C., a Texas limited liability company (together with any of its subsidiaries from time to time providing Services hereunder, the "Manager"). The Manager and the Company are sometimes referred to in this Agreement, collectively, as the "Parties" and each individually as a "Party." Defined terms used in this Agreement but not otherwise defined herein have the meanings ascribed to them on Schedule I hereto.

#### **RECITALS**

- A. The Manager is an affiliate of AWSR Investments, LLC, a Texas limited liability company ("CB Entity").
- B. Cinemark USA, Inc., a Texas corporation ("Cinemark") is an affiliate of CNMK Texas Properties, LLC, a Texas limited liability company ("CNK Entity").
- C. CNK Entity and CB Entity are parties to that certain Limited Liability Company Agreement, dated as of April 20, 2018 (the "LLC Agreement"), pursuant to which CB Entity and CNK Entity have formed the Company to own and operate a certain family entertainment facility or facilities branded Strike + Reel incorporating games, rides, bowling, restaurants, bars and motion picture theater auditoriums (the "Facilities").
- D. The Manager has or will use commercially reasonable efforts to obtain the qualified, experienced and necessary personnel to manage each Facility and the Manager also has or will use commercially reasonable efforts to obtain the qualified personnel necessary to provide accounting and administrative services in connection with the operation of a Facility.
- E. The LLC Agreement provides in part that the Company shall engage the Manager to provide the Services pursuant to this Agreement.
- F. The Company desires to retain the Manager as the exclusive provider of the Services with respect to any Facility and the Manager is willing to provide such services, upon the terms and subject to the conditions set forth in this Agreement.

**NOW**, **THEREFORE**, in consideration of the mutual promises and covenants contained in this Agreement and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties, intending to be legally bound, agree as follows:

# ARTICLE I CONSTRUCTION AND DEVELOPMENT SERVICES

1.1 Provision of Construction and Development Services. Subject to Article III below, the Company hereby engages the Manager to provide the construction and development services described in Schedule II hereto (collectively, the "Development Services") in connection with the acquisition, construction and development of a Facility. Subject to Article III below, the Manager is hereby vested with full authority to take all such actions, do all such things and execute and enter into on behalf of the Company all such contracts, agreements, leases, licenses, instruments, commitments, undertakings and understandings (collectively, "Contracts") relating to such activities listed on Schedule II except as specified in Section 3.1.

## 1.2 <u>Standards for Provision of Development Services</u>.

- (a) The Manager shall devote such time and personnel as the Manager deems appropriate to provide the Development Services. The Manager shall not be required to spend its full time and attention to provide the Development Services.
- (b) The Company will provide such cooperation as is needed from time to time in connection with the provision of the Development Services hereunder. In connection with the provision of the Development Services hereunder, the Manager shall obtain and maintain at the Company's expense, for itself or on behalf of the Company, as the case may be, all licenses, permits and authorizations from any Governmental Authorities which are necessary for the construction and development of a Facility. The Company shall cooperate with the Manager in obtaining all consents, authorizations, approvals, permits and licenses necessary for the construction and development of a Facility. The Manager shall use commercially reasonable efforts to be at all times qualified and licensed to do business and be in good standing in all jurisdictions where such qualification or license is required to enable the Manager to provide the Development Services as contemplated by the terms of this Agreement.
- (c) The Manager shall conduct all activities and duties hereunder in good faith and shall use its commercially reasonable efforts to obtain terms for the supply of goods and services to a Facility and the Business comparable with the terms that the Manager and its subsidiaries and Affiliates are able to obtain for the Manager Related Facilities, provided, however, that the Manager is not required to purchase such goods and services on the same terms as Manager purchases similar goods and services for Manager Related Facilities. In connection with the construction and development of a Facility, the Manager may purchase necessary goods, supplies and services from or through the Manager or any of its Affiliates so long as such prices are comparable with the prices and terms of goods, supplies and services of like quality available from non-affiliated third parties in an arm's length transaction.

1.3 <u>Theater Services</u>. The Manager acknowledges that the Company has entered into the Theater Services Agreement with Cinemark pursuant to which Cinemark shall provide the Theater Services, which includes certain expertise regarding the construction and design of the motion picture theater auditoriums and the selection of certain technologies related thereto (the "*Theater Construction Expertise*"). The Theater Construction Expertise shall be provided exclusively by Cinemark and shall not be construed to constitute Development Services provided by the Manager. The Manager shall coordinate the provision of Development Services under this Agreement with the provision of the Theater Construction Expertise by Cinemark, including the implementation of any recommendations or plans resulting therefrom.

# ARTICLE II MANAGEMENT SERVICES

2.1 <u>Provision of Management Services</u>. Subject to <u>Article III</u> below, the Company hereby engages the Manager to manage a Facility and the business conducted at a Facility (the "*Business*") and to provide the Management Services in accordance with this Agreement, and the Manager hereby agrees to manage a Facility and the Business and to provide the Management Services in accordance with this Agreement. Subject to <u>Article III</u> below, the Manager is hereby vested with full authority and responsibility for the day-to-day management of a Facility and the Business and is empowered to take all such actions, do all such things and execute and enter into all such Contracts in this regard, on behalf of and in the name of the Company, as the Manager may, in its reasonable discretion, deem necessary or desirable for such purpose; subject to the provisions of <u>Article III</u>. For the avoidance of doubt, it is agreed that the Manager shall have the responsibility and authority to make any and all decisions with respect to the day-to-day, ordinary course of business operations of a Facility and the Business, subject to the provisions of <u>Article III</u>.

## 2.2 Standards for Provision of Management Services; Cooperation.

- (a) The Manager shall devote such time and personnel as the Manager deems appropriate to provide the Management Services that are comparable to those employed by the Manager in connection with the operation of the Manager Related Facilities. The Manager shall not be required to spend its full time and attention to provide the Management Services.
- (b) The Company will provide such cooperation as is needed from time to time in connection with the provision of the Management Services hereunder. In connection with the provision of the Management Services hereunder, the Manager shall use commercially reasonable efforts to obtain and maintain at the Company's expense, for itself or on behalf of the Company, as the case may be, all licenses, permits and authorizations from any Governmental Authorities which are necessary for the operation of the Business in the manner contemplated by this Agreement, including, but not limited to, licenses to serve food and alcoholic beverages. The Company shall cooperate with the Manager in obtaining all consents, authorizations, approvals, permits and licenses necessary to operate the Business. The Manager shall use commercially reasonable efforts to be qualified and licensed to do business and be in good standing in all jurisdictions where such qualification or license is required to provide Management Services as contemplated by the terms of this Agreement.

- (c) The Manager shall conduct all activities and duties hereunder in good faith and shall use commercially reasonable efforts to obtain terms for the supply of goods and services to and by the Facilities and the Business comparable with the terms that the Manager and its subsidiaries and Affiliates are able to obtain for the Manager Related Facilities. In the management of the Facilities, the Manager may purchase necessary goods, supplies and services from or through the Manager or any of its Affiliates so long as such prices are comparable with the prices and terms of goods, supplies and services of like quality available from non-affiliated third parties in an arm's length transaction.
- (d) As part of the Management Services, the Manager shall provide "point of sale" systems necessary to effectuate the retail transactions incident to the Business for the Facilities that are the same as the point of sale software systems used by the Manager in the operations of the Manager Related Facilities.
- (e) The Manager shall use commercially reasonable efforts to obtain all software, permits, licenses, sublicenses, authorizations, intellectual property and other intangibles and rights necessary to operate the Facilities and perform under this Agreement, including any third party software, applications and/or licenses. The Manager shall use commercially reasonable efforts to obtain any necessary consents and/or provide documentation to evidence the foregoing upon the Company's request.
- 2.3 <u>Compliance with Leases</u>. In providing the Management Services, to the extent a Facility is subject to a Lease (as defined below), the Manager shall use commercially reasonable efforts to comply in all material respects with, and not violate in any material respect, the terms of any real property lease associated with the Facilities (each such lease, a "*Lease*" and collectively, the "*Leases*").
- 2.4 <u>Compliance with Laws</u>. The Manager shall use commercially reasonable efforts to cause the Facilities to comply in all material respects with any Law affecting the Facilities and issued by any Governmental Authority having jurisdiction thereof. Further, the Manager shall not unreasonably refuse or delay compliance with any specific instructions that are given by the Company and that are reasonably necessary to cause the Facilities to comply with such Laws. The Company shall cooperate with the Manager in complying with such Laws, including paying all costs of such compliance.
- 2.5 <u>Theater Services.</u> The Manager acknowledges that the Company has entered into the Theater Services Agreement with Cinemark pursuant to which Cinemark shall provide the Theater Services. The Theater Services shall be provided exclusively by Cinemark and shall not be construed to constitute Management Services provided by the Manager. The Manager shall consult with Cinemark to receive Cinemark's recommendations with respect to the Theater Services to be provided by Cinemark under the Theater Services Agreement. The Manager shall coordinate providing its Services under this Agreement with the Theater Services to be provided by Cinemark under the Theater Services Agreement. The Manager shall cause the Company to pay all fees and expenses payable to Cinemark under the Theater Services Agreement.
- 2.6 <u>Internet Websites</u>. The Manager will create, or cause to be created, an Internet website for the Company listing each Facility on such website and providing on such website the ability for customers to purchase Facility admissions, prepaid cards and movie tickets for each Facility.

## ARTICLE III LIMITATIONS ON MANAGER AUTHORITY

- 3.1 <u>Actions Requiring Prior Written Consent of the Company</u>. Notwithstanding anything to the contrary contained in this Agreement, the Manager shall not, unless specifically authorized by the Company in writing, take any of the following actions with respect to any of the Company or any of its assets or properties, including the Facilities:
- (a) incur or commit to incur on behalf of the Company any capital expenditure (or series of related capital expenditures) that are, in the aggregate, more than five percent (5%) over amounts included in an Approved Budget; <u>provided</u>, <u>however</u>, that nothing in this Agreement shall be construed to prohibit or prevent the Manager from making emergency repairs necessary to prevent (i) damage to or destruction of the assets, properties or premises of the Facilities or (ii) protect the safety of the Facilities employees, customers or other invitees;
- (b) subject to Section 6.3, incur or commit to incur on behalf of the Company any expenditure(s) that would result in the aggregate of expenditures in all accounting categories, (other than Excluded Expenditures), exceeding an Approved Budget by more than five percent (5%) for each Facility for such entire Fiscal Year. Notwithstanding the preceding sentence, prior written consent will not be required if an expenditure for the Business exceeds the amount set forth in the Approved Budget during an interim reporting period if the Manager reasonably determines that such expenditure will not result in the aggregate of expenditures in all accounting categories for the Business exceeding those set forth in an Approved Budget for such entire Fiscal Year by more than five percent (5%) in the aggregate for a Facility;
- (c) enter into on behalf of the Company any transaction or Contract that is neither contemplated by this Agreement nor reasonably related to the Services being provided hereunder;
- (d) enter into on behalf of the Company any Contract in excess of \$50,000 per annum that cannot be terminated by the Company without premium or penalty on less than thirty (30) days prior written notice;
- (e) incur, guarantee, grant any security interest in respect of, or otherwise become liable for, any indebtedness for borrowed money, letters of credit, notes, bonds, mortgages or similar Contracts or lease obligations that are required to be treated as capitalized leases in accordance with generally accepted accounting principles in the United States ("GAAP");
  - (f) agree to subject the assets of the Company to any other lien or encumbrance, other than Permitted Liens; or
- (g) change any policy regarding the prices charged for tickets for any of the motion pictures exhibited in a Facility without the prior approval of the Board.

In the event that the authorization of the Company is required under this <u>Section 3.1</u> in respect of any action to be taken or not to be taken hereunder or any amount to be incurred or not to be incurred hereunder, the Manager shall request such authorization in writing as provided in <u>Section 12.5</u>.

3.2 <u>Company Approval</u>. For the purpose of this Agreement, any consent or approval required or permitted to be given by the Company shall require the approval of the Board (including at least one Board member not affiliated with the Manager) or, the approval of an executive officer of the Company not affiliated with the Manager to the extent that specific authority to grant such consent or approval has been expressly delegated to such officer by the Board, subject to any additional approval required by the Members in accordance with the LLC Agreement. Any notice by the Manager to the Company shall be given in accordance with <u>Section 12.5</u>.

## ARTICLE IV STAFFING SERVICES

- 4.1 <u>Provision of Staffing Services</u>. The Company hereby engages the Manager to provide the Staffing Services at a Facility in accordance with the terms of this Agreement, and the Manager hereby agrees to provide the Staffing Services at a Facility in accordance with the terms of this Agreement.
- 4.2 <u>Standards for the Provision of Staffing Services; Cooperation</u>. The Manager shall devote such time and personnel as the Manager deems appropriate to provide the Staffing Services. The Manager shall not be required to spend its full time and attention to provide the Staffing Services. Nothing in this Agreement shall prevent or restrict the Manager from terminating the employment or engagement of any of its employees, agency workers, consultants, contractors or other workers at any time or from taking any other actions with respect to its employees, agency workers, consultants, contractors or other workers, in its sole and absolute discretion.

## 4.3 Manager as Employer.

(a) The Parties hereto agree that the Manager will use commercially reasonable efforts to recruit, employ, train and supervise any and all employees necessary to enable the Manager to provide the Services hereunder, that such employees will at all times be employees of the Manager, a Manager Subsidiary or Manager Affiliate, and that the Manager will be responsible for (i) making all determinations related to such employees, including determinations with respect to wages, salaries, fringe benefits and other compensation, employment duration, the assignment of duties and the negotiation and settlement of labor disputes, (ii) making all payments in connection with the compensation of such employees, collecting and remitting any and all payroll taxes or other withholdings in connection therewith and filing any and all tax returns as are required with respect to such taxes, and (iii) procuring and maintaining adequate workers' compensation insurance as may be required by Law (it being understood that all expenses (including expenses relating to any of the foregoing) associated with the employment of Facility-level employees shall be Company Expenses).

- (b) In staffing each Facility, the Manager will use its commercially reasonable efforts to employ persons in such numbers and of such skill levels as are comparable in number and skill level to the persons the Manager employs in connection with the operation of the Manager Related Facilities.
- (c) The Manager acknowledges and agrees that, in any and all matters related to a Facility and the Business, it shall comply in all material respects with the Fair Labor Standards Act, the Occupational Safety and Health Act and any other law, rule or regulation relating to employment or human health and safety in connection with the Staffing Services. The Company acknowledges that employees at a Facility may be employed by a subsidiary or Affiliate of Manager.
- (d) The Parties hereto agree that the Company shall have no liability to any of the employees hired by the Manager, except for reimbursing the Manager for Company Expenses and otherwise as specifically set forth in this Agreement.

# ARTICLE V FINANCIAL INFORMATION, ACCESS AND CONSULTATION

- 5.1 Maintenance of Books and Records. The Manager shall keep full and materially accurate books of account and such other records in accordance with generally accepted accounting principles reflecting the Company operations of a Facility showing all revenue and expenses of the Company from a Facility during the term of this Agreement. The Manager's books and records for the Company will correspond with the Manager's fiscal year. Such Company records and accounts shall be prepared and maintained for each Facility in accordance with good business practices and in accordance with the method the Manager prepares such records and accounts for the Manager Related Facilities, and shall be sufficiently detailed to enable the statements of operations, balance sheets and statements of cash flows for the Company to be prepared from time to time for each Facility. For the avoidance of doubt, this Section 5.1 shall not require the Manager to provide any books and records of any Person other than the Company, including, but not limited to, any books and records of any Company Affiliates. Company Subsidiaries. Manager Affiliates, or Manager Subsidiaries.
- 5.2 Monthly and Quarterly Financial Reporting. The Manager shall within thirty (30) days after the end of each Fiscal Month ending after the Effective Date furnish to the Company a monthly Facility profit and loss statement of the Company in the form attached as Exhibit A (which monthly Facility profit and loss statement shall also be provided in a spreadsheet format), a trial balance reflecting all ledger account balances for the Company for each Facility, and a summary of all pro-rata Company costs and expenses allocated to a Facility by Manager for such period. Within forty-five (45) days after the end of each Fiscal Quarter ending after the Effective Date, unless such Fiscal Quarter ends on the last day of a Fiscal Year, in which case within sixty (60) days after the end of such Fiscal Quarter, furnish to the Company a consolidated unaudited balance sheet, income statement, statement of comprehensive income, statement of Member's equity and statement of cash flows for the Company, in each case in accordance with U.S. GAAP, which, for the avoidance of doubt, shall not require any footnotes. For the avoidance of doubt, this Section 5.2 shall not require the Manager to provide any financial records of any Person other than the Company, including, but not limited to, any financial records of any Company Affiliates, Company Subsidiaries, Manager Affiliates, or Manager Subsidiaries.

- 5.3 Inspection Rights. The representatives of the Company shall be entitled at all reasonable times and with reasonable prior notice to the Manager to inspect and obtain copies of all documents, records and accounts of the Company under the control of the Manager relating to a Facility and/or the Business and to discuss the financial statements and performance of a Facility with appropriate personnel of the Manager. The frequency, manner and duration of such inspections and discussions shall be conducted at times and dates mutually agreed by the Manager and the Company and without undue hindrance to the proper conduct of the operations of the Business or the activities of the Manager and its employees, agents and independent contractors. Any expenses incurred by the Parties in connection with any such inspections shall be the sole cost and expense of the Company. For the avoidance of doubt, this Section 5.3 shall only provide the Company with the right to inspect any documents, records and accounts of the Company Affiliates, Company Subsidiaries, Manager Affiliates, or Manager Subsidiaries.
- 5.4 Year-End Financials . The Manager shall cause the Company's accountants to prepare unaudited full year end financial statements of the Company for any such Fiscal Year (i.e., balance sheets, statements of income, statements of comprehensive income, statements of cash flows and statements of changes in Member's equity), which, for the avoidance of doubt, shall not require any footnotes, for presentation to the Company within sixty (60) days after the end of each Fiscal Year. The Company will also provide full year end audited financial statements (with related footnotes) of the Company within 180 days after the end of each Fiscal Year. In connection with the preparation of any such Company financial statement or any related audit or review thereof, the Manager shall make the Company books and records of a Facility available for such purposes in a timely manner. Any such financial statement preparation, audit or review shall be at the sole cost and expense of the Company. The costs and expenses associated with the full year end audited financial statements shall be billed directly to the Company. For the avoidance of doubt, this Section 5.4 shall only require the preparation of unaudited and audited financial statements for the Company, and not for any other Person, including, but not limited to any Company Affiliates, Company Subsidiaries, Manager Affiliates, or Manager Subsidiaries.
- 5.5 <u>Notice of Claims, Proceedings and Material Damage</u>. The Manager shall notify the Company in writing in a prompt and timely manner of receipt by the Manager of any written threat of, or, upon becoming aware thereof, the actual commencement or resolution of, any claims or Proceedings of the Company involving a Facility, or of any material damage, or material repair needs, of a Facility that the Company would be liable for.

## ARTICLE VI BUDGETS

- 6.1 <u>Draft Annual Budgets</u>. Prior to the Effective Date, the Manager and the Company shall agree upon an initial Annual Budget for the remainder of the Fiscal Year following the Effective Date in the form attached hereto as <u>Exhibit B</u>. With respect to each Fiscal Year thereafter during the term of this Agreement, the Manager shall prepare and provide to the Company, no later than the fifteenth (15th) day of December, a proposed draft budget for such Fiscal Year relating to each Facility in the form attached hereto as <u>Exhibit B</u>. The Manager shall consult with Cinemark when preparing the draft budget for line items relating to the motion picture theater auditoriums, including film rentals, in-theatre advertising and projector repair and maintenance.
- 6.2 Approved Annual Budgets. The Company shall consider and comment upon the proposed draft budget prepared by the Manager with respect to each Facility pursuant to Section 6.1, and the Manager shall from time to time amend and revise such draft budget in accordance with any reasonable written requests of the Company mutually agreed to by the Company and the Manager (each such budget including the initial Annual Budget referred to in Section 6.1, as so amended from time to time, an "Approved Budget"). If the Manager submits a proposed Annual Budget or an amendment to the Annual Budget for the Company's approval, and the Company does not request any changes or make any comments with respect to such Annual Budget, within thirty (30) days of submission, then such proposed Annual Budget shall be deemed the Approved Budget. Until such time as the proposed budget is approved by the Company, the last Approved Budget shall be the Approved Budget for such year except that (i) any portion of the proposed budget which is specifically approved or not in dispute shall be deemed approved and effective and (ii) during any interim period the Manager may reasonably exceed the Approved Budget for the prior Fiscal Year for (a) taxes, utility charges and other items not within the Manager's reasonable control and (b) increases in contract services, personnel and other costs to the extent required to maintain the same level of services for the attendance volumes anticipated at a Facility provided during the previous Fiscal Year if the costs of such items has increased above the Approved Budget for the prior Fiscal Year. In the event the Parties are unable to agree on an Approved Budget for a particular Fiscal Year within ninety (90) days of the beginning of such Fiscal Year, Manager shall operate a Facility related to the disputed portion of the proposal based upon the last approved Annual Budget.
- 6.3 Operation in Accordance with Approved Budgets. The Parties acknowledge and agree that Facility operating results (e.g., Facility attendance, revenues, operating expense, net income and cash flows) are subject to significant variance within any budgeted Fiscal Year and that no draft budget or Approved Budget can be expected to accurately predict such variances. Consequently, the Parties understand that actual interim operating results may vary from the Approved Budget. The Manager shall have the authority to make those expenditures that have been approved in the Approved Budget and which do not exceed the Approved Budget by more than five percent (5%) in the aggregate with respect to any Facility; provided that, notwithstanding the foregoing, the Manager may reasonably exceed the Approved Budget and make additional expenditures as Company Expenses for (a) taxes, utility charges and other items not within the Manager's reasonable control, and (b) increases in contract services, personnel and other costs, including but not limited to film rentals, food and beverage costs, game and

retail, and salaries and wages (the expenses listed in (a) and (b), the "Excluded Expenditures"), to the extent required to maintain the same level of services for the attendance volumes anticipated at a Facility, if the costs of such items have increased above the Approved Budget. If a condition of an emergency nature should exist which precludes consultation with and approval by the Company and requires that immediate repairs be made for the preservation and protection of the Facility or to protect the safety of the Facility employees, customers or other invitees, however, the Manager shall be authorized to take all actions and to make the necessary expenditure to repair and correct such condition, regardless of whether any provision has been made in the Approved Budget for such expenditure. Cash Expenditures made by the Manager in connection with any such emergency expenditure shall be paid as a Company Expense. Notwithstanding the foregoing, the Manager shall use commercially reasonable efforts to minimize expenditures not in the Approved Budget regardless of the Company's pre-approval of such expenditure. The Manager shall use commercially reasonable efforts to ensure that the actual costs of maintaining and operating a Facility shall not exceed the Approved Budget by more than five percent (5%) with respect to any Facility. All expenses shall be charged to the proper account in accordance with the Approved Budget. Expenses may be reasonably classified or reclassified in excess of the annual budgeted amount for an accounting category, provided that the Manager shall use commercially reasonable efforts to ensure the aggregate of expenditures in all accounting categories (other than Excluded Categories) does not exceed the Approved Budget by more than five percent (5%). If the Manager determines that any interim variances are reflective of expected operating results for the remainder of the Fiscal Year and that such variances may result in the Approved Budget for a Facility for such entire Fiscal Year exceeding those set forth in an Approved Budget for such Facility by more than five percent (5%) for such entire Fiscal Year (other than Excluded Expenditures), including any contingency with respect thereto provided for in the Approved Budget for the entire Fiscal Year, then the Manager may notify the Company and request that the Approved Budget for the Fiscal Year be amended in accordance with Section 6.2. Notwithstanding anything to the contrary in this Agreement, the Company's sole remedy for any breach by the Manager of this Article VI shall be termination of the Agreement in accordance with Section 10.2.

# ARTICLE VII MANAGEMENT COMPENSATION AND PAYMENT OF COSTS AND EXPENSES

- 7.1 <u>Management Compensation</u>. The Company shall pay to the Manager the compensation set forth on <u>Exhibit E</u> as follows:
  - (a) Payments under Tier 1 shall be paid within thirty (30) days of the opening of a new Facility.
- (b) Payments under Tier 2, if any, shall be paid annually within fifteen (15) days after delivery by the Manager to the Company of the unaudited full year end financial statements.

- (c) Payments under Tier 3 shall be paid upon the occurrence of a Trigger Event concurrently with payment to the CB Entity for such Trigger Event. For purposes of payments under Tier 3, "*Trigger Event*" shall mean (i) a Company Sale Transaction (as such term is defined in the LLC Agreement), or (ii) a transfer under a Call Option Notice, a CB Offer Notice, or a Sale Election Notice (as each such term is defined in the LLC Agreement).
- 7.2 Company Expenses. The Company shall be responsible for the payment of all costs and expenses incurred in connection with the construction, development and operation of a Facility or the operation of the Business, whether such costs or expenses are incurred directly by the Company or incurred by the Manager or its Affiliates in connection with the provision of the Services hereunder, including, without limitation, (a) the Facility's pro-rata shares of any Facility-level costs and expenses incurred on behalf of a Facility as part of any arrangements entered into by the Manager on behalf of a Facility and Manager Related Facilities, (b) Cash Expenditures and (c) all expenses and liabilities relating to workers compensation insurance and workers compensation claims by Facility-level employees (all such costs and expenses for which the Company are so responsible being referred to, collectively, as the "Company Expenses"). The Manager shall have the right to utilize its corporate employees and assets in connection with the operation of a Facility as required under this Agreement. When doing so, the Manager shall be reimbursed for all non-Facility level costs and expenses incurred by the Manager or its Affiliates allocable to work done for the Company, as set forth on Exhibit F ("Manager Overhead Costs"), including the salaries, benefits, bonuses and other compensation of home-office personnel who render Services and all office and overhead expenses that are incurred other than at the Facility -level. Without limiting the generality of the foregoing, Company Expenses shall include, without limitation, (i) Cash Expenditures and all of the costs and expenses of the type described on Schedule IV hereto, (ii) any costs and expenses in connection with any third party consents, permits or licenses relating to any Facility that are related to the Services, and (iii) any capital expenditures with respect to a Facility but only to the extent any such costs have been approved in advance and in writing by the Board. To the extent that the Manager corporate employees are required to, or requested to by the Company, to perform services beyond those contemplated herein (i.e., assisting with tax audits, managing litigation, etc.), then the Company and the Manager shall meet and consult in good faith to determine a fair and reasonable expense allocation for such services which will be fully reimbursable as Company Expenses. Amounts due to the Manager under this Section 7.2 shall be paid within thirty (30) days of receipt by the Company of an invoice for such amounts.

## 7.3 Company Account; Payment of Company Expenses; Quarterly Statements.

(a) All monies advanced by the Company to fund Company Expenses and all cash from operations of each Facility shall be deposited in separate bank accounts of the Company (the "Company Account") over which the Manager, as agent for the Company, shall have joint access and control with the Company. The Manager is hereby authorized to pay from the Company Account, on behalf of the Company, all Company Expenses and to use funds in the Company Account to reimburse the Manager for any Company Expenses previously paid by the Manager. Checks or other documents of withdrawal drawn upon the Company Account by the Manager shall be signed by representatives of the Manager as agent for the Company. The Parties hereby acknowledge and agree that the Manager will manage the finances of the Business on a combined Facility basis and shall be entitled to use cash flow generated from one Facility to fund operating expenses and losses of any other Facility.

(b) Concurrently with the delivery of the quarterly financial statements required by Section 5.2, the Manager shall issue to the Company a statement (the "Quarterly Statement") setting forth in reasonable detail all payments made from the Company Account in respect of Company Expenses with respect to each Facility and on a combined basis during such Fiscal Quarter. In the event that the Company reasonably and in good faith questions or disputes any element of any Quarterly Statement, written notice thereof shall be provided by the Company to the Manager within thirty (30) days of the receipt of the Quarterly Statement by the Company, and the Parties shall thereafter use commercially reasonable efforts to resolve to their mutual satisfaction any such questions and disputes during the twenty (20) days following delivery of such written notice by the Company. In the event the Parties are unable to resolve any such questions or disputes, the unresolved matters may be submitted to arbitration pursuant to Section 12.14.

# ARTICLE VIII INTELLECTUAL PROPERTY

- 8.1 Reservation of Rights. Any Facility subject to this Agreement shall be operated, marketed and branded as a "Strike + Reel" Facility at all times while the Services are being provided under this Agreement with respect to such Facility. The Manager confirms and acknowledges that the Company is and shall be at all times, the exclusive owner of all rights, title and interest in and to the Company Trademarks, including the goodwill associated therewith, (ii) the Manager shall not contest or challenge the Company's rights in such Company Trademarks at any time, (iii) nothing herein shall restrict the use of the Company Trademarks by the Company and the Company Subsidiaries or licensees at any time or in any manner, (iv) the Manager shall not acquire any ownership right in the Company Trademarks by virtue of said marks' use pursuant to this Agreement, and (v) any goodwill derived from the use of the Company Trademarks in connection with a Facility shall inure to the exclusive benefit of the Company.
- 8.2 Quality Control. The Manager acknowledges and agrees that: (i) all use of the Company Trademarks under this Agreement shall be, at all times, subject to the Company's reasonable quality control and (ii) to protect the valuable goodwill associated with the Company Trademarks, the Manager shall assure that the Company Trademarks are used solely in a manner consistent with the quality of the goods and services offered under the Company Trademarks. Upon receiving notice from the Company that the quality of the goods and services offered under the Company Trademarks does not meet the Company's acceptable quality standards, the Manager shall promptly take all remedial action to assure that goods and services offered under the Company Trademarks shall meet the quality standards that are acceptable to the Company's reasonable satisfaction.
- 8.3 <u>No Implied Rights.</u> Nothing in this Agreement provides or implies any right for the Manager to use the Company Trademarks except as otherwise provided for in this Article VIII.

# ARTICLE IX INDEMNITY

- 9.1 <u>Indemnification by the Company</u>. The Company shall indemnify and hold harmless the Manager and its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (and the members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons) (such indemnified Persons, the "*Manager Indemnified Persons*") from, against and in respect of any and all claims, Proceedings, obligations, liabilities, liens, encumbrances, losses, damages, assessments, fines, penalties, taxes, fees, costs (including costs of investigation, defense and enforcement of this Agreement, including attorneys' fees and disbursements), expenses or amounts paid in settlement (in each case, including reasonable attorneys' and experts fees and expenses) (collectively, "*Losses*"), in the case of clauses (b), (c), and (d) below, whether or not involving a Third Party Claim, to the extent such Losses are incurred or suffered by the Manager Indemnified Persons as a result of, arising out of or directly or indirectly relating to:
- (a) any claim or Proceeding made, initiated or threatened by any Person other than a Party (a "Third Party Claim") that relates to the provision of the Services under this Agreement, the management and operation of a Facility or the Business, or the consequences of the Manager's compliance with the terms of this Agreement or any directions provided by the Company relating to the operation of the Business, other than to the extent such Losses are caused by the bad faith, gross negligence or willful misconduct of the Manager or any of its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (or the shareholders, members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons);
  - (b) all Company Expenses;
  - (c) all Manager Overhead Costs;
  - (d) any breach of this Agreement by the Company;
  - (e) any potential regulatory issues related to the affiliation of the Company and the CNK Entity; and
- (f) any potential action brought by a shareholder of Cinemark and/or the CNK Entity arising from claims related to conflict of interest.
- 9.2 <u>Indemnification by the Manager</u>. The Manager shall indemnify and hold harmless the Company, the Company Subsidiaries and the members, managers, officers, directors, employees and other agents, representatives and Affiliates thereof (and the members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons) (such indemnified Persons, the "*Company Indemnified Persons*") from, against and in respect of any and all Losses to the extent such Losses are incurred or suffered by the Manager Indemnified Persons as a result of, arising out of or directly or indirectly relating to:
- (a) any Third Party Claim that relates to the provision of the Services under this Agreement or the management and operation of a Facility or the Business, in each case, to the extent such Losses are caused by the bad faith, gross negligence or willful misconduct of the Manager or any of its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (or the shareholders, members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons); or

- (b) any Third Party Claim brought by or on behalf of any of the Manager's employees, including Third Party Claims alleging age discrimination, sex discrimination or sexual harassment; <u>provided</u> that this clause (b) shall not be deemed to (i) provide for indemnification to the Company Indemnified Persons for any cost relating to employees of the Manager that would otherwise constitute a Company Expense (for example, compensation of Facility-level employees in the ordinary course of business or workers compensation -related expenses) or (ii) reduce or modify the obligations of the Manager or the Company pursuant to <u>Section 9.3</u>.
- (c) any Third Party Claim alleging Manager has violated the Fair Labor Standards Act, the Occupational Safety and Health Act with respect to the operations of a Facility or any other rule related to employment or human health and safety in connection with the Staffing Services and the Occupational Safety and Health Act related to the construction of a Facility only if such Occupational Safety and Health Act violation is caused by the bad faith, gross negligence or willful misconduct of the Manager.

#### 9.3 Insurance.

- (a) The Manager shall maintain all appropriate insurance coverage and fidelity bonds for a Facility, and the maintenance of such insurance coverage shall be consistent with Manager's practices for the Manager Related Facilities including the minimum coverages and amounts outlined in Exhibit C. All costs and expenses of maintaining insurance in connection with the Business and the operation of a Facility shall be Company Expenses.
- (b) The Manager shall notify the Company concurrently with notification to the Company's insurance carrier of any casualty, disaster, loss, damage or injury occurring at a Facility that is reported by the Manager to an insurance carrier, and otherwise promptly notify the Company of any actual claim or loss not occurring in the Company's ordinary course of business. The Manager shall not use a Facility, or permit the same to be used, for any purpose which will make void or voidable any of such insurance policies, and shall not keep or allow to be kept at a Facility any material, machinery, equipment, substance or other thing which may make void or voidable any of such insurance policies. The Manager shall, upon the request of the Company, assist the Company in any reasonable manner, as The Company may request, in the settlement of any claim under any of such insurance policies.
- (c) The amount of Losses for which indemnification is available under <u>Section 9.1</u> and <u>9.2</u> shall be calculated net of any amounts actually recovered by the Company Indemnified Persons or the Manager Indemnified Persons, as the case may be, under insurance policies with respect to such Losses. In the event that any indemnifying person makes any payment under <u>Section 9.1</u> or <u>9.2</u> in respect of any Losses, such indemnifying person shall be subrogated, to the extent of such payment, to the rights of such indemnified person against any insurer or other third Person with respect to such Losses.

# ARTICLE X TERM AND TERMINATION

10.1 <u>Term of Agreement</u>. This Agreement shall apply to each Facility acquired, owned or operated by the Company and shall commence on the Effective Date and shall continue until the tenth (10th) anniversary of the Effective Date, unless terminated earlier or extended pursuant to <u>Section 10.2</u> (the term of this Agreement, as so modified is referred to herein as the "*Term*").

#### 10.2 Modification of Term; Termination.

- (a) The Company shall have the right at any time to terminate this Agreement for Cause by giving fifteen (15) days advance notice in writing to the Manager; and shall have the right to terminate this Agreement without Cause, by giving two-hundred and seventy (270) days advance notice in writing to the Manager; or
- (b) The Manager may terminate this Agreement at any time upon the occurrence of a Company Default by giving fifteen (15) days advance notice in writing to the Company, and shall have the right to terminate this Agreement for any reason, with or without cause, after the first anniversary of the opening of the first Facility to the general public by giving ninety (90) days advance notice in writing to the Company; or
- (c) The Manager may terminate this Agreement at any time if (i) the CB Entity no longer holds Class B Units (as such term is defined in the LLC Agreement) in the Company, (ii) Lee Roy Mitchell no longer serves as an active member of the board of directors of Cinemark Holdings, Inc. for any reason, including without limitation retirement, death or disability, or (iii) breach by Cinemark of the Theater Services Agreement which remains uncured for thirty (30) days following such breach, by giving fifteen (15) days advance notice in writing to the Company; or
  - (d) This Agreement shall automatically terminate upon termination of the Theater Services Agreement.
- Services, Management Services and Staffing Services under this Agreement shall commence with respect to each Facility as may be specified in a notice by the Company with respect to each Facility. After commencement of the Services, the provision of the Services under this Agreement shall continue in respect to such Facilities, unless terminated earlier as expressly provided in Section 10.2, until the fifteenth (15th) calendar day after the Company shall have given the Manager written notice of termination with respect to such Service(s).

10.4 Effects of Termination . The termination of this Agreement shall not terminate the rights and obligations of the Parties under Article XI, which rights and obligations shall only terminate upon the termination of the Transition Service Periods. The termination of this Agreement in its entirety, howsoever arising shall be without prejudice to the rights and duties of the Parties accrued prior to termination or during any Transition Service Period. The Manager's access and control over any Company Account shall terminate upon termination of this Agreement in its entirety, or, if such access is required for the Manager to perform one or more of the Transition Services, upon the termination of the Transition Service Period applicable to such Transition Services. Notwithstanding the foregoing, the covenants, agreements, terms and conditions of this Agreement which expressly or impliedly have effect after termination (including, for the avoidance of doubt, those covenants, agreements, terms and conditions contained in Articles VII, VIII, X XI and XII of this Agreement and all indemnification and confidentiality provisions contained in this Agreement) shall continue to be enforceable notwithstanding termination, including the termination of any Transition Service Period. If this Agreement is terminated for any reason, any amounts accrued but not yet paid to the Manager hereunder shall be due and payable within ninety (90) days after the time of such termination, any amounts accrued but not yet paid to the Company hereunder shall be due and payable within ninety (90) days after the time of such termination and the Manager shall deliver to the Company the originals of all books, records, Contracts and all other documents, certificates, permits or instruments relating to a Facility, including documents evidencing termination of the Manager's right to take any action with respect to the Company Account. If possession by the Manager of any such books, records, Contracts, documents, certificates, permits or instruments are necessary for the performance of any Transition Service, Manager shall deliver the same to the Company upon termination of the Transition Service Period applicable to such Transition Services.

### ARTICLE XI TRANSITION SERVICES

- 11.1 <u>Transition Services</u>. If this Agreement terminates for any reason other than Cause, the Manager agrees to provide the services set forth on <u>Exhibit D</u>, and any other mutually agreed services determined to be necessary or appropriate to the Company in the same manner as the Services are required to be provided by this Agreement and to otherwise cooperate with the Company in transitioning the Services to a new manager(s) or in winding down the operations of the Company, as applicable, including but not limited to granting the Company access to and otherwise making available for copies any information related to the Company held by the Manager (collectively, the "*Transition Services*") for the respective periods set forth on <u>Exhibit D</u> (each such period, a "*Transition Service Period*"), commencing on the date on which the termination is effective ("*Transition Start Date*"). The Company agrees to pay the Manager as provided in this Agreement during the Transition Service Period.
- 11.2 <u>Software Use & License</u>. Subject to the terms and conditions of this Agreement, the Manager agrees to continue to provide any software and/or licenses that the Manager provided during the Term that are necessary for the performance of the Transition Services.

#### 11.3 Employees.

- (a) Offer of Employment. During a Transition Service Period, the Company, Cinemark and any of Cinemark's Affiliates (any of the foregoing that makes any such offer, the "New Facility Employer") may make offers of employment to any Facility-Level Employees. During the Transition Service Period, the Manager will make any employees of the Manager that work on an annual basis at the Manager's corporate offices and who dedicate at least eighty percent (80%) of their working time to the operation and management of a Facility (the "Facility Corporate Employees") reasonably available to the New Facility Employer for employment interviews. Upon the commencement of the Transition Service Period, the New Facility Employer will have the option, in its sole discretion, to attempt to hire any Facility Corporate Employee. To the extent such offers of employment are accepted, upon accepting such offers of employment, such Facility Corporate Employees and Facility-Level Employees' employment with the Manager shall terminate. The Manager will assist in the transition of such employees from the Manager to the New Facility Employer. The New Facility Employer may not make offers of employment to any employees who are not Facility-Level Employees or Facility Corporate Employees.
- (b) Severance Expenses. If, upon termination of this Agreement, (i) New Facility Employer does not offer employment to any Facility-Level Employee or Facility Corporate Employee, and (ii) as a direct result of the termination of this Agreement, the Manager terminates the employment of such employee, the Company will be solely responsible for, and shall reimburse the Manager for all costs incurred by the Manager associated with the failure of the New Facility Employer to provide employment to any Facility-Level Employee or Facility Corporate Employee, including, but not limited to, any severance or additional compensation. The Manager shall provide evidence of such severance costs and any other documentation or support reasonably requested by the Company prior to any such payment by the Company.
- 11.4 Financial and Tax Cooperation and Exchange of Information. The Manager shall provide and transition to the Company all financial information relating to a Facility in a manner compatible with the Company's financial reporting system. The Manager and the Company shall provide each other with such assistance as reasonably may be requested by either of them with respect to a Facility in connection with (a) the preparation of any tax return, amended tax return or claim of refund or (b) any audit or other examination by any taxing authority, or any judicial or administrative proceedings relating to liability for taxes or tax reporting by the Company (including preparation of tax reporting information to the members of the Company). The Manager and the Company will make themselves (and their respective employees) available, on a mutually convenient basis, to provide explanations of any documents or information provided under this Section 11.4. Any information obtained under this Section 11.4 shall be kept confidential, except as may be otherwise necessary in connection with the filing of tax returns or claims for refund or in conducting an audit or other proceeding.

- 11.5 <u>Post-Closing Access</u>. Subject to the restrictions of any applicable legal requirement, the Manager, on the one hand, and the Company, on the other hand, shall, following the termination of the Services and upon reasonable notice, afford such other party or parties and their representatives reasonable access (including the right to make photocopies, at the sole cost and expense of the parties requesting such access), to the extent necessary in connection with the Manager's operation of a Facility prior to the Transition Start Date or the Company's operation of a Facility after the Transition Start Date, during normal business hours and in a manner so as not to interfere unreasonably with such other Party, to such other Party's records related solely to a Facility, including upon specific request, a complete copy of the books and records to the extent permitted by applicable legal requirements and to the extent such books and records are under such other Party's or its Affiliates' control.
- 11.6 <u>Further Assurances</u>. The Parties will cooperate reasonably with each other and with their respective representatives in connection with any steps required to be taken as part of their respective obligations under this <u>Article XI</u>, and will (a) furnish upon request to each other such further information; (b) execute and deliver to each other such other documents; and (c) do such other acts and things, including obtain any consents and approvals required (if any), all as the other Party may reasonably request for the purpose of carrying out the intent of this <u>Article XI</u> and the Transition Services.
- 11.7 No Third Party Beneficiaries. Nothing in this Article XI, express or implied, is intended to confer on any person other than the parties hereto or their respective successors and assigns any rights, remedies, obligations or liabilities under or by reason of this Agreement. Nothing in this Article XI shall create any third party beneficiary rights in any employee or any beneficiary or dependents thereof or in any person other than the Parties to this Agreement.

# ARTICLE XII MISCELLANEOUS

# 12.1 Exclusions and Limitations of Liability.

- (a) All warranties, representations, guarantees, conditions and terms, other than those expressly set out in this Agreement, whether express or implied by statute, common law, trade usage or otherwise and whether written or oral are hereby expressly excluded to the fullest extent permissible by Law.
- (b) None of the Parties shall in any circumstances be liable for any claim, whether arising in contract, tort or otherwise, for consequential, economic, special or other indirect loss of any other Party, including losses calculated by reference to lost profits, contracts, business, goodwill, income, production or accruals of any such other Party (it being understood that if in connection with any Proceeding a Party is required to pay damages to a third party based on consequential, economic, special or other indirect losses of such third party, such liability for such damages shall be deemed to be a direct loss of the Party that is required to pay such amounts to such third party).

- (c) The Company acknowledges and agrees that the Manager shall not have any liability to the Company for any Losses made, suffered or incurred by the Company as a result of (i) the Manager's complying with the terms of this Agreement or any directions or authorizations that are provided to the Manager by the Company as specifically contemplated by this Agreement or (ii) if the Manager requests in writing an authorization from the Company under Section 2.3 (together with a reasonably detailed written explanation of the reasons for such request) in order to take a specified action but the Company does not provide such requested authorization, the consequences of not taking the actions that were the subject matter of such authorization request. In addition, the Company acknowledges and agrees that, notwithstanding anything to the contrary contained in this Agreement, the Manager shall not be required to incur any cost or make any expenditure that is not contemplated by the Approved Budget then in effect and shall have no liability to the Company for any Losses made, suffered or incurred by the Company as a result of the Manager's not incurring any such cost or making any such expenditure or failing to take any related actions (including rendering the Services) as a result thereof.
- (d) The Manager, its directors, agents, officers, employees, subsidiaries and Affiliates, as agents of the Company, shall not be liable to the Company or to any other Person for any act or omission committed in the performance of this Agreement unless such act constitutes bad faith, gross negligence, fraud or willful and wanton misconduct. Notwithstanding any other provision in this Agreement, in no event shall the Company make any claims against the Manager or its Affiliates on account of any alleged errors of judgment made in good faith in the development or operation of a Facility in accordance with the terms of this Agreement.
- (e) The Parties accept that the limitations and exclusions set out in this <u>Section 12.1</u> are reasonable having regard to all the circumstances.
- 12.2 <u>Company Subsidiaries</u>. The Parties acknowledge that the Company may cause one or more Facilities to be owned by a Company Subsidiary. The Manager agrees that the Company may designate one or more Company Subsidiaries to receive the Services to be performed by the Manager with respect to any Facility and all references herein to the Company shall be construed to include any such Company Subsidiary. For the purpose of this Agreement, Company Account may include one or more accounts of a Company Subsidiary designated by the Company to be used with respect to the Services provided by Manager to such Company Subsidiary.
- 12.3 Succession and Assignment; No Third-Party Beneficiaries. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. No Party may assign, delegate, sub-contract or otherwise transfer either this Agreement or any of its rights, interests or obligations hereunder without the prior written approval of the Company (in the case of a transfer by the Manager) or the Manager (in the case of a transfer by the Company); provided, that the Manager shall be entitled to (a) designate one or more Affiliates of the Manager to perform its obligations hereunder and/or (b) sub-contract with third parties the performance of its duties under this Agreement, provided that no such designation or sub-contract arrangement shall relieve any Party of its obligations under this Agreement. Except as expressly provided herein, this Agreement is for the sole benefit of the

Parties and nothing in this Agreement (whether expressed or implied) will give or be construed to give any Person, other than the Parties, any legal or equitable rights in connection with this Agreement. Any purported assignment in breach of this <u>Section 12.3</u> shall be void and confer no rights on the purported assignee.

12.4 <u>Transaction Costs</u>. Except as otherwise specified herein, the Company shall bear the costs and expenses in connection with the preparation, negotiation, execution and performance of this Agreement.

#### 12.5 Notices and Communications.

(a) All notices, requests, demands, claims and other communications required or permitted to be delivered, given or otherwise provided under this Agreement must be in writing and must be either personally delivered, mailed by first class mail (postage prepaid and return receipt requested), sent by electronic mail with confirmation of transmission by the transmitting equipment or sent by reputable overnight courier service (charges prepaid) to the recipient at the address below indicated:

If to the Company or a Company Subsidiary:

FE Concepts, LLC 12400 Coit Road, Suite 800 Dallas, TX 75251

Attention: Gary Witherspoon E-mail: gw@cbcap.com

with a copy (which shall not constitute notice) to:

Cinemark USA, Inc. 3900 Dallas Parkway, Suite 500 Plano, Texas 75093

Attention: Michael Cavalier, General Counsel

E Mail: mcavalier@cinemark.com

If to the Manager, to it at:

Cinema Operations, L.L.C. 12400 Coit Road, Suite 800 Dallas, Texas 75251

Attention: W. Mark Moore E-Mail: mmoore@epgusa.com

with a copy (which shall not constitute notice) to:

Cinemark USA, Inc. 3900 Dallas Parkway, Suite 500 Plano, Texas 75093

Attention: Michael Cavalier, General Counsel

E Mail: mcavalier@cinemark.com

(b) Each of the Parties may specify an address or email address different than that set forth in <u>Section 12.5(a)</u> by giving notice in accordance with <u>Section 12.5(a)</u> to each of the other Parties.

12.6 <u>Severability</u>. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. In the event that any provision hereof would, under applicable Law, be invalid or unenforceable in any respect, each Party hereto intends that such provision will be construed by modifying or limiting it so as to be valid and enforceable to the maximum extent compatible with, and possible under, applicable Law.

#### 12.7 Amendments, Waivers and Remedies.

- (a) No amendment or waiver of any provision of this Agreement will be valid and binding unless it is in writing and signed, in the case of an amendment, by each of the Parties hereto, or in the case of a waiver, by the Party against whom the waiver is to be effective.
- (b) No waiver by any Party of any breach or violation of, or default under or inaccuracy in, any representation, warranty or covenant hereunder, whether intentional or not, will extend to any prior or subsequent breach or violation of, or default under or inaccuracy in, any such representation, warranty or covenant or affect in any way any rights arising by virtue of any such prior or subsequent occurrence. No delay or omission on the part of any Party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof.

#### 12.8 Confidentiality.

- (a) Subject to Section 12.8(b), the Parties shall treat as strictly confidential all information received or obtained as a result of entering into or performing this Agreement which relates to: (i) the provisions of this Agreement, or any document entered into pursuant to this Agreement; (ii) the negotiations relating to this Agreement; (iii) the provision of the Services or the Business; and/or (iv) any of the other Parties, or their respective businesses, operations, assets, liabilities, financial condition or results of operations.
- (b) The restrictions in <u>Section 12.8(a)</u> shall not apply to prevent the use or disclosure of any confidential information if the relevant use or disclosure: (i) is required by applicable Law or any Governmental Authority of competent jurisdiction, subject to the condition that the disclosing Party will provide notice of such requirement to the non-disclosing Parties prior to disclosing any confidential information; (ii) is made to any legal, financial or other adviser, auditor, financing source, shareholder or other Affiliate of the disclosing Party,

subject to the condition that the Party making the disclosure shall be responsible for ensuring that such Persons comply with this Section 12.8 as if they were parties to this Agreement; (iii) is made to the officers or employees of the disclosing Party or any Affiliate thereof who need to know the information for the purposes of the transactions effected or contemplated by this Agreement, subject to the condition that the Party making the disclosure shall be responsible for ensuring that such Persons comply with this Section 12.8 as if they were parties to this Agreement; (iv) is in or becomes part of the public domain other than through an action of any Party; or (v) is approved in writing in advance by the Company (in the case of a disclosure by the Manager) or the Manager (in the case of a disclosure by the Company), as the case may be.

- 12.9 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which together will constitute but one and the same instrument.
- 12.10 Entire Agreement. This Agreement constitutes the entire agreement among the Parties hereto with respect to the subject matter hereof and supersedes any and all prior discussions, negotiations, proposals, undertakings, understandings and agreements, whether written or oral, with respect to such subject matter.
- 12.11 No Partnership, Etc.. Nothing in this Agreement or any document referred to in it or any arrangement contemplated by it shall be construed as creating a joint venture or partnership between the Parties for any purpose whatsoever and no Party shall have the power or authority to bind any other Party or impose any obligations on it to the benefit of any third party.
- 12.12 Construction. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement. Except as otherwise explicitly specified to the contrary, (a) references to a Section, Article or Schedule mean a Section or Article of, or Schedule to, this Agreement, (b) the word "including" will be construed as "including without limitation," (c) references to a particular statute or regulation include all rules and regulations thereunder and any predecessor or successor statute, rules or regulation, in each case as amended or otherwise modified from time to time and any reference to a document is to that document as varied, supplemented or replaced from time to time by agreement between the parties thereto, (d) the headings contained in this Agreement are for convenience purposes only and will not in any way affect the meaning or interpretation of this Agreement, (e) words in the singular or plural form include the plural and singular form, respectively, (f) use of any gender includes the other genders, (g) any reference to writing shall include any modes of reproducing words in a legible and non -transitory form, (h) references to a particular Person include such Person's successors and assigns to the extent not prohibited by this Agreement, and (i) the Schedules and Recitals hereto form part of this Agreement and shall have effect as if set out in full in the body of this Agreement, and any reference to this Agreement includes the Schedules and Recitals hereto.

- 12.13 <u>Governing Law.</u> This Agreement, the rights of the Parties hereunder and all Proceedings arising in whole or in part under or in connection herewith, will be governed by and construed in accordance with the domestic substantive laws of the State of Texas, without giving effect to any choice or conflict of law provision or rule that would cause the application of the laws of any other jurisdiction.
- 12.14 Arbitration. Any claim, dispute or other matter in question between the Parties hereto arising out of or relating to this Agreement, the rights of the Parties hereunder or the breach hereof, shall be decided by arbitration in accordance with the rules of the American Arbitration Association in effect on the Effective Date before three arbitrators; one designated by the Company, one designated by the Manager and the third designated in accordance with the Rules of the American Arbitration Association. Any such arbitration shall be conducted in Dallas, Texas, unless the Parties mutually agree to another location. The arbitrators shall be qualified by education, training or experience as may be appropriate according to the nature of the claim, dispute or other matter in question. The foregoing agreement to arbitrate and any other agreement to arbitrate shall be specifically enforceable under the prevailing arbitration law. The award rendered by the arbitrators shall be final, and judgment may be entered upon it in accordance with applicable Law in any court having jurisdiction thereof. To the extent permitted by law, by agreeing to engage in the arbitration provided for in this Section 12.14, the Parties waive their right to appeal any decision made by the arbitrators. The demand for arbitration shall be made within a reasonable time after the claim, dispute or other matter in question has arisen; and in no event shall it be made after the date when institution of legal or equitable proceedings based on such claim, dispute or other matter in question would be barred by the applicable statute of limitations. All costs and expenses (including reasonable attorneys' fees and costs) incurred in connection with any such arbitration shall be borne in the manner which the arbitrators making the determination shall direct. Notwithstanding the provisions of this Section 12.14, either Party may seek appropriate injunctive relief for any threatened breach.
- 12.15 <u>Affiliate Transactions</u>. Subject to <u>Article III</u>, in connection with the provision of the Management Services under this Agreement, the Manager may purchase necessary goods, supplies, rights and services (a) from or through any of its Affiliates or (b) pursuant to arrangements Manager Related Facilities so long as (i) any such transaction, series of transactions or arrangement is disclosed to the Company and approved by the Board including at least one Board member not affiliated with the Manager, (ii) in respect of any such Affiliate transaction or series of transactions, the terms and conditions thereof are competitive with the prices and terms of goods, supplies, rights and services of like quality available from non -Affiliated third parties in an arm's length transaction, and (iii) in respect of such shared arrangements, the allocations of costs with respect thereto are fair and reasonable and, if applicable, consistent with the methodology used by the Manager and its subsidiaries and Affiliates to allocate similar costs among the Manager Related Facilities.
- 12.16 <u>Authority; Binding Agreement.</u> Each Party represents and warrants that (a) it is a corporation, limited liability company, partnership or other entity, as applicable, duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, (b) it has the requisite power and authority to execute, deliver and perform this Agreement, (c) the execution, delivery and performance by it of this Agreement has been duly and validly authorized by all necessary actions on its part, and (d) this Agreement constitutes the legal, valid and binding obligations of such Party, enforceable against such Party in accordance with its terms.

- 12.17 <u>Further Assurances</u>. Each Party hereto shall further execute and deliver all such other appropriate supplemental agreements and other instruments and take such other action as which are or may be necessary or appropriate to effectuate and carry out the provisions of this Agreement, may be necessary to make this Agreement fully and legally effective, binding and enforceable as between the Parties and as against third parties, or as the other Parties may reasonably request.
- 12.18 <u>Cooperation</u>. The Parties will, and will cause their respective Affiliates to, in good faith, diligently communicate, consult and cooperate with each other to facilitate each other's performance of their respective and joint responsibilities and duties under this Agreement, including, but not limited to, any licensing responsibilities and duties. The Parties will, and will cause their respective Affiliates to, deal with each other on all matters related to this Agreement and otherwise in good faith in an open manner as promptly as possible under the given circumstances. Wherever an approval or consent is required in this Agreement by any Party, such approval or consent shall not be unreasonably withheld, delayed or conditioned. Each Party shall act reasonably with respect to the Party's responsibilities and obligations under this Agreement.
- 12.19 Other Activities. The Company acknowledges that the Manager and its Affiliates are in the business of developing and operating entertainment facilities that may be competitive with a Facility and that the Manager and its Affiliates may in the future be presented with opportunities to develop new entertainment facilities and acquire existing entertainment facilities. Except as may be set forth herein, the Parties agree that nothing in this Agreement shall be construed to restrict the Manager from pursuing opportunities to develop new entertainment facilities and acquire existing entertainment facilities, and the pursuit of those ventures or activities by the Manager or its Affiliates shall not be deemed wrongful or improper, even to the extent they may be competitive with a Facility or might otherwise constitute an opportunity for, the Company. Without limiting the foregoing, the Manager and its Affiliates may pursue concepts, ventures and opportunities on behalf of themselves rather than on behalf of the Company and need not offer, and the Company renounces any interest or expectancy as to participate in, such concepts, ventures and opportunities to the Company. Notwithstanding the foregoing, for a period of five (5) years after the termination of this Agreement neither Manager nor its Affiliates shall own, operate or acquire entertainment facilities similar to a Facility within a eight (8) mile radius from any Facility owned by the Company.

[SIGNATURE PAGES FOLLOW]

**IN WITNESS WHEREOF**, each of the undersigned has executed this Management Services Agreement as of the date first above written.

# THE COMPANY:

# FE CONCEPTS, LLC

By: /s/ W. Mark Moore

Name: W. Mark Moore

Title: Manager

# THE MANAGER:

# CINEMA OPERATIONS, L.L.C.

By: /s/ W. Mark Moore

Name: W. Mark Moore

Title: Manager

Signature Page to Management Services Agreement

#### SCHEDULE I

#### **CERTAIN DEFINED TERMS**

- "Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under direct or indirect common control with such Person. For the purposes of this definition, "control" when used with respect to any specified Person means the power to direct or cause the direction of the management and policies of such Person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise, and the terms "controlling" and "controlled" have meanings correlative of the foregoing.
  - "Agreement" has the meaning set forth in the Preamble hereto.
  - "Ancillary Revenues" has the meaning set forth in Section 2.3.
  - "Approved Budget" has the meaning set forth in Section 6.2.
- "Authorized User(s)" shall mean any authorized agents, affiliates, subsidiaries, divisions or subcontractors of the Company who shall require access to or use of the Manager Software in connection with the operation or transition of a Facility.
  - "Board" means the Board of Managers of the Company.
  - "Business" has the meaning set forth in Section 2.1.
  - "Business Day" means a day (other than a Saturday or Sunday) on which banks are open for business in Dallas, Texas.
- "Cash Expenditures" means all cash expenditures made by Manager pursuant to or in accordance with this Agreement, or which the Company shall otherwise authorize or approve, during any applicable or pertinent period in connection with the construction, development or operation of a Facility or any part thereof, including, but not limited to, film rental payments, rents under Facility leases, payroll and payroll expenses of Facility-level personnel of whatsoever nature and taxes, concession expenses, advertising, supplies, utility charges, repairs, maintenance and replacements (whether deemed to be a capital item or expense for accounting purposes), and insurance premiums and all other expenses of operation of a Facility. Cash Expenditures shall also include (i) real estate or other taxes, assessments, levies or charges and (ii) debt service or any other sums or charges payable under any mortgage or deed of trust affecting a Facility or any part thereof.
- "Cause" means (i) the commission of fraud, embezzlement or theft by the Manager as determined by a court of competent jurisdiction; (ii) a breach by the Manager of this Agreement, if such breach is not cured within five (5) Business Days (for any monetary breach) or sixty (60) days (for any non-monetary breach that the Manager makes reasonable efforts to cure during such cure period) after the Company provides written notice of such breach to the Manager; (iii) the intentional wrongful damage by the Company to the business reputation of the Company or its subsidiaries in a manner that has a material adverse effect on the Company and its subsidiaries, all only if so determined by a court of competent jurisdiction or (iv) the occurrence of a Class B Default or a Class B Conversion under the LLC Agreement.

- "Cinemark" means Cinemark USA, Inc.
- "Company" has the meaning set forth in the Preamble.
- "Company Account" has the meaning set forth in Section 7.3(a).
- "Company Default" means a breach by the Company of this Agreement if such breach is not cured within five (5) Business Days (for any monetary breach) or sixty (60) days (for any non-monetary breach that the Company makes reasonable efforts to cure during such cure period) after the Manager provides written notice of such breach to the Company.
  - "Company Expenses" has the meaning set forth in Section 7.2.
  - "Company Indemnified Person" has the meaning set forth in Section 9.1.
- "Company Software" means that certain proprietary software owned by the Company and used by the Manager to operate a Facility as of the Transition Start Date; including all related documentation and enhancements to Company Software.
- "Company Subsidiary" means any corporation, limited liability company, limited partnership or other entity that is directly or indirectly wholly owned by the Company.
- "Company Trademarks" means the registered and unregistered names, logos, insignia, trademarks, trade dress and other intellectual property specified by the Company from time to time.
  - "Contracts" has the meaning set forth in Section 2.1.
- "Development Services" means all supervision and administrative services (other than the Management Services and the Staffing Services) necessary for the construction and development of a Facility as more fully described in Schedule II, including the provision of any and all main office employees as may be necessary for the supervision and implementation of the construction and development of a Facility.
- "Documentation" shall mean the user guides, manuals and associated documentation for use by the Company and Authorized Users in connection with the Manager Software.
  - "Effective Date" has the meaning set forth in the Preamble hereto.
  - "Facilities" has the meaning set forth in the Recitals hereto.
  - "Facility Corporate Employee" has the meaning set forth in Section 11.3.
- "Facility-Level Employees" means employees employed by the Manager who work at a Facility (including Facility managers, assistant managers, management trainees and other employees).

Schedule I - 2

- "Fiscal Month" means each fiscal month within the Company's Fiscal Year, as determined by the Manager.
- "Fiscal Quarter" means each fiscal quarter which shall consist of three Fiscal Months.
- "Fiscal Year" means the Company's fiscal year which shall be a 52-53 week tax year ending on the Sunday closest to December 31 of each year.
  - "GAAP" has the meaning set forth in Section 3.1(e).
- "Governmental Authority" means any United States federal, state or local government or any foreign government, or any political subdivision thereof, any multinational organization or authority or any authority, agency or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, any court or tribunal (or any department, bureau or division thereof), or any arbitrator or arbitral body.
- "Law" means any United States federal, state or local, or any foreign, law, statute, standard, ordinance, code, rule, regulation, resolution or promulgation, any order, writ, judgment, injunction, decree, stipulation, ruling, determination or award entered by or with any Governmental Authority, or any license, franchise, permit or similar right granted under any of the foregoing, or any similar provision having the force or effect of law.
  - "Lease" and "Leases" have the meanings set forth in Section 2.4.
  - "LLC Agreement" has the meaning set forth in the Recitals hereto.
  - "Losses" has the meaning set forth in Section 9.1.
- "Management Services" means all management and operational services (other than the Development Services and Staffing Services) necessary for the day-to-day operation of the Business, including:
- (a) the day-to-day management and administrative operations of a Facility and the Business in the areas of concessions purchasing, facilities management, finance and accounting, human resources, marketing and public relations, operations, ticket sales, projection and sound equipment maintenance and property management, each as more fully described in <u>Schedule III</u>; and
- (b) the provision of any and all main office employees as may be necessary for the supervision and direction of the day-to-day management and administrative operations of a Facility and the Business.
  - "Manager" has the meaning set forth in the Preamble hereto.
  - "Manager Indemnified Persons" has the meaning set forth in Section 9.1.
  - "Manager Overhead Costs" has the meaning set forth in Section 7.2.

Schedule I - 3

- "Manager Related Facility" means an entertainment facility (other than a Facility) owned or operated by the Manager or its Affiliates.
- "Manager Subsidiary" means any corporation, limited liability company, limited partnership or other entity that is directly or indirectly wholly owned by the Manager.
  - "Negative Cash Flow Funding Amount" has the meaning set forth in Section 7.3.
  - "Negative Cash Flow Projection" has the meaning set forth in Section 7.3.
  - "New Facility Employer" has the meaning set forth in Section 11.3.
  - "Parties" has the meaning set forth in the Preamble hereto.
- "Permitted Liens" means (i) liens for taxes not yet due and payable; (ii) liens for taxes which are being contested in good faith; (iii) liens of any landlord, carrier, warehouseman, mechanic, materialman and any like liens arising in the ordinary course of business for sums that are not delinquent; (iv) easements, rights of way, zoning ordinances and other similar liens affecting real property; (v) liens and other encumbrances in favor of banks which are borrowings of the Company approved by the Board and (vi) liens and other encumbrances arising by operation of law.
- "*Person*" means any individual or corporation, association, partnership, limited liability company, joint venture, joint stock or other company, business trust, trust, organization, governmental or regulatory body or authority or other entity of any kind.
- "*Proceeding*" means any action, cause of action or suit (whether in contract or tort or otherwise), litigation (whether at law or in equity, whether civil or criminal), controversy, assessment, arbitration, investigation, hearing, charge, complaint, demand, notice or proceeding to, from, by or before any Governmental Authority.
  - "Quarterly Statement" has the meaning set forth in Section 7.3(b).
  - "Services" means the Development Services, the Management Services and the Staffing Services.
- "Software" shall mean that certain proprietary software owned by the Manager and used by the Manager to operate a Facility as of the Transition Start Date.
- "Staffing Services" means the recruitment, employment, training and supervision by the Manager of such employees at a Facility (including Facility managers, assistant managers, management trainees and other employees) as are reasonably necessary to enable the Manager to provide the Management Services, including responsibility for the oversight of all everyday human resources issues relating to a Facility, including issues with respect to compensation, fringe benefits, employment duration, the assignment of duties, negotiations and settlements with any labor disputes, procuring and maintaining adequate workers' compensation insurance as may be required by Law, collection and remitting of payroll taxes and the filing of any tax returns as are required with respect to such taxes.

- "Term" has the meaning set forth in Section 10.1.
- "Theater Construction Expertise" has the meaning set forth in Section 1.3.
- "Theater Services" means the consulting services related to auditorium construction, film booking and buying, training, theater operations and such other services as may be provided by Cinemark under the Theater Services Agreement.
- "Theater Services Agreement" means the Theater Services Agreement between the Company and Cinemark of even date herewith pursuant to which Cinemark has agreed to provide the Theater Services.
  - "Third Party Claim" has the meaning set forth in Section 9.1(a).
  - "Trigger Event" has the meaning set forth in Section 7.1(c).
  - "Transition Services" has the meaning set forth in Section 11.1(a).
  - "Transition Service Period" has the meaning set forth in Section 11.1(a).
  - "Transition Start Date" has the meaning set forth in Section 11.1(a).

Schedule I - 5

#### **SCHEDULE II**

#### DEVELOPMENT SERVICES

<u>Discipline</u> <u>Service</u>

Site Selection Identification of potential Facility locations, obtaining feasibility or demographic studies, financial

analysis of potential Facility sites, recommendations to the Board regarding site selection;

acquisition of real estate.

Planning Facility design, obtaining zoning, infrastructure plans, building permits, engaging architects and

other services required for construction planning, construction cost analysis, and Board

presentations related to construction planning, arranging construction and permanent financing for

Facility.

Construction Obtaining all construction plans and permits, entering into contracts on commercially reasonable

terms on behalf of the Company with architects, engineers, tradesmen and other independent contractors to perform services and supervising the administration and monitoring the performance

of all work to be performed and services to be rendered under all such contracts.

Development Taking all actions necessary to equip Facility including entering into purchase agreements or leases

for furniture, fixtures and equipment, supervision of interior design and build out.

Leasing If applicable, negotiation of any landlord lease for real property and any equipment leases for

equipment used at a Facility.

Detailed Services Schedule II-A lists in further detail specific matters to be included in the Development Services.

Schedule II - 1

#### **SCHEDULE II-A**

#### **REAL ESTATE**

- 1. Market and Site Sourcing:
- 2. Real Estate Contract, Contract Due Diligence (to include soils, title work, environmental), REA and Closing:
- 3. <u>Site Location Due Diligence:</u>
  - Civil Engineer
  - Environmental Engineer
  - Geotechnical Engineer
  - Architect
  - City incentive consultants
- 4. <u>City Entitlement Due Diligence:</u>
  - Zoning
  - Zoned for intended use, PD or zoning change required?
  - Parking Code
  - Site Planning and platting
  - Utilities at the site or need to be extend?
  - On site Detention Required?
  - Are Off Sites Required? Who handles? Who pays? Are City Incentives need and available?
  - Is the city pro-development of this site, neutral or anti-development?
  - Are the neighbors pro-development of this site, neutral or anti-development?
  - Impact Fees
  - Alcohol sales acceptable?
- 5. Excess pad sites/property:
  - Development issues
  - Marketing
  - Sales Contract

## CONSTRUCTION

- 1. <u>In development:</u>
  - Advise RE and senior management on construction issues
  - Secure due diligence reports (geotechnical, environmental, traffic...)
  - Coordinate schematic site and building design
  - Identify and engage consultants
  - Work thru entitlement process
  - Work thru civil engineering issues
  - Identify and design off site work
  - Direct development of building elevations

Schedule II-A - 1

## 2. <u>In design</u>:

- Review and development of site plan
- Establish the floor plans
- Review and development of auditorium section sight line studies
- Establish finish materials
- Review and development of millwork package
- Review of kitchen/equipment design
- Internal meetings/reviews with Ops, Theatre Tech, F&B
- Periodic drawing reviews and coordination meetings/calls as needed
- Development of Interior and Exterior Sign package
- Vendor coordination/review
- Coordinate permitting process
- Coordinate bidding process
- Analyze bids and award contract

## 3. <u>In construction</u>:

- Request and Authorize Purchase Orders for Owner Furnished Equipment
- Site visits during construction
- Review, assess and establish scheduling milestones
- Coordinate contractors and outside vendors (signage, millwork, HVAC equipment...)
- Coordinate final set-up of furniture
- Review construction deficiencies and monitor completion of punch-list corrections
- Coordinate system start-up and training of operations staff
- Advise Marketing on opening issues

# 4. <u>In operation, service and maintain:</u>

- Roof
- After Hours Lighting
- Kiosks (with Theatre Technology)
- Customer Parking Areas
- Doors
- Drainage System
- Driveways-S/C
- Electrical System
- Doors
- Fire panel /extinguisher
- Concession Equipment
- Recliners
- Floor Coverings
- Glass
- HVAC System
- Lighting interior & exterior
- Painting-Ext of Theatre

- Plumbing / toilet partitions
- Signage
- Sprinkler System (with Purchasing)
  Trash Removal/Compactor (with Purchasing)
  Windows/Frames/Glass
- Wiring
- Restoration services
- Burglar alarm service and repair
- Keying and coring of the building
- Bio Hazard removal
- CCTV Ceiling tile replacement
- Soda tower service
- CO2 line repair
- Digital menu board systems
- Auditorium LED signage
- Elevator/ Escalators
- Emergency lighting
- Energy management systems
  Millwork in the concession and restroom areas
- Screens/ masking/ drapery repair
- Pest control
- Refrigeration equipment
- Managers safe
  Ceiling tile replacement
  Speaker replacement
  Wall vinyl

- Wall carpet replacement

#### **SCHEDULE III**

#### MANAGEMENT SERVICES

Discipline **Service** Food, beverage, concessions and Purchasing, contracting for (on behalf of and in the name of the Company) and managing the purchase Purchasing and sale of all merchandise, materials, food, beverages, concessions, supplies and other products to be sold by or used in connection with the ordinary course operation of a Facility (including retail products) and setting prices with respect to any such products sold at a Facility. Service Contracting Entering into, on behalf of and in the name of the Company, such service, maintenance and other Contracts, or otherwise obtaining or providing such service or maintenance as shall be necessary or appropriate for the ordinary course operation and maintenance of a Facility, including the equipment and systems located in or servicing such Facilities, contracting for (on behalf of and in the name of the Company) the provision of utilities, elevator maintenance, telephone service, interior cleaning, window cleaning, landscape maintenance, rubbish removal, parking lot maintenance, fuel, heating and air conditioning maintenance, security, vermin and insect extermination and other necessary utilities and services, materials and supplies. Facilities Management Managing a Facility in the ordinary course, including the maintenance of all restaurant furniture, kitchen equipment, games, rides, bowling equipment, projection and sound equipment, mechanical and electrical equipment and plant, cleaning and security, time-management systems and fire and intruder alarm systems. Purchasing, or contracting for, all information systems (including point of sale systems and all hardware Information Systems and software) necessary, appropriate or convenient for the operation of the Business in a manner consistent with the operation of the Manager Related Facilities; including the installation and maintenance of all information systems.

Schedule III - 1

#### Finance and Accounting

Providing the following everyday treasury, finance and accounting services:

Rendering the necessary auditing, accounting and bookkeeping services generally required in the management of the affairs and operation of the Business, including:

- (a) maintenance of accounting records for a Facility and the Business;
- (b) payment as agent for the Company of amounts owing to third party suppliers of goods or services;
- (c) the preparation of relevant financial statements relating to a Facility provided for in this Agreement including arranging for the annual audit of the books and records of the Company;
- (d) the preparation and filing when due of all federal, state and local tax returns;
- (e) the maintenance of fixed asset ledgers for the Company;
- (f) cash management of the Business, including opening Company Accounts and liaison with banks at which Company Accounts are located;
- (g) preparation of budgets; and
- (h) preparation of financial forecasts.

Facility-Level Marketing; Advertising and Public Relations Day-to-day marketing, advertising and public relations with respect to a Facility, including (a) advertising allowances, sponsorship campaigns, radio and press listings, monitoring of existing on-screen advertising arrangements, but not including any marketing or public relations in connection with the launch and continued marketing of any new or different brand other than the "Strike + Reel" brand and (b) receiving, considering and handling the complaints of all guests and users of any of the services or facilities of a Facility.

Everyday Operational Decisions and Planning

Subject to <u>Article III</u>, everyday operational decisions arising during the ordinary course of business at a Facility, including policies with respect to show-times, hours of operation, ticket sales and ticketing issues and regional manager audits.

Collections and Payment; Bank Accounts

Collecting all revenues of the Business and paying (on behalf of the Company) all operating expenses relating to a Facility on behalf of and in the name of the Company.

Schedule III - 2

Property Management

Everyday property management of a Facility which may arise during the ordinary course of business, including (a) arranging for the making or installing, in the name of the Company, of all alterations, renovations, repairs, decorations, replacements, and equipment installations as may be reasonably necessary for the continued operation of a Facility in a manner consistent the standards required of and maintained by the Company and (b) all landlord and tenant issues, utilities and refuse removal.

Certain Taxes

Remitting on behalf of the Company when due to the proper Governmental Authorities all taxes on box office admissions that are not measured by net income and all sales taxes on other Facility revenues.

Permits

Obtaining and maintaining, on behalf of and in the name of the Company, as the case may be, all licenses, permits and authorizations from any Governmental Authorities that are necessary for the operation of a Facility in the manner required by this Agreement including all occupancy and food and beverage permits, alcoholic beverage licenses and other required permits (it being understood that the Company shall be required to execute and deliver any and all applications and other documents and to otherwise cooperate with the Manager in applying for, obtaining and maintaining all such licenses, permits and authorizations).

**Detailed Services** 

Schedule III-A lists in further detail specific matters to be included in the Management Services.

Schedule III - 3

#### **SCHEDULE III-A**

#### THEATRE TECHNOLOGY

- Appoint a person who has the responsibilities of a "Theater Technology Assistant (TTA)" or equivalent that would be responsible for, and the point of contact for the operation and maintenance of the digital equipment and the CSC / booth engineers
- Have the "TTA" or equivalent trained using the provided supplied training material and pass a digital cinema knowledge test
- Perform the programing and operate the digital projection and TMS equipment per Cinemark's digital cinema instruction manuals
- Perform all weekly, monthly, and quarterly maintenances on the digital projection and TMS equipment per Cinemark's maintenance manuals
- Perform troubleshooting and emergency support using the CSC as their only point of contact for digital cinema issues

#### ACCOUNTING/FINANCE/TAX

- General Ledger
- Fixed Assets
- AP
- AR
- Cash Management
- Income & Sales Tax Reporting
- Property Tax Reporting

#### <u>IT</u>

- POS System
- Web Site
- Internet & Mobile Ticketing
- Gift Card Sales

#### FOOD & BEVERAGE

- Procuring Beverages
- Procuring Candy & Popcorn
- Procuring Enhanced Food Items

### INSURANCE/RISK MANAGEMENT

- GL
- Property/Casualty
- Excess Coverage
- Unemployment Compensation

1.

## **HUMAN RESOURCES**

- Employees
- Health and Welfare Benefits
- Payroll
- Employment Taxes

# **MARKETING**

- 1. Digital Media
- 2. Marketing & Promotions / Corporate Branding:
- 3. Creative Services / Advertising:
  - Create advertising campaigns
  - Create content and manage digital menu boards and digital displays
  - Manage and create content for traditional media assets (i.e., Print, radio, billboards, TV, etc.)
- 4. Event Marketing / Sponsorship:
- 5. PR / Communications:
  - Manage press relations
- 6. Business Development / NCM Advertising:

# SCHEDULE IV SPECIFIED EXAMPLES OF COMPANY EXPENSES

#### COST OF SALES:

Film rental

Concessions

Food

Beverages

Merchandise

## **FACILITY OPERATING COSTS:**

#### Facility-level employee related:

Salaries and Wages

Commission & Bonus

Payroll Taxes (FICA, Medicare, SUI, FUTA)

Other Employee Expense

**Employee Training** 

Employee Benefits Outside Payroll Processing Service Fee

Any virtual print fees or alternative content fees paid pursuant to any digital projector agreements for the digital projectors at a

Facility

Benefits Administration Costs

## Maintenance, cleaning and supplies:

Armored Car Fees

Outside Janitorial Service

Exterminating

Film Delivery

Rubbish Removal

Security Guards

Bank Charges

Credit Card Fees

Supplies (cleaning, bulbs, computer, ticket stock, uniforms)

Repairs and Maintenance

Miscellaneous Costs

# Utilities:

Electric

Water and Sewer Fuel Oil and Gas Telephone and Internet

## <u>Insurance and professional fees</u>:

Insurance

Self-insured retention charges

## Advertising and promotion:

Directories

Advertising-Coop

Promotion Expense

Schedule IV - 1

# Other:

Postage and overnight delivery

Office Supplies

Dues and Subscriptions

Licenses and Permits

Over/Short Theater Cash

Property Taxes

**Equipment Rental** 

Travel Expense (airfare, car rental, mileage, lodging, meals)

Consulting Costs

Legal Expense

# Rent:

Rent to third parties

Common Area Maintenance

Percentage Rent

Schedule IV - 2

#### THEATER SERVICES AGREEMENT

THIS THEATER SERVICES AGREEMENT (this "Agreement") is dated and made effective as of April 20, 2018 (the "Effective Date"), by and between FE Concepts, LLC, a Texas limited liability company (the "Company") and CNMK Texas Properties, LLC, a Texas limited liability company ("Cinemark" and, together with any of its subsidiaries or parent from time to time providing Services hereunder, the "Consultant"). The Consultant and the Company are sometimes referred to in this Agreement, collectively, as the "Parties" and each individually as a "Party." Defined terms used in this Agreement but not otherwise defined herein have the meanings ascribed to them on Schedule I hereto.

#### RECITALS

- A. Consultant and AWSR Investments, LLC, a Texas limited liability company, are parties to that certain Limited Liability Company Agreement, dated as of April 20, 2018 (the "*LLC Agreement*"), pursuant to which CB Entity and Consultant have formed the Company to own and operate a certain family entertainment facilities, or facilities, incorporating games, rides, bowling, restaurants, bars and motion picture theater auditoriums (the "*Facilities*").
- B. The Consultant has the qualified, experienced and necessary personnel to consult with the Company regarding the construction and operation of the motion picture theater auditoriums (the "Auditoriums") to comprise a portion of a Facility.
- C. The LLC Agreement provides in part that the Company shall engage the Consultant to provide certain services pursuant to Article I of this Agreement (the "*Theater Services*").
- D. Cinema Management, L.L.C. (the "Manager") is a party to a Management Services Agreement with the Company to provide comprehensive management services regarding the construction, development, operation and staffing of any Facility (collectively, the "Management Services").
- E. The Company desires to retain the Consultant to provide the Theater Services with respect to any Facility and the Consultant is willing to provide such Theater Services, upon the terms and subject to the conditions set forth in this Agreement.
- **NOW**, **THEREFORE**, in consideration of the mutual promises and covenants contained in this Agreement and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties, intending to be legally bound, agree as follows:

### ARTICLE I THEATER SERVICES

- 1.1 <u>Auditorium Construction</u>. The Consultant agrees to consult with the Company and the Manager regarding the construction of the Auditoriums to be constructed at any Facility. The Consultant shall advise the Company and the Manager with respect to:
  - (a) recommendations regarding the number, size and design of the Auditoriums to be located in a Facility;
- ( b ) recommendations regarding architects, engineers, contractors, suppliers and vendors in connection with the construction of the Auditoriums;
  - (c) recommendations regarding furniture, fixtures and equipment to be used in the Auditoriums; and
- (d) consultations as may be reasonably requested by the Company or Manager during the construction of the Auditoriums at a Facility.
  - 1.2 <u>Theater Technologies</u>. The Consultant agrees to provide the following services relating to theater technologies:
- (a) select all digital projection, TMS, screen and sound equipment for purchase by the Company or the Manager on the Company's behalf;
  - (b) install, or supervise the installation of, all digital projection and TMS equipment;
  - (c) perform remote support of the digital cinema systems through CSC;
  - (d) manage the warranty and RMA of the digital cinema systems; and
  - (e) perform or supervise annual maintenance, upgrades and emergency support using the Consultant's booth engineers.
- 1.3 <u>Film Booking and Buying</u>. The Consultant agrees to book and license or cause to be licensed on the Company's behalf, films and other attractions for exhibition in the Auditoriums. The Consultant is hereby authorized by the Company to execute film license agreements on behalf of the Company for any Facility.
- 1.4 <u>Training</u>. The Consultant agrees to provide necessary and appropriate training both at a Facility and the Consultant Related Facilities to the Manager's facility-level personnel who will be assigned to conduct the exhibition of films in the Auditoriums, including ticketing personnel, certain concession operations and operators of film projection equipment and any other training necessary to fully plan, construct, equip, and successfully operate the Auditoriums. Training will be conducted at the Consultant's sole election at a Facility or at the Consultant's facilities.

1.5 <u>Tl</u>	heater Operations.	The Consultant wi	l consult	with the	Company	and the	Manager	regarding	the motion	picture	theater
operations including	g:										

- (a) ticket pricing policies;
- (b) film booking;
- (c) film settlement and settlement of any disputes with the distributors of content booked by the Consultant;
- (d) scheduling of motion picture exhibition times;
- (e) publication of show times;
- (f) management of studio relations and weekly trailer programming;
- (g) management of day-to-day studio relations, in-theater and on-line promotions;
- (h) management of trailer placements;
- (i) management of studio partnership marketing funds as applicable to the films shown at a Facility
- (j) arranging for on-screen advertising;
- (k) recommendations regarding ticketing systems and any third party ticketing vendors;
- (l) recommendations regarding personnel requirements in connection with motion picture exhibition in the Auditoriums;

and

- (m) recommendations regarding budget for line items relating to the motion picture theater auditoriums, including without limitation film rental expense, in theatre marketing and projector maintenance and repair.
- 1.6 <u>Exclusive Provider of Theater Services</u>. Unless otherwise expressly agreed by the Consultant with respect to any Facility, the Consultant shall be the exclusive provider of the Theater Services contemplated by this Agreement with respect to each Facility acquired, owned or operated by the Company.
- 1.7 <u>Company Approval</u>. For the purpose of this Agreement, any consent or approval required or permitted to be given by the Company regarding Theater Services shall require the approval of the Board (including either Lee Roy Mitchell or at least one Board member not affiliated with the Consultant) or, the approval of an executive officer of the Company not affiliated with the Consultant to the extent that specific authority to grant such consent or approval has been expressly delegated to such officer by the Board, subject to any additional approval required by the Members in accordance with the LLC Agreement. Any notice by the Consultant to the Company shall be given in accordance with Section 6.5.

### ARTICLE II ACKNOWLEDGEMENTS

- 2.1 <u>Delegation of Duties</u>. Notwithstanding anything to the contrary contained in this Agreement, the Consultant shall have the right to delegate or subcontract its duties under this Agreement to any Person; provided, however, that no such delegation or subcontracting shall relieve the Consultant of liability under this Agreement and further provided that the Consultant shall supervise the activities of such subcontractor, and provided, further that the Consulting Fee continues to be paid to the Consultant.
- 2.2 No Restrictions on the Consultant. The Consultant shall not be required to spend its full time and attention to provide any of the Services. The Consultant shall devote such time and personnel as the Consultant deems appropriate to provide the Services pursuant to this Agreement. The Company acknowledges that the Consultant and its Affiliates are in the business of developing and acquiring motion picture exhibition theaters that may be competitive with a Facility and that the Consultant and its Affiliates may in the future be presented with opportunities to develop new theaters and acquire existing theaters. Except as may be set forth herein, the Parties agree that nothing in this Agreement shall be construed to restrict the Consultant from pursuing opportunities to develop new theaters and acquire existing theaters, and the pursuit of those ventures or activities by the Consultant or its Affiliates shall not be deemed wrongful or improper, even to the extent they are competitive with, or might otherwise constitute an opportunity for, the Company. Without limiting the foregoing, the Consultant and its Affiliates may pursue concepts, ventures and opportunities on behalf of themselves rather than on behalf of the Company and need not offer, and the Company renounces any interest or expectancy as to participate in, such concepts, ventures and opportunities to the Company.
- 2.3 Film Settlement Policies. The Company acknowledges that subject to the next sentence, (i) with respect to booking services, numerous subjective and nonquantifiable matters of judgment go into the decision of which films to book and the terms of such booking and the Consultant, for a variety of reasons, may find it necessary to book films of certain distributors; (ii) separate geographic areas may be unique and involve factors relevant to policies employed in the negotiation and effectuation of film settlement ("Settlement Policies") that are not of equal importance to other geographical areas, (iii) Settlement Policies for one area may not necessarily be evaluated against or compared to the Settlement Policies the Consultant or its Affiliates employ or employed in other geographic areas or markets, and (iv) the Consultant will not be required to employ the same or similar Settlement Policies hereunder as it or its Affiliates follow in other geographic areas or markets or for other theaters owned, leased or managed by the Consultant in the same geographical area; provided that the Consultant shall conduct the Settlement Policies in a good and businesslike manner and in a manner consistent with Consultant Related Facilities. The Company confirms the Consultant has made no representation or covenant as to the Settlement Policies or booking policies it has or will have with respect to any theaters which are owned, leased or managed by the Consultant or its Affiliates or which are to be managed pursuant to the Management Agreement other than that Consultant will employ the same policies and practices in respect thereof on behalf of the Company as it does on behalf of the Consultant's theaters and that the Consultant will act in good faith in respect thereof.

- 2.4 <u>No Restriction on Payment of Consulting Fee</u>. The Company agrees not to directly or indirectly create or otherwise cause or suffer to exist or become effective any consensual encumbrance or the restriction on the ability of the Company to pay to the Consultant any fees or other amounts payable to the Consultant under this Agreement.
- 2.5 <u>Management Agreement</u>. The Consultant acknowledges that the Company has entered into the Management Agreement with the Manager pursuant to which the Manager shall provide the Management Services. The Company hereby authorizes the Consultant to coordinate providing the Theater Services with the Manager. The Consultant is authorized to rely on instructions from the Manager and advise the Manager's personnel regarding the various recommendations to be provided by the Consultant under this Agreement. The Company authorizes and directs the Manager to pay all Consulting Fees to the Consultant, and reimburse the Consultant for all Consultant Expenses incurred by the Consultant, as provided in this Agreement.

### ARTICLE III CONSULTING FEE

- 3 . 1 <u>Consulting Fees.</u> As compensation for its services under this Agreement, the Company will pay Consultant fees (the "*Consulting Fees*") as set forth on Schedule II hereto. Consultant shall invoice the Company monthly in arrears for Consulting Fees due with respect to the Theater Services performed by Consultant each month during the term of this Agreement. All invoices shall be due and payable within thirty days after receipt by the Company.
- 3.2 <u>Consulting Expenses.</u> The Company shall reimburse the Consultant for its out-of-pocket expenses incurred in providing the Theater Services, including, but not limited to travel expenses to any Facility or proposed Facility site, any Company Trademark related fees or expenses advanced by the Consultant in its reasonable discretion and such other expenses as may be reasonably incurred by the Consultant as set forth on <u>Schedule III</u> hereto ("*Consultant Expenses*").

# ARTICLE IV INDEMNITY

4.1 <u>Indemnification by the Company</u>. The Company shall indemnify and hold harmless the Consultant and its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (and the members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons) (such indemnified Persons, the "*Consultant Indemnified Persons*") from, against and in respect of any and all claims, Proceedings, obligations, liabilities, liens, encumbrances, losses, damages, assessments, fines, penalties, taxes, fees, costs (including costs of investigation, defense and enforcement of this Agreement, including attorneys' fees and disbursements), expenses or amounts paid in settlement (in each case, including reasonable attorneys' and experts fees and expenses) (collectively, "*Losses*"), in the case of clauses (b) and (c) below, whether or not involving a Third Party Claim, to the extent such Losses are incurred or suffered by the Consultant Indemnified Persons as a result of, arising out of or directly or indirectly relating to:

- (a) any claim or Proceeding made, initiated or threatened by any Person other than a Party (a "Third Party Claim") that relates to the provision of the Services under this Agreement, the management and operation of a Facility or the Business, or the consequences of the Consultant's compliance with the terms of this Agreement or any directions provided by the Company or the Manager relating to the operation of the Business, other than to the extent such Losses are caused by the bad faith, gross negligence or willful misconduct of the Consultant or any of its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (or the shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons);
  - (b) all Consultant Expenses;
  - (c) any breach of this Agreement by the Company.
- 4 . 2 <u>Indemnification by the Consultant</u>. The Consultant shall indemnify and hold harmless the Company, the Company Subsidiaries and the members, managers, officers, directors, employees and other agents, representatives and Affiliates thereof (and the members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons) (such indemnified Persons, the "*Company Indemnified Persons*") from, against and in respect of any and all Losses to the extent such Losses are incurred or suffered by the Company Indemnified Persons as a result of, arising out of or directly or indirectly relating to any Third Party Claim that relates to the provision of the Theater Services under this Agreement to the extent such Losses are caused by the bad faith, gross negligence or willful misconduct of the Consultant or any of its shareholders, managers, officers, directors, employees and other agents, representatives and Affiliates (or the shareholders, members, managers, officers, directors, employees and other agents, representatives and Affiliates of such Persons);
- 4.3 <u>Insurance Recoveries</u>. The amount of Losses for which indemnification is available under Section 4.1 and 4.2 shall be calculated net of any amounts actually recovered by the Company Indemnified Persons or the Consultant Indemnified Persons, as the case may be, under insurance policies with respect to such Losses. In the event that any indemnifying person makes any payment under Section 4.1 or 4.2 in respect of any Losses, such indemnifying person shall be subrogated, to the extent of such payment, to the rights of such indemnified person against any insurer or other third Person with respect to such Losses.

# ARTICLE V TERM AND TERMINATION

5.1 <u>Term of Agreement</u>. This Agreement shall commence on the Effective Date and shall continue until the tenth (10th) anniversary of the Effective Date, unless terminated earlier or extended pursuant to <u>Section 5.2</u> (the term of this Agreement, as so modified is referred to herein as the "*Term*").

### 5.2 Modification of Term; Termination.

- (a) The Consultant shall have the right at any time by giving (i) fifteen (15) days advance notice in writing to the Company to terminate this Agreement for non-payment under this Agreement if such non-payment is not resolved within fifteen (15) business days following written notice or (ii) after the first Facility has been in operation for a full year, two-hundred and seventy (270) days advance notice to terminate this Agreement, for any reason, with or without cause; or
- (b) The Company may terminate this Agreement at any time for Cause by giving fifteen (15) days advance notice in writing to the Consultant; provided that the Company shall not be permitted to terminate this Agreement without Cause without the prior written consent of the Consultant at any time prior to the expiration of the Term; or
- (c) This Agreement shall automatically terminate upon termination of the Management Agreement and termination of the Transition Services Period (if applicable).
- 5.3 <u>Commencement and Termination of Specified Theater Services</u>. The provision of the Theater Services under this Agreement shall commence with respect to each Facility on such date prior to the acquisition of the site for such Facility (or acquisition of any existing Facility) as the Company may specify. After commencement of the Theater Services, the provision of the Theater Services under this Agreement shall continue in respect of such Facilities, unless terminated earlier as expressly provided in <u>Section 5.2</u>.
- 5.4 Effects of Termination. The termination of this Agreement in its entirety (or the termination of any of the Theater Services with respect to the Facilities), howsoever arising, shall be without prejudice to the rights and duties of the Parties accrued prior to termination. The covenants, agreements, terms and conditions of this Agreement which expressly or impliedly have effect after termination shall continue to be enforceable notwithstanding termination including all indemnification and confidentiality provisions contained in this Agreement. If this Agreement (or the termination of any of the Theater Services with respect to the Facilities) is terminated for any reason, any amounts accrued but not yet paid to the Consultant hereunder shall be due and payable within thirty (30) days after the time of such termination.

## ARTICLE VI MISCELLANEOUS

## 6.1 Exclusions and Limitations of Liability.

(a) All warranties, representations, guarantees, conditions and terms, other than those expressly set out in this Agreement, whether express or implied by statute, common law, trade usage or otherwise and whether written or oral are hereby expressly excluded to the fullest extent permissible by Law.

- (b) None of the Parties shall in any circumstances be liable for any claim, whether arising in contract, tort or otherwise, for consequential, economic, special or other indirect loss of any other Party, including losses calculated by reference to lost profits, contracts, business, goodwill, income, production or accruals of any such other Party (it being understood that if in connection with any Proceeding a Party is required to pay damages to a third party based on consequential, economic, special or other indirect losses of such third party, such liability for such damages shall be deemed to be a direct loss of the Party that is required to pay such amounts to such third party).
- (c) The Company acknowledges and agrees that the Consultant shall not have any liability to the Company for any Losses made, suffered or incurred by the Company as a result of (i) the Consultant's complying with the terms of this Agreement or any directions or authorizations that are provided to the Consultant by the Company or the Manager as specifically contemplated by this Agreement or (ii) if the Consultant requests in writing an authorization from the Company or the Manager (together with a reasonably detailed written explanation of the reasons for such request) in order to take a specified action but the Company or the Manager does not provide such requested authorization, the consequences of not taking the actions that were the subject matter of such authorization request.
- (d) The Consultant, its directors, agents, officers, employees, subsidiaries and Affiliates, as agents of the Company, shall not be liable to the Company or to any other Person for any act or omission committed in the performance of this Agreement unless such act constitutes bad faith, gross negligence, fraud or willful and wanton misconduct. Notwithstanding any other provision in this Agreement, in no event shall the Company make any claims against the Consultant or its Affiliates on account of any alleged errors of judgment made in good faith in the development or operation of the Auditoriums in accordance with the terms of this Agreement.
- (e) The Parties accept that the limitations and exclusions set out in this <u>Section 6.1</u> are reasonable having regard to all the circumstances.
- 6.2 <u>Company Subsidiaries</u>. The Parties acknowledge that the Company may cause one or more Facilities to be owned by a Company Subsidiary. The Consultant agrees that the Company may designate one or more Company Subsidiaries to receive the Theater Services to be performed by the Consultant with respect to any Facility and all references herein to the Company shall be construed to include any such Company Subsidiary; provided however, that the Company and any Company Subsidiary receiving any Theater Services from the Consultant shall be jointly and severally liable for the Consulting Fee, Consultant Expenses and indemnification obligations payable under this Agreement.
- 6.3 <u>Succession and Assignment; No Third-Party Beneficiaries</u>. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. No Party may assign, delegate, sub-contract or otherwise transfer either this Agreement or any of its rights, interests or obligations hereunder without the prior written approval of the Company (in the case of a transfer by the Consultant) or the Consultant (in the case of a transfer by the Company); <u>provided</u>, that the Consultant shall be entitled to (a) designate one or more Affiliates of the Consultant to perform its obligations hereunder and/or (b) sub-contract with

third parties (other than a Class B Competing Business) the performance of its duties with respect Theater Services provided that no such designation or sub-contract arrangement shall relieve any Party of its obligations under this Agreement; provided, further, that the Consultant shall not be entitled to assign this Agreement to a Class B Competing Business without the Company's and Manager's prior written consent (except where such assignment is a result of the change in Control of Consultant). The Manager is intended to be a third party beneficiary of this Section 6.3. Except as expressly provided herein, this Agreement is for the sole benefit of the Parties and nothing in this Agreement (whether expressed or implied) will give or be construed to give any Person, other than the Parties, any legal or equitable rights in connection with this Agreement. Any purported assignment in breach of this Section 6.3 shall be void and confer no rights on the purported assignee.

6.4 <u>Transaction Costs</u>. Except as otherwise specified herein, the Company shall bear the costs and expenses in connection with the preparation, negotiation, execution and performance of this Agreement.

#### 6.5 Notices and Communications.

(a) All notices, requests, demands, claims and other communications required or permitted to be delivered, given or otherwise provided under this Agreement must be in writing and must be either personally delivered, mailed by first class mail (postage prepaid and return receipt requested), sent by electronic mail with confirmation of transmission by the transmitting equipment or sent by reputable overnight courier service (charges prepaid) to the recipient at the address below indicated:

If to the Company or a Company Subsidiary:

FE Concepts, LLC 12400 Coit Road, Suite 800 Dallas, TX 75251 Attention: Gary Witherspoon

E-mail: gw@cbcap.com

### If to the Consultant:

CNMK Texas Properties, LLC 3900 Dallas Parkway Plano, Texas 75093 Attention: Michael Cavalier, General Counsel

E Mail: mcavalier@cinemark.com

(b) Each of the Parties may specify an address or email address different than that set forth in <u>Section 6.5(a)</u> by giving notice in accordance with <u>Section 6.5 (a)</u> to the other Parties.

6.6 <u>Severability</u>. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. In the event that any provision hereof would, under applicable Law, be invalid or unenforceable in any respect, each Party hereto intends that such provision will be construed by modifying or limiting it so as to be valid and enforceable to the maximum extent compatible with, and possible under, applicable Law.

### 6.7 Amendments, Waivers and Remedies.

- (a) No amendment or waiver of any provision of this Agreement will be valid and binding unless it is in writing and signed, in the case of an amendment, by each of the Parties hereto, or in the case of a waiver, by the Party against whom the waiver is to be effective; provided, that Section 6.3 may not be amended without the consent of the Manager.
- (b) No waiver by any Party of any breach or violation of, or default under or inaccuracy in, any representation, warranty or covenant hereunder, whether intentional or not, will extend to any prior or subsequent breach or violation of, or default under or inaccuracy in, any such representation, warranty or covenant or affect in any way any rights arising by virtue of any such prior or subsequent occurrence. No delay or omission on the part of any Party in exercising any right, power or remedy under this Agreement will operate as a waiver thereof.

#### 6.8 Confidentiality.

- (a) Subject to Section 6.8(b), the Parties shall treat as strictly confidential all information received or obtained as a result of entering into or performing this Agreement which relates to: (i) the provisions of this Agreement, or any document entered into pursuant to this Agreement; (ii) the negotiations relating to this Agreement; (iii) the provision of the Theater Services or the Business; and/or (iv) any of the other Parties, or their respective businesses, operations, assets, liabilities, financial condition or results of operations.
- (b) The restrictions in Section 6.8(a) shall not apply to prevent the use or disclosure of any confidential information if the relevant use or disclosure: (i) is required by applicable Law or any Governmental Authority of competent jurisdiction, subject to the condition that the disclosing Party will provide notice of such requirement to the non-disclosing Parties prior to disclosing any confidential information; (ii) is made to any legal, financial or other adviser, auditor, financing source, shareholder or other Affiliate of the disclosing Party, subject to the condition that the Party making the disclosure shall be responsible for ensuring that such Persons comply with this Section 6.8 as if they were parties to this Agreement; (iii) is made to the officers or employees of the disclosing Party or any Affiliate thereof who need to know the information for the purposes of the transactions effected or contemplated by this Agreement, subject to the condition that the Party making the disclosure shall be responsible for ensuring that such Persons comply with this Section 6.8 as if they were parties to this Agreement; (iv) is in or becomes part of the public domain other than through an action of any Party; or (v) is approved in writing in advance by the Company (in the case of a disclosure by the Consultant) or the Consultant (in the case of a disclosure by the Company), as the case may be. For the avoidance of doubt, it is understood and agreed that the Consultant and/or its Affiliates are required to make

public disclosures of material facts and events under U.S. federal securities law and that nothing contained in this Agreement shall be construed to prevent the Consultant or such Affiliates from making such public disclosures (including, without limitation, public filings or disclosures relating to this Agreement, the Company, or the Business of the Company) as they determine, in their sole and absolute discretion, is required by applicable law or regulations.

- 6.9 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which will be deemed an original, but all of which together will constitute but one and the same instrument.
- 6.10 Entire Agreement. This Agreement constitutes the entire agreement among the Parties hereto with respect to the subject matter hereof and supersedes any and all prior discussions, negotiations, proposals, undertakings, understandings and agreements, whether written or oral, with respect to such subject matter.
- 6.11 No Partnership, Etc.. Nothing in this Agreement or any document referred to in it or any arrangement contemplated by it shall be construed as creating a joint venture or partnership between the Parties for any purpose whatsoever and no Party shall have the power or authority to bind any other Party or impose any obligations on it to the benefit of any third party.
- 6.12 Construction. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement will be construed as if drafted jointly by the Parties and no presumption or burden of proof will arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement. Except as otherwise explicitly specified to the contrary, (a) references to a Section, Article or Schedule mean a Section or Article of, or Schedule to, this Agreement, (b) the word "including" will be construed as "including without limitation," (c) references to a particular statute or regulation include all rules and regulations thereunder and any predecessor or successor statute, rules or regulation, in each case as amended or otherwise modified from time to time and any reference to a document is to that document as varied, supplemented or replaced from time to time by agreement between the parties thereto, (d) the headings contained in this Agreement are for convenience purposes only and will not in any way affect the meaning or interpretation of this Agreement, (e) words in the singular or plural form include the plural and singular form, respectively, (f) use of any gender includes the other genders, (g) any reference to writing shall include any modes of reproducing words in a legible and non-transitory form, (h) references to a particular Person include such Person's successors and assigns to the extent not prohibited by this Agreement, and (i) the Schedules and Recitals hereto form part of this Agreement and shall have effect as if set out in full in the body of this Agreement, and any reference to this Agreement includes the Schedules and Recitals hereto.
- 6.13 <u>Governing Law.</u> This Agreement, the rights of the Parties hereunder and all Proceedings arising in whole or in part under or in connection herewith, will be governed by and construed in accordance with the domestic substantive laws of the State of Texas, without giving effect to any choice or conflict of law provision or rule that would cause the application of the laws of any other jurisdiction.

- Arbitration. Any claim, dispute or other matter in question between the Parties hereto arising out of or relating to this Agreement, the rights of the Parties hereunder or the breach hereof, shall be decided by arbitration in accordance with the rules of the American Arbitration Association in effect on the Effective Date before three arbitrators; one designated by the Company, one designated by the Consultant and the third designated in accordance with the Rules of the American Arbitration Association. Any such arbitration shall be conducted in Dallas, Texas, unless the Parties mutually agree to another location. The arbitrators shall be qualified by education, training or experience as may be appropriate according to the nature of the claim, dispute or other matter in question. The foregoing agreement to arbitrate and any other agreement to arbitrate shall be specifically enforceable under the prevailing arbitration law. The award rendered by the arbitrators shall be final, and judgment may be entered upon it in accordance with applicable Law in any court having jurisdiction thereof. To the extent permitted by law, by agreeing to engage in the arbitration provided for in this Section 6.14, the Parties waive their right to appeal any decision made by the arbitrators. The demand for arbitration shall be made within a reasonable time after the claim, dispute or other matter in question has arisen; and in no event shall it be made after the date when institution of legal or equitable proceedings based on such claim, dispute or other matter in question would be barred by the applicable statute of limitations. All costs and expenses (including reasonable attorneys' fees and costs) incurred in connection with any such arbitration shall be borne in the manner which the arbitrators making the determination shall direct. Notwithstanding the provisions of this Section 6.14, either Party may seek appropriate injunctive relief for any threatened breach.
- 6.15 Affiliate Transactions. In connection with the provision of the Services under this Agreement, the Consultant may purchase necessary goods, supplies, rights and services (a) from or through any of its Affiliates or (b) pursuant to arrangements Consultant Related Facilities so long as (i) any such transaction, series of transactions or arrangement is disclosed to the Company and approved by the Board including at least one Board member not affiliated with the Consultant, (ii) in respect of any such Affiliate transaction or series of transactions, the terms and conditions thereof are competitive with the prices and terms of goods, supplies, rights and services of like quality available from non-Affiliated third parties in an arm's length transaction, and (iii) in respect of such shared arrangements, the allocations of costs with respect thereto are fair and reasonable and, if applicable, consistent with the methodology used by the Consultant and its subsidiaries and Affiliates to allocate similar costs among the Consultant Related Facilities.
- 6.16 <u>Authority; Binding Agreement</u>. Each Party represents and warrants that (a) it is a corporation, limited liability company, partnership or other entity, as applicable, duly organized, validly existing and in good standing under the laws of its jurisdiction of organization, (b) it has the requisite power and authority to execute, deliver and perform this Agreement, (c) the execution, delivery and performance by it of this Agreement has been duly and validly authorized by all necessary actions on its part, and (d) this Agreement constitutes the legal, valid and binding obligations of such Party, enforceable against such Party in accordance with its terms.
- 6.17 <u>Further Assurances</u>. Each Party hereto shall further execute and deliver all such other appropriate supplemental agreements and other instruments and take such other action as may be necessary to make this Agreement fully and legally effective, binding and enforceable as between the Parties and as against third parties, or as the other Parties may reasonably request.

[SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, each of the undersigned has executed this Theater Services Agreement as of the date first above written.

# THE COMPANY:

# FE CONCEPTS, LLC

By: /s/ W. Mark Moore
Name: W. Mark Moore

Title: Manager

# THE CONSULTANT:

# CNMK TEXAS PROPERTIES, LLC

By: /s/ Michael Cavalier

Name: Michael Cavalier

Title: EVP – General Counsel

Signature Page to Theater Services Agreement

#### **SCHEDULE I**

#### CERTAIN DEFINED TERMS

- "Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under direct or indirect common control with such Person. For the purposes of this definition, "control" when used with respect to any specified Person means the power to direct or cause the direction of the management and policies of such Person, directly or indirectly, whether through the ownership of voting securities, by contract or otherwise, and the terms "controlling" and "controlled" have meanings correlative of the foregoing.
  - "Agreement" has the meaning set forth in the Preamble hereto.
  - "Auditoriums" has the meaning set forth in the Recitals hereto.
  - "Board" means the Board of Managers of the Company.
  - "Business" means the business conducted at any Facility.
  - "Business Day" means a day (other than a Saturday or Sunday) on which banks are open for business in Dallas, Texas.
- "Cause" means (i) the commission of fraud, embezzlement or theft by the Company as determined by a court of competent jurisdiction; (ii) a breach by the Company of this Agreement, if such breach is not cured within five (5) Business Days (for any monetary breach) or sixty (60) days (for any non-monetary breach that the Company makes reasonable efforts to cure during such cure period) after the Consultant provides written notice of such breach to the Company; or (iii) the intentional wrongful damage by the Company to the business reputation of the Consultant or its subsidiaries in a manner that has a material adverse effect on the Consultant and its subsidiaries, all only if so determined by a court of competent jurisdiction.
  - "Change in Control" means the occurrence after the Effective Date of any one or more of the following:
- (i) any Person becomes the beneficial owner, directly or indirectly, of Common Stock of the Consultant or Cinemark Holdings, Inc. representing more than fifty percent (50%) of the Consultant or Cinemark Holdings, Inc.'s then outstanding Common Stock;
  - (ii) any consolidation, merger, recapitalization or similar transaction involving the Consultant or Cinemark

(iii) the sale, lease, exchange or other transfer (in one transaction or a series of related transactions) of all or any material portion of the assets of the Consultant or Cinemark Holdings, Inc.

"Cinemark" means CNMK Texas Properties, LLC.

Holdings, Inc.; or

Schedule I - 1

- "Class B Competing Business" means an entertainment facility incorporating restaurants, games and bowling.
- "Company" has the meaning set forth in the Preamble hereto.
- "Company Indemnified Person" has the meaning set forth in Section 4.2.
- "Company Subsidiary" means any corporation, limited liability company, limited partnership or other entity that is directly or indirectly wholly owned by the Company.
  - "Consultant" has the meaning set forth in the Preamble hereto.
  - "Consultant Expenses" has the meaning set forth in Section 3.2.
  - "Consultant Indemnified Persons" has the meaning set forth in Section 4.2.
- "Consultant Related Facility" means an entertainment facility (other than a Facility) owned or operated by the Consultant or its Affiliates.
  - "Consulting Fee" has the meaning set forth in Section 3.1.
  - "CSC" means the Customer Support Center where the network operating control center for a Facility is located.
  - "Effective Date" has the meaning set forth in the Preamble hereto.
  - "Facilities" has the meaning set forth in the Recitals hereto.
- Governmental Authority" means any United States federal, state or local government or any foreign government, or any political subdivision thereof, any multinational organization or authority or any authority, agency or commission entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, any court or tribunal (or any department, bureau or division thereof), or any arbitrator or arbitral body.
- "Law" means any United States federal, state or local, or any foreign, law, statute, standard, ordinance, code, rule, regulation, resolution or promulgation, any order, writ, judgment, injunction, decree, stipulation, ruling, determination or award entered by or with any Governmental Authority, or any license, franchise, permit or similar right granted under any of the foregoing, or any similar provision having the force or effect of law.
  - "LLC Agreement" has the meaning set forth in the Recitals hereto.
  - "Losses" has the meaning set forth in Section 4.1.
- "Management Agreement" means the Management Services Agreement between the Company and the Manager of even date herewith pursuant to which the Manager has agreed to provide the Management Services.
  - "Management Services" has the meaning set forth in the Recitals hereto.

- "Manager" has the meaning set forth in the Recitals hereto.
- "Parties" has the meaning set forth in the Preamble hereto.
- "*Person*" means any individual or corporation, association, partnership, limited liability company, joint venture, joint stock or other company, business trust, trust, organization, governmental or regulatory body or authority or other entity of any kind.
- "*Proceeding*" means any action, cause of action or suit (whether in contract or tort or otherwise), litigation (whether at law or in equity, whether civil or criminal), controversy, assessment, arbitration, investigation, hearing, charge, complaint, demand, notice or proceeding to, from, by or before any Governmental Authority.
- "RMA" means Return Merchandise Authorization for monitoring and tracking the shipment and return of parts or replacement parts to and from the Facility to the appropriate vendor.
  - "Settlement Policies" has the meaning set forth in Section 2.3.
  - "Term" has the meaning set forth in Section 5.1.
  - "Theater Services" means the services to be provided by the Consultant pursuant to Article I of this Agreement.
  - "Third Party Claim" has the meaning set forth in Section 4.1(a).
  - "TMS" means the Theater Manager Server for the theater auditoriums located in a Facility.
  - "Transition Services Period" has the meaning set forth in the Management Agreement.

Schedule I - 3

# CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES - OXLEY ACT OF 2002

#### I, Mark Zoradi, certify that:

- 1. I have reviewed this report on Form 10-Q of Cinemark Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15 (e) and 15d-15 (e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15 (f) and 15d-15 (f)) for the registrant and have:
  - designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that
    material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during
    the period in which this report is being prepared;
  - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
  - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date:	August 8, 2018	
By:	/s/Mark Zoradi	
Mark Zora	adi	
Chief Exe	cutive Officer	

# CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES – OXLEY ACT OF 2002

#### I, Sean Gamble, certify that:

- 1. I have reviewed this report on Form 10-Q of Cinemark Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15 (e) and 15d-15 (e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15 (f) and 15d-15 (f)) for the registrant and have:
  - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
  - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date:	August 8, 2018		
By:	/s/Sean Gamble		
Sean Ga	ımble		
Chief Fi	nancial Officer		

# CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADDED BY SECTION 906 OF THE SARBANES - OXLEY ACT OF 2002

This certification is provided pursuant to 18 U.S.C. Section 1350, as added by Section 906 of the Sarbanes-Oxley Act of 2002 and accompanies the quarterly report on Form 10-Q (the "Form 10-Q") for the quarter ended June 30, 2018 of Cinemark Holdings, Inc. (the "Issuer").

I, Mark Zoradi, the Chief Executive Officer of Issuer certify that to the best of my knowledge:

- (i) the Form 10-Q fully complies with the requirements of section 13(a) or section 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m(a) or 78o(d));
- (ii) the information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Issuer.

Dated: August 8, 2018

/s/Mark Zoradi

Mark Zoradi

Chief Executive Officer

Subscribed and sworn to before me this 8th day of August 2018.

/s/ Matthew Creed

Name: Matthew Creed Title: Notary Public

My commission expires: 12/27/2020

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

# CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADDED BY SECTION 906 OF THE SARBANES – OXLEY ACT OF 2002

This certification is provided pursuant to 18 U.S.C. Section 1350, as added by Section 906 of the Sarbanes-Oxley Act of 2002 and accompanies the quarterly report on Form 10-Q (the "Form 10-Q") for the quarter ended June 30, 2018 of Cinemark Holdings, Inc. (the "Issuer").

I, Sean Gamble, the Chief Financial Officer of Issuer certify that to the best of my knowledge:

- (i) the Form 10-Q fully complies with the requirements of section 13(a) or section 15(d) of the Securities Exchange Act of 1934 (15 U.S.C. 78m(a) or 78o(d));
- (ii) the information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Issuer.

Dated: August 8, 2018

/s/Sean Gamble

Sean Gamble

Chief Financial Officer

Subscribed and sworn to before me this 8th day of August 2018.

/s/ Matthew Creed

Name: Matthew Creed Title: Notary Public

My commission expires: 12/27/2020

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.